AD-772 809

ARCHITECTURAL PRINCIPLES FOR VIRTUAL - COMPUTER SYSTEMS

Robert P. Goldberg

Harvard University

Prepared for:

Electronic Systems Division

February 1973

DISTRIBUTED BY:

National Technical Information Service U. S. DEPARTMENT OF COMMERCE 5285 Port Royal Road, Springfield Va. 22151

Unclassified	H	$J = J_{J} \times A =$
Security Classification	TPOL DATA PED	1 2 00
(Security classification of title, body of abstract and indexing	ng annotation must be entered when t	ne overall report is classified)
ORIGINATING ACTIVITY (Corporate author) Harvard University	20. REPORT	SECURITY CLASSIFICATION
Cambridge, MA 02138	26. GROUP	
REPORT TITLE		
ARCHITECTURAL PRINCIPLES FOR VIRTUAL C	OMPUTER SYSTEMS	
. DESCRIPTIVE NOTES (Type of report and inclusive dates)		
None AUTHOR(S) (First name, middle initial, last name)		
Robert P. Goldberg		
REPORT DATE	78. TOTAL NJ. OF PAGES	7b. NO. OF REFS
February 1973 . Contract or grant no		MBE (215)
F19628-70-C-0217	ESD-TR-73-10	5
D. PROJECT NO.		
c.	96. OTHER REPORT NO(5) (Any (his report)	other numbers that may be assigned
<i>d</i> .		
0. DISTRIBUTION STATEMENT		
Thesis	12. SPONSORING MILITARY AC Deputy for Command Ha Electronic System	and Management Systems
Approved for public release; distribution unlimit The sis Div. Engineering & Applied Physics Harvard University The thesis develops principles for dear systems. The virtual computer system as a solution to a problem of present The most important new result of the a virtual computer system (VCS) and a model. The approach adopted is to car rich, complex architectures likely to of the additional system structure in generation virtual machine software a This result leads us away from an infi- implicitly on techniques used in this to develop a generalized model of a pro- model allows us to understand differ- interpret a number of proposed implet Furthermore, the model leads natural the Hardware Virtualizer (HV) which is	ited. 12. SPONSORING MILITARY AC Deputy for Command Hq Electronic System L G Hanscom Field, signing machines to su (VCS) is an important computer system techn thesis is the model of the derivation of desi onsider the introducti o be found in IV generation, to mapping techniques sho terpretation of virtuard generation systems. process running on a I ent properties of virt mentacions of VCS's in ly to an implementation provides an efficient	and Management Systems as Division (AFSC) Bedford, MA 01730 pport virtual computer construct which arises ology. f a process running on gn principles from that on of VCS's into the ation systems. Because he somewhat ad hoc third uld be doomed to failure. I machines that depends Instead, we are able V generation VCS. The ual machines and to terms of the model. n of virtual machines, and simplified mechanism
The thesis develops principles for des systems. The virtual computer system as a solution to a problem of present The most important new result of the a virtual computer system (VCS) and in model. The approach adopted is to car rich, complex architectures likely to of the additional system structure in generation virtual machine software in this result leads us away from an init implicitly on techniques used in this to develop a generalized model of a proposed implex Furthermore, the model leads natural the Hardware Virtualizer (HV) which for virtual machines. A number of develop a generalizer of develop is the software of the software is a software of the softw	ited. 12. SPONSORING MILITARY AC Deputy for Command Hq Electronic System L G Hanscom Field, (VCS) is an important computer system techn thesis is the model of the derivation of desi onsider the introduction o be found in IV generation, to mapping techniques sho terpretation of virtuard generation systems. process running on a I ent properties of virt mentacions of VCS's in ly to an implementation provides an efficient tailed examples illust	and Management Systems as Division (AFSC) Bedford, MA 01730 pport virtual computer construct which arises ology. f a process running on gn principles from that on of VCS's into the ation systems. Because he somewhat ad hoc third uld be doomed to failure. I machines that depends Instead, we are able V generation VCS. The ual machines and to terms of the model. on of virtual machines, and simplified mechanism trate how the Hardware
The thesis develops principles for des systems. The virtual computer system as a solution to a problem of present The most important new result of the a virtual computer system (VCS) and model. The approach adopted is to con- rich, complex architectures likely to of the additional system structure in generation virtual machine software to This result leads us away from an inti- implicitly on techniques used in this to develop a generalized model of a pro- model allows us to understand differ- interpret a number of proposed impler Furthermore, the model leads natural the Hardware Virtualizer (HV) which for virtual machines. A number of de Vitualizer might operate in an actual	ited. 12. SPONSORING MILITARY AC Deputy for Command Hq Electronic System L G Hanscom Field, (VCS) is an important computer system techn thesis is the model of the derivation of desi onsider the introducti o be found in IV generation, to mapping techniques sho terpretation of virtuar rd generation systems. process running on a I ent properties of virtuar mentacions of VCS's in ly to an implementation provides an efficient tailed examples illust IV generation systems.	and Management Systems as Division (AFSC) Bedford, MA 01730 pport virtual computer construct which arises ology. f a process running on gn principles from that on of VCS's into the ation systems. Because he somewhat ad hoc third uld be doomed to failure. I machines that depends Instead, we are able V generation VCS. The ual machines and to terms of the model. n of virtual machines, and simplified mechanism rate how the Hardware
The sis Div. Engineering & Applied Physics Harvard University The thesis develops principles for des systems. The virtual computer system as a solution to a problem of present The most important new result of the a virtual computer system (VCS) and model. The approach adopted is to con- rich, complex architectures likely to of the additional system structure in generation virtual machine software a This result leads us away from an infi- implicitly on techniques used in this to develop a generalized model of a proposed implea- furthermore, the model leads natural the Hardware Virtualizer (HV) which points of the source of the software of the source	ited. 12. SPONSORING MILITARY AC Deputy for Command Hq Electronic System L G Hanscom Field, signing machines to su (VCS) is an important computer system techn thesis is the model of the derivation of desi onsider the introducti o be found in IV generation, to mapping techniques sho terpretation of virtuard generation systems. process running on a I ent properties of virt mentacions of VCS's in ly to an implementation provides an efficient tailed examples illust IV generation system.	and Management Systems as Division (AFSC) Bedford, MA 01730 pport virtual computer construct which arises ology. f a process running on gn principles from that on of VCS's into the ation systems. Because he somewhat ad hoc third uld be doomed to failure. I machines that depends Instead, we are able V generation VCS. The ual machines and to terms of the model. n of virtual machines, and simplified mechanism rate how the Hardware

Service States States

.

SALTA HITLE

NIS ADAD

Unclassified Security Classification		LINI		LIN		1. IN	
KEY NORDS		ROLE	WT	ROLE	WT	ROLE	WT
virtual computer system virtual machine computer systems architecture operating system design virtual memory segmentation paging		ROLE	W T	FOLE	• T	ROLE	WT
	1 C.						

The second state of the second

LEGAL NOTKE

When U.S. Government drawings, specifications or other data are used for any purches other than a definitely related government procurement operation, the solutions other than a definitely responsibility nor any obligation whatsoever; and this fact that the povernment any have formulated, fulnished, or in any way suppilled the solid drawings, specifications, or other data is not to be regarded by implication or otherwise as in any manner licensing the holder or any other person is conveying any rights or permission to manufacture, use, or sell any patented invention that may in any way be related therets.

OTHER NOTICES

Do not return this copy. Retain or desiroy.

. .

.

12

ESD-TR-73-105

ARCHITECTURAL PRINCIPLES FOR VIRTUAL COMPUTER SYSTEMS

Robert P. Goldberg

February 1973

DEPUTY FOR COMMAND AND MANAGEMENT SYSTEMS HQ ELECTRONIC SYSTEMS DIVISION (AFSC) I.. G. Hanscom Field, Bedford, Massachusetts 01730 TRONIC SYST

Approved for public release; distribution unlimited.

(Prepared under Contract No. F19628-70-C-0217 by Harvard University Cambridge, Massachusetts 02138.)

 $|_{\mathcal{X}}$

J#1 21 1974

ξ.

FOREWORD

This report was prepared in support of Project 2801 by Harvard University, Cambridge, Massachusetts under Contract F19628-70-C-0217, monitored by Major Roger R. Schell, HQ. ESD(MCIT), and was submitted January 16, 1973.

This technical report has been reviewed and is approved.

772L mmo

MELVIN B. EMMONS, Colonel, USAF Director, Information Systems Technology Deputy for Command & Management Systems

ABSTRACT

The thesis develops principles for designing machines to support virtual computer systems. The virtual computer system (VCS) is an important construct which arises as a solution to a problem of present computer system technology. The most important new result of the thesis is the model of a process running on a virtual computer system and the derivation of design principles from that model. The approach adopted is to consider the introduction of VCS's into the rich, complex architectures likely to be found in IV generation systems. Because of the additional system structure in the IV generation, the somewhat ad hoc third generation virtual machine software mapping tec! niques should be doomed to failure. This result leads us away from an interpretation of virtual machines that depends implicitly on techniques used in third generation systems. Instead, we are able to develop a generalized model of a process running on a IV generation VCS. The model allows us to understand different properties of virtual machines and to interpret a number of proposed implementations of VCS's in terms of the model. Furthermore, the model leads naturally to an implementation of virtual machines, the Hardware Virtualizer (HV), which provides an efficient and simplified mechanism for virtual machines. A number of detailed examples illustrate how the Hardware Virtualizer might operate in an actual IV generation system.

PREFACE

Carles Barton and the art of the state of th

I am deeply indebted to my advisor and friend, Dr. U. O. Gagliardi for his wisdom and kindness. It was Dr. Gagliardi's encouragement and confidence in me at several crucial points during my academic career that allowed the research to continue to a successful conclusion. In particular, collaboration with Dr. Gagliardi on an earlier paper, "Virtualizeable Architectures" led to some key insights that have become an important part of the thesis.

I would also like to express my gratitude to the other nembers of my committee, Professors T. E. Cheatham, Jr. and R.V. Book, and Mr. G. H. Mealy for the time they have invested in reading the thesis and for their pertinent comments. In addition, Mr. Mealy's question (several years ago) of whether a virtual 360/67 could be run under CP-67 essentially started the research on this thesis.

I would like to thank my colleagues at both MIT and Harvard for the numerous discussions about virtual machines over the years. Particular note for their stamina is sue to J.P. Buzen, H.S. Schwenk, S.E. Madnick, A.L. Brown, and R.M. Kierr.

I would also like to thank A.W. Armenti, S.W. Galley, and J.F. Haverty for a very careful reading of the thesis. Any felicities of style can be traced to their persistence and suggestions.

Three people aided in the preparation of the thesis and

iv

sincere thanks is due them. They are Lloyd Dickman who suggested and executed some of the figures in Chapter 3, my cousin Paul Goldberg who drew the figures in Chapter 2 and calculated and plotted the data in the graphs of Chapter 4, and my wife Judith Goldberg who prepared the remainder of the figures.

Excause of the long duration of the work, portions of the research were carried out at MIT Lincoln Laponatory and MIT Project MAC as well as Harvard University Center for Research in Computing Technology. It is a pleasure to cite this succort and acknowledge the encouragement given by J.F. Nolan, J.J. Vitygenald, ad J.C R. Licklider.

I a did like to express my gratitude to my family and friends, especially my mother and sister, for their sincere incerest and good wishes. Finally, my wife, Judy, deserves special recognition for her remarkable batience and subport during the preparation of the thesis. Her many significant contributions, both real and virtual, are greatly appreciated. [This research was supported in part by AF13(528)-5157, DAHC15-69-C-0347, and F19628-7C-C-0217. The thesis was prepared using the Honeywell MULTICS System.]

TABLE OF CONTENTS

	<i>4</i>
Prefece	iv
List of Figures	vii
Synopsis	xii
1. Introduction	1
1.9 Overview	1
1.1 Plan of the Iresis	10
2. Virtual Computer SystemsTerminology and Conkground	14
2.0 Plan of Chapter 2	14
2.1 Terminology for VCS*s	15
2.2 Comparison with Pelated Notions	28
2.3 Examples of Virtual Machines	32
2.4 Related Literature	39
 Principles for III Generation Virtual Computer Systems- Inadequacy of Current Designs 	- 43
3.0 Plan of Chapter 3	43
3.1 Characteristics of III Generation Computer System	s 45
3.2 Empirical Hardware Requirements	47
3.3 Hybrid Virtual Machines and Requirements	55
3.4 Ac Hoc Hardware "Improvements"	58
3.5 III Generation Empirical Software Requirements	67
3.6 Software Primitives for Type II VCS's	73
3.7 Conclusion	77

Sec. 1

1 H . .

i. Netroj LIST OF FIGURES

and the second second and the second second and the

Page

Figure

Same rates

and a structure of the state of the

a contraction of the second second

<u>1+1</u>	TYDICAL VUS UP-67
?-?	Level vs. Layer in III Generation VCS
2-3	Virtual Machine vs. Other Constructs
2-4	Type I vs. Type II VCS
2-5	Machine Environments
3-1	III Generation Vintual Processor Map
3-2	III Generation VCS Case Studies Summary
3-3	EVM Mode Mapping
3-4	Trap Pegister and TOO Instruction
3-5	Virtual Machine Support for Trap Register
3-6	Sensitive Locations in Unaddressable Memory
3-7	Type II VMM
3-8	III Generation Type II Case StudiesSummary
3-9	Process Trees
4-1	System Asse
4-2	The Process Map Ø
+-3	The VMmap f
' +	VM Levels
+	foo
4-6	Process Exception and VM Fault
t-?	Type II VMmap
4-8	Type II Virtual Machine

3.	Cass	Studies of Sche III Generation Machines	198
	8.0	Introduction	199
	3.1	IHM 360/87	198
	2 • ۲	IFM 363/65	263
	B • 3	HITAC 840L	262
	ន.4	ICM 360/85	203
	B.5	Cata General NOVA	203
	8.6	Honeywell DEP 516	273
	8.7	GF 635,655 (Honeywell 5036)	205
	8.8	Multidata Model A (SFL)	215
	u•d	XOS 943	205
	8.10	CFC PDP-1(206
	9.11	BBN TENEX	206
C.	Case	Studies of III Generation Type II Systams	213
	0.0	Introduction	205
	C.1	HITAC 8400	263
	C.2	360/67 UMMPS	209
	C.3	09/353	211
	C + 4	MIT PDP-10 ITS	214
	C.5	CMS, The Carphidge Monitor System	215
Э•	Gloss	dry	217

Jibliography

and the second second

TANA SEA

n ar we are the set

22?

. .

viii

Sec 26

' + •	Principles for 10 Generation Virtual Computer Systems The Virtual Machine Map and its Firmware Inclamentation				
	4.0	Plan of Chapter 4	63		
	4 • 1	Characteristics of IV Generation Computer Systems	83		
	4.2	Model of a IV Generation VCS	92		
	·:• 3	Unsuitability of Software f	107		
	4.4	The Venice Paper	114		
	4.5	The Hardware Virtualizer	118		
	4.8	Detailed Illustrations	138		
	4.7	Some Practical Problems of IV Generation HV	175		
	4.8	II and III Generations Revisited	178		
5. Conclusion		usion	182		
	5.0	Summary	182		
	5.1	Suggestions for Further Research	184		

ì

10 เป็นระเมืองแระสำเร็จที่เหมือนสำเร็นให้ที่ระดีปลิติสินิติสนิติสนิติสนิติสนิตารแรง เป็นการยางแหน่งการการการการการ

Appendices

and the second second second second

「ないちましまかいとうないななないという」というかってい

;

ŝ

4 - 4

ĩ

ville e.

and in the second and the second residential the second strategy is the second second second second second second

trisseensterkerkerkerensetheinbelt (112) Friesener officieren de stattebester N/I Edm

·· ·

7•	Tutor	ial on CP-E7		187
	A . C	Introduction		187
	4.1	CP-67°s Virtual	Memory	188
	A.2	CF-67°s Virtual	1/0	190
	Δ.3	C?-67's Virtual	Processor	194

CARLES AND AND A CONTRACT OF A CONTRACTACT OF A CONTRACT OF A CONTRACT.

The second second second second second second

4-9	Type I Pecucsion	106
+-10	Type II Recursion	106
+-1j	Ping Mapping Schemes	110
+-12	Example of Ring Pelocation Register	112
'+ - 1 3	Verice Proposal	116
4-14	Venice Proposal with Pecursion	116
4-15	LVMID Instruction	123
4-16	VM Tree Structure	127
4-17	VM-Fault	130
- -18	HV Performance vs. n	136
4-19	HV Performance VS. Pk	1 37
+-21	System Base	139
+-21	Notation for Segment Table of Running Process	141
+ - 22	The VCSTAB and VCSCB*s	143
4-23	Notation for P-E VCSCF	144
+-24	Notation for Paged VCSCB	144
+-25	Example 1: The LVMID Instruction	146
4-28	Fxample 1: Execution and Exception	147
4-27	Example 1: Execution and System Base Updata	151
+-28	Example 1: Execution and VM-Fault	153
+-29	Example 1: Associator	154
4-30	Example 2: Execution and Associator	156
4-31	Example 3: VP Execution	153
+-32	Example 4: Execution of Paged VCS	169
4-33	Example 4: Paged Associator	161
4-34	HV VS. 645, 369/67	163

ide na treast de contrativantes de contrativantes de contrativantes de contrativantes de la contrativante de contrativantes de contrativantes

DAA WALANDARA HURARIKI SA MILANIKA MARANA

912577

AN ALCONOMIC STREET

x

En Martin Print States

4-35	Example 5: Execut	ion and Associator	164
+ - 3E	Example 6: Semaoh	ones and Process Queues	165
4-37	Example 6: System	Base Detail for Semaphonas	168
4-38	Example 6: System	Base Detail After p[ser]	169
+-39	Example 7: Multip	le Processor's Frocessor Map	171
4-40	Example 8: Type I	I VCS	173
+-41	Interim 1/0 Propos	31	177
+-4ĉ	"f=R-B, &=identity	" Machine	183
1-1	Mapping Relocated	Virtual Memory to Peal Memory	189
7-5	Virtual - Reat I/C	Interface	193
0-1	OS/36' Does Not Ma	sk SVC*s	213
2-2	Type II HVM Under	PDF-10 ITS	216

14 · · ·

Ľ

anter history

<u>.</u>

. . .

Same the second states

-:i

SYNOPSIS

The thesis develops principles for designing machines to support virtual computer systems. The virtual computer system (VCS) is an important construct which arises as a solution to a particularly vexing problem of present computer system technology.

multi-access, While the recent development of large nulti-orogramming, multi-processing systems has simplified and improved access to computer systems by the bulk of the user community, there has been one important class of users unarte to profit from these recent advances. This class is the system programmers whose programs must be run on a bare machine and not on an <u>extended</u> machine, e.g. under the operating system. System programs, e.g. other operating systems or different versions of the same operating system, require cirect addressability to the resources of the system and do not call upon the operating system to manage these resources. Thus, system cregrams may not co-exist with normal production uses of the system. This forces most installations into clumsy and situation has "stand-alone" inefficient administrative solutions such 25 Jebugging.

Recently, a technique for resolving these difficulties has been devised. The solution utilizes a construct called a virtual computer system or virtual machine which is, basically, a very efficient simulated copy (or copies) of the bare hast machine. These copies differ from each other and from the host only in their exact configurations, e.g. the amount of memory or particular I/O devices attached. Therefore, not only standard user programs but system programs and complete operating systems that run on the real computer system will run on the VOS with identical results lexcept for certain special timing dependencies). Thus, the VCS provides a generalization over familiar systems by also being a multi-environment system.

ないないというないないないないないろうなたいでないまであって

The recent origin of the field has forced the development of a terminology to represent ideas in VCS*s. Chapter 2 introduces some basic notions in VCS*s and proposes terminology to represent them. The terminology is new but some excerpts were previously published by the author in "Virtual Machires: Semantics and Examples" [53].

Despite the rather significant benefits that derive from virtual machines, only a limited number of current systems feature this facility. This is largely due to the unsuitable nature of existing hardware. In order to support a VCS on contemporary equipment, a particular software construction must be employed. However, this construction may only be applied to machines with certain properties. Chapter 3 examines the construction of VCS's on third generation computer systems and derives a set of empirical requirements to determine if a machine may be virtualized. The appendices treat the application of these empirical rules to a number of case studies. This material is new although some preliminary results were reported by the author in "Virtual Hachine Systems" [54] and "Hardware

xiii

Requirements for Virtual Machine Systems" [52].

and the second second second

DEVELOPED INTERPORTATION OF THE PARTY IS IN THE PARTY IS INTERPARTY IS IN THE PARTY IS INTERPARTY ISTURATION IS INTERPARTY I

The remainder of the thesis (Chapter 4) develops a structure for computer systems that cermits the orderly introduction of IV gereration computer VCS's into future computer systems. systems will likely feature a formal implementation (in firmware) Because of the existence of a large of the process model. database needed by the firmware (and privileged software) 10 support the process model, it will be difficult to employ the software mapping technicues used in III generation VCS*s. Δ nodel is developed to represent the execution of processes on a IV generation VCS. The model features two maps: (1) a process map called & which maps process names, e.g. segments, samaphores, process-id*s, into resource names, e.g. memory locations, processor numbers, and (2) a virtual machine mac (VMmap) f which maps virtual resource names into real resource names. The process map is strictly an intra-level mac expressing а within a virtual machine; the VHmap relationship is an inter-level map expressing a relationship between (the resources of) two adjacent levels of (virtual) machines. Thus, the action of running a process on a VCS consists of running it under the composed map f o 0. If the VCS itself is running a VCS under it, then the action consists of running a process under f o f o 4. Thus, for VCS recursion, only the f map must be invoked recursively, not the process map 0. This result implies that for a complex & but a rudimentary f, recursion should be easy.

The IV generation VCS model leads directly to a design, the Hardware Virtualizer (HV), for proposed implementations of IV

xiv

generation VCS*s. The design has the characteristic that all process exceptions are handled directly within the executing VCS without software intervention. All resource faults (VM-faults) by B VCS are directed to its Virtual Machine Monitor (VMM) without knowledge of processes on the VCS. Fault handling; invocation and execution of VCS*s works directly regardless of recursion.

Betailed illustrations are provided for several possible choices of the VMmap f, e.g. relocation and bounds (R-B), paging. Preliminary performance estimates indicate that for an R-B VMmap f (with associative memory), VCS performance will be approximately that of the real machine, regardless of the deoth of recursion. A paged VMmap f provides performance comparable to the real machine for several levels of recursion over a range of likely onerating conditions.

Although the model and proposed implementations of Chapter 4 arise in an attempt to guarantee virtualization for IV generation architectures, by suitable simplification and interpretator of the model, the principles which emerge are applicable to other architectures, as well. In particular, the model suggests introducing paging in the IPM 360/67 as a formal f-map, rather than an ad hoc 4-map as was done. This leaps to a greatly simplified structure for CP-67.

All of the material presented in Chapter 4, the IV generation VCS model, suggested implementations, performance expectations, examples, etc., is new and original. An early proposal for a firmware-pssisted implementation of VCS's was tescribed by the author [with U.O. Gagliardi] in "Virtualizable Architectures" [51]. However, the interpretation of that procosal in Chapter 4, its strengths and weaknesses in terms of the IV generation VCS model, is new and original.

terse sector de secondaria la terra de la contrada da contrada da partecidada en la contrada da contrada da con

A MANDANA MANANA MAN

ality of the fraction of a first of the

and the second second

Page 1

CHAPTER 1. INTRODUCTION

1.9 OVERVIEW



The Need for Virtual Machines

Charles Shower physics and an environment of the state of the

Recent developments in computer system cesign have heen directed toward improving the ease of computer use by the normal user population. Some of the more gramatic developments include introduction of large multi-access, multi-programming, and the These systems have eliminated the need multi-crocessing systems. for physical access to the main computing facility by the users, and for direct. software access to the system resources by the Thus, users sit at teletypes or submit lots via users' programs. remote card readers and need not be corcerned either the with status lights or switches on the central crocessor, or with of the allocation of memory, I/O devices, or processor time to their programs. These functions are now performed (transparently **to** the users) by the operating system. When either users or their programs repuire services to be performed, they call upor the. operatirg which performs the actual system ganioulation and allocation of nerources. In some sense, the operating system (together the nardware) provides an idealized or <u>extended</u> with view of a computer sytem. Thus, program preparation, debugging, and running has become significantly easier for most of the user population.

Unfortunately, one very important group of users has been unable to reap the benefits of these developments and advances. These and the usens who, for a variety of reasons, desire or need lirect access to the resources of the system. Their programs are written to run on a bare machine. Calling upon an operating system (the so-called <u>extended machine</u>) will not do for there. We will term such users <u>system programmers</u> and the programs they desine to run, system programs. Because system programs cannot run under the normal operating system in the production environment, it is usually recessary to treat them as special When they must be run, the normal operating system is cases. prought dowr and a system program may run "stand-alone." This practice is inconvenient for both the system programmers and the normal users, and inefficient and wasteful of the svstem recources. Furthermore, it can introduce addition! personnel and procedural complexities. Indeed, it is ironic that those who develop and maintain operating systems are often the very people who take the least advartage of their efforts.

A few specific examples of some system programming tasks might clarify these points.

> (1) Debugging the operating system-- The operating system is written to run on the hare machine. Currently, debugging and improving the operating system is treated as a second class activity, often assigned to computer time in the middle of the night.

> (2) Running diagnostic software-- The software for

Fage 2

Fage 3

testing malfunctions in the various units in the system, e.g., is a tabe drive recording correctly, is a disk drive saeking correctly, etc. normally runs stand-alone on the bare machine on on a rudimentary monitor (Test and Diagnostic Monitor) which itself runs on a bare machine.

(3) Punning other operating systems-- When there is more than one operating system for a given computer system (or even different releases of the same operating system) it is often easier to run a subsystem (compiler, interpreter, etc.) under its own operating system than to take the time and cost to convert it to another operating system.

<u>Mhat are Virtual Machines?</u>

Pecently, a technicue for resolving these cifficulties has been deviser. The solution utilizes a construct called a <u>virtual</u> <u>concuter system</u> (VCS) or <u>virtual machine</u>. [Section 2.1 covers nost of this terminology.] Basically a virtual machine is a very efficient simulated copy (or copies) of the tare host machine. Because of its functionality, it is possible to run system programs: because of its efficiency, it is reasonable to run them in the normal production environment. The program which mediates between the virtual machine and the actual resources of the system (the real or host machine) is called the <u>virtual</u> <u>machine monitor</u> (VMM). Since the virtual machine is identical to

Fage 4

eproduced from est available copy.

the nest naching in all significant respects, the VMM need not instruction-by-instruction Suc Jok the familian software interpretation technique that normally must be used. Rather, virtual machines may be constructed in such a way that most instructions of the virtual machine execute on the host directly. Just those instructions which cannot be permitted to execute directly must be interpreted. Thus, we incorporate these rotions into our definition of a virtual computer system in order 10 distinguish it from a number of other objects which have often been casually called virtual machines.

> A virtual computer system is a haroware-software duplicate of a real existing computer system in which a statistically dominant subset of the virtual processor's instructions execute on the host processor in native mode.

Thus, a VCS provides an efficient operation of one on more cories of a complete conduter system, similar to the host lor re; 1) system. These cooics differ from each other and from the host only in their exact configurations, e.g. the emount of memory on the particular I/O device, attached. The virtual and neal processors must either be identical or members of the same concuter family, e.g. IBM System/360-370. Therefore, not only standard user programs but system programs and complete operating systems that run on the real computer system will run or the VCS with identical results Texcept for certair special timina dependencies). Thus the VCS provides a generalization over the farilian multi-access, multi-programming, *«ulti-crocessino* systems, by also providing a multi-environment system.

In addition to permitting the three system programming

examples, above, to opexist with normal production uses of the system, virtual machines introduce an additional segree of system flexibility. This flexibility is a consequence of the additional level of binding in a virtual machine, between the resourc 3 referenced by system programs running on the VCS and the actual resources that they correspond with. Some immediate advantages aret

> (1) Punning with a virtual configuration which is different from the real configuration-- This use can be important for running systems with more virtual memory or processors than actually exist, or debugging computer retworks and telecommunications applications. (2) Measuring operating systems-- Since the VMM mediates between the virtual and real resources of a system, it can measure how an operating system on a virtual machine is manipulating its resources (without requiring a modification to the operating system).

> (3) Adding hardware enhancements to a system-- If hardware enhancements, e.g. paging, are added to the host machine, it may still be possible to run a virtual machine which does not incorporate these enhancements. Under these conditions the operating system running on the virtual machine need not be modified. Thus, with relatively simple VMM modifications, the hardware enhancements can be utilized to provide improved resource handling and capabilities.

Fage 5

<u>Sucrent Virtual Machine Uncerstanding</u>

Despite the nather significant benefits that derive from virtual machines, only a limited number of current systems feature this facility. This situation is a result of (1) the nec-nt origin of the concepts, (2) misunderstandings of the techniques for implementing virtual machines, and (3) the largely unsuitable nature of existing hardware. Thus, since existing computer systems were not designed to support virtual machines, trying to implement them is largely a hit-or-miss procosition.

In the thesis, we explore the problems associated with implementing virtual machines on contemporary hardware. By developing an empirical basis and a set of hardware nules for determining which existing systems support virtual machines, we are able to verify the inaceduacy of current designs [Chapter 3].

At last, there is a growing realization of the importance of designing machines which are virtualizable. i.e. support virtual machines. To date, there seem to be two occosing schools of thought. While both views have some validity, both have significant disadvantages.

The first view [55], typified by Lauer and Snow [73], angues that since the virtual machine must have all the functionality of a real machine, the simpler the real machine is the easier will be the virtual machine construction. [See Section 2.4 and 4.8.] In particular, they suggest eliminating supervisor state, memory mapping, etc. What they introduce instead is a special relocation-bounds type memory map, in which the apsolute contents of the register may be added to but not need or written. Such an approach greatly simplifies virtual machine construction, but it has as a major weakness that the virtual machine's programming environment is rather severe and devoid of structure. This fact implies that the development of large, interesting systems on the virtual machine marks be difficult. The authors observe that their procosal "... lacks some of the important features of modern systems, particularly segmented virtual memories and a suitable parameter passing mechanism..." [76].

The second view, introduced by U.O. Gagliardi and the author [51] argues that in complex (likely IV generation) computer systems with a firmware implementation of the process model and layered segmented address structure, there will be a need for firmware support of virtual machires as well. [See Section 2.4 and 4.4.) What emerges is a proposal which directly supports the image of a process executing on a vintual machine. The major advantage of this proposal is that it is applicable to very complex computing environments that are likely to be developed in the near future. The principal disadvantage is that, since this approach does not propose a direct hardware resource map, a certain amount of software intervention is still required. In addition, there are some fundamental limitations on the kinds of recursive structures (running a VMM on a virtual machine) that can be supported.

Results of the Research

understation and the state of the

A STATE AND A STAT

The principal result of the thesis is the cevelopment of a

Fage 7

Page 8

model for running a process on a complex (IV generation) computer system [Sec"ion 4.2]. The model identifies a formal resource map (virtual machine map) and distinguishes it from the more familiar process map. This allows us to understand a number of very complex pheromena in a relatively simple way, and pertits us to interpret the two earlier proposals [51,76] as merely special (sub-optimal) applications of the model [Sections 4.4 and 4.8]. Furthermore, the model leads us directly to an optimal proposed implementation [Section 4.5] which is, in some sense, a synthesis of the better ideas of the two previous proposals. Thus, the proposal is coplicable to the complete range of computer systems, including the complex (IV generation) future systems, yet retains the directness and simplicity of a hardware resource map which needs little software support. In addition. since this implementation is cerived directly from the mocel, the recursive invocation of virtual machines, i.e., running a VMM under a VMM atc., poses no additional problems. Furthermore, the implementation should be very efficient. while performance measures are hard to compare, there is evidence that for certain likely choices of maos, the performance of the virtual machine will be extremely close to that of the real machine. Although the model and proposed implementations of Chapter 4 arise in an sttempt. to guarantee virtualization for I۷ generation achitectures, by suitable simplification and interpretation of the model the principles which emerge are applicable to other generations of architectures [Section 4.8].

Perhaps the major contribution of the thesis is a clear understanding of the architectural crinciples for virtual machine support. By following the guidelines set forth in the thesis, the designer of a future computer system can be assured that his machine will be virtualizeable. Thus, the dissemination of these principles should have a highly significant impact upon both the

thecretical and practical aspects of computer science.

.....

Computer architecture, like other architecture, is the art of determining the needs of the user of a structure then designing to meet those needs as effectively and as possible within economic t ne techrological constraints. Architecture **rust** include engineering considerations, so that the design will be economical and feasible: but the emphasis in architecture is upon the needs of the user, whereas in engineering the the emphasis is on the needs of the fabricator [21].

Fage 9

The state of the source of the

Page 10

1.1 PLAN OF THE THESIS

HARE AS A LAD - HORE OUR - D. POR. 1-

The remainder of the thesis is divided into four chapters. Each chapter is preceded by an introduction which outlines the material to follow, but perhaps some brief organizational remarks mere will be helpful.

a the and the second states of the second second

The most important new result of the thesis, the model of a process running on a virtual computer system (VCS) and the terivation of design principles from that model, is presented in Charter 4. This chapter is largely self-contained, and the reader knowledgeable about current virtual machine applications and technology should be able to skip directly to Chapter 4, if desired. The approach adopted in this chapter is to consider the introduction of VCS's into the rich, complex architectures likely to be found in IV generation systems. Recause of the additional system structure in the IV generation, the somewhat ad hoc third generation virtual machine software mapping techniques should be dromed to failure (Sections 4.1 and 4.3). This result leads us away from an interpretation of virtual mochines that depends implicitly on techniques used in third generation systems. Instead, we are able to develop a generalized model of 3 process running on a IV generation VCS [Section 4.2]. The model allows us to understand different properties of virtual machines and to interpret a number of proposed implementations of VCS's in terms of the model [Section 4,4]. Furthermore, the model leads naturally to an implementation of virtual machines, the Hardware Virtualizer (4V) which provides an efficient and simplified mechanism for virtual machines [Section 4.5]. A number of detailed examples illustrate how the Hardware Virtualizer might operate in an actual IV generation system [Section 4.6]. In addition, the principles developed in this chapter are generally applicable to other architectures as well as the highly complex IV generation. Proper simplification and intercretation of the model leads to retrospective principles for how the second and third generation systems <u>should</u> have been constructed to support virtual machines [Section 4.8].

The earlier chapters develop a basic understanding of virtual machines and existing techniques for implementing them, and provide additional motivation for seeking the breakthroughs described in Chapter 4. In Chapter 2 we introduce some of the pasic virtual machine terminology and properties, and survey existing implementations and related literature. The approach we take to terminology in Chapter 2 is largely descriptive and qualitative. Later, in Section 4.2, a number of those notions are re-cast in terms of the VCS model of Chapter 4.

Chapter 3 examines the existing technology for third generation virtual machines and shows why and how it is inacequate. From the construction used in IBF's CP-67 [Appendix 4] we derive the general software implementation of third generation VCS's. A close scrutiny of typical third generation systems leads to a series of empirical hardware requirements for virtualizable third generation architectures [Section 3.2]. Application of these rules in a number of case studies indicates that most third generation systems cannot be virtualized

Page 11

Page 12

[Appendix B]. This result points out the general unsuitability of third generation architectures (for virtualization) and the weaknesses of the software construction technique. Thus, we are forced to look elsewhere for the key issues and solutions in designing virtualizable architectures [Chapter 4].

Other material presented in Chapter 3 includes requirements for hybrid virtual machines (HVM) [Section 3.3], ad hoc improvements to third generation hardware [Section 3.4], software requirements for Type II (extended machine host) virtual machines [Section 3.5], and suggested virtual machine software crimitives [Section 3.5]. Additional third generation subjects are provided in the first three appendices which deal with a CP-67 tutorial [Aopendix A], case studies of some third generation machines [Appendix P], and case studies of some third generation Type II pperating systems [Appendix C].

Chapter 3 should be of particular interest to two classes of readers. The first class includes those individuals running third generation systems and contemplating the introduction of a virtual machine facility. The various empirical rules and studies should help to determine if virtual machines are possible on the reader's system and what practical options may be taken. The second class of readers includes computer system designers who can observe now relatively insignificant design decisions have rendered most third generation machines unvirtualizable. It will be necessary for these designers to avoid such errors in designing the virtual machine support for the IV generation.

Chapter 5 gives a very brief summary of the research and

Page 13

points out several promising areas for additional study.

nen petin rime inder sie en sitt für die Bernstein er Bernstein als die der Mitter er Bernstein et die sitter e

12.20.20

3<u>-</u> ;

The final material of the thesis, Appendix D, is a glossary of some of the terms used in the thesis. In addition tc the specialized terminology for virtual machines, e.g. virtual machine recursion property; э number o f general though not widely-known terms, e.g. interior decor, are also defined.

Perioduce in the continue of t

CHAPTER C. VIRTIAL COMPUTER SYSTEMS Terminology and Background

2.1 FLAN OF CHAPTER ?

Chapter 2 is divided into four sections. The first section introduces some of the basic terminology that is used in the thesis. Since vintual computer systems have only recently come into existence, we have been forced to develop rew terms to represent ideas about VCS*s. The approach taker in this section is somewhat descriptive. Later, in Section 4.2, a number of these rotions are recest in terms of the VCS model of Chapter 4.

In the second section, we contrast the notion of a virtual machine with a number of related concepts. Next, examples of existing virtual computer systems are cited. In the final section, published literature relating to virtual machines is priefly surveyed.

Fage 14

" The way we wat the to the interest of the prover and the second and

Page 15

2.1 TERMINCLOGY FOR VCS*S

A CONTRACTOR OF THE OWNER OWNER OF THE OWNER OWNER

The term Virtual Computer System has been loosely applied to a range of different computer system organizations. Recently, as noted in Chapter 1, its use has been reserved for a specific type of computer system entity.

> Computer nardware-software Δ Virtual System is a duplicate of a real existing computer system in which a dominant subset of the statistically virtual processor's instructions execute directly on hest the processor in native mode.

There are two parts to this definition.

Ś

Environment-- A virtual computer system must simulate a real existing computer system. Programs and operating systems which system must run on the virtual system with ori the real กับก identical effect. Since the simulated machine may run at 3 different speed from the real one, timing dependent processor and I/O code not perform exactly as intended. This is very rav. the limitation, however, that characterizes similar to interchangeability of programs among different members of a "compatible" computer family, such as between the IBM 353/43 and the 36%/75.

Most instructions being executed must be Implementation-the host CPU processed directly by without recourse to instruction by instruction interpretation. This guarantees that the virtual machine will run on the host with relative It also compels the virtual machine to be similar or efficiency. idertical to the host, and forbids tampering with the control

store to add an entirely new order code.

The VCS and/or host configurations are made up of one or more central processors, main memory, and I/O devices (and possibly I/O processors too). Usually the virtual computer system configuration will have only one central processor. That processor may be called a <u>virtual machine</u>. However, the liferature often informatly uses virtual machine and virtual computer system, VH and VCS, as interchangeable. We shall do likewise.

Sirce a VCS is a hardware-software duplicate of a "real existing computer system" there is always the notion of a <u>real</u> <u>computer system</u>, RCS, or <u>real machine</u>, RM, whose execution is functionally equivalent to the VCS.

The program executing on the host machine that creates the VCS environment is called the <u>Virtual machine monitor</u>, <u>VMM</u>. OP-E7 [9,65,85], which is ciscussed below, gives the VMM the generic name of "control program". Only in the discussion of OP-E7 will we depart from the use of VMM.

Self-victualizing vs. Family-victualizing VMM

The VCS differs from the host only in its exact configuration, e.g. the amount of memory available on the particular I/O devices attached. A further distinction might be in the precise characteristics of the virtual processor. The "implementation requirement" in the VCS definition, above, implies that the virtual processor must be similar or identical to the host processor. If they are not identical, then the
virtual machine must be a member of the same processor family as the host. For example, the virtual and host processors might be different models of a compatible product line, e.e. 373/155 and 360/39, 360/67 and 360/65, or Data General Supernova and Nova. The virtual and host processors might even be the same model processor, but only differ in the fullness of the instruction set implemented, e.g. universal instruction set vs. standard instruction set.

A Manufacture and a second second

Apple in the or the way the

This distinction may be captured by the following two categories:

<u>Self-virtualizing</u> (SV)-- The virtual machine is identical to the nost.

<u>Family-virtualizing</u> (FV)-- The virtual machine is a member of the same computer tamily as the host.

In a self-virtualizing VCS, the functionally equivalent real computer system is identical to the host. Therefore, we sometimes call the host machine, the real machine. We then say that the real/host machine has a <u>virtualizable architecture</u> [51].

If a VGS is self-virtualizing, then it is possible to run another copy of the VMM on the VGS, thus, producing another level of virtual machines. This demonstrates the virtual machine <u>recursion property</u> and is of distinct practical importance since it allows alterations to be made to the VMM running on the virtual machine and permits testing the VMM in the normal operating environment. ISee Section 4.2 for the development of VM recursion in terms of the VCS model.] We say that the VMM

running on a level n VM, then it produces a level n+1 VM. In particular, the real/host machine is level 0. In Figure 2-1, the 512K virtual 360/67 running CP-67 is level 1 and its two virtual machines are level 2.

Level vs. Laver

のためやくないでのないないないであるというないでいたので、

The notion of VCS <u>levels</u> should not be confused with the concept of <u>layers</u> in a computer system. As will be discussed in Chapters 3 and 4, III and IV generation computer systems have structures which implement layers of restricted access to data or instructions. In III generation systems, the two layers are called Master/Slave (supervisor/problem, JSUALLY moce Executive/User, etc.). In IV generation systems, the layers will likely be called rings [57,106]. Since a layer is a relationship between two parts of an individual VCS and a level is a relationship between two VCS's, they are somewhat independent notions. In particular, each level of a VCS must appear to have the same number of layers as the real machine. However, as will be seen later, [Section 4.3] software implementation of VM*s have utilized the restricted access of layers to simulate the effect of levels. This is merely an implementation technique and not a general principle. These issues are discussed further in "Virtualizeable Architectures" [51] and Section 4.2. Figure 2-2 illustrates the relationship between levels and layers for the CP-67 VCS shown in Figure 2-1.

and in the second structure of the second state of the second second second second second second second second

「ないない」というないというないないで、このないないないないでいってい

a second a second s



と言葉の法語の語言のなど、と言語の語言

いていたたいしい

TYPICAL VCS — CP-67 FIGURE 2-1



instanting and a start of the s and the second second second second second

Virtual Machines and Related Constructs

implementation recuirement of the VCS definition The indicates that a "statistically dominant subset of the virtual processor's instructions execute cirectly on the host processor in native mode." To provide functional equivalence between the VCS and real computer system RCS, those instructions which do not execute directly must be simulated on an instruction-by-instruction basis. Thus, the "performance" of the virtual machine is bound by the real machine and by a "complete software interpreter machine," <u>CSIM</u>. Rather than set any precise performance objective or require any myopic machine dependent implementation as part of the definition of virtual machine. we prefer to treat VCS as a broad class of cotential systems. However, for a particular system under consideration, we normally apply the term VCS if the direct execution subset is maximal.]

Thus, the "complete software interpreter machine" is not a VCS since no instruction of the CSIM executes directly. Neither is the real machine. Although all instructions of the RM execute directly, the RM is not a hardware-<u>software</u> duplicate, and indeed it requires no VMM.

A recently distinguished VCS entity is the <u>hybrid virtual</u> <u>machine</u> HVM. The <u>HVM</u> is a VM in which <u>all</u> supervisor state, e.g. Master Mode on Ping 0, instructions are interpreted. The HVM has oeen found to be a useful and easy-to-construct artifact where more familiar VCS techniques have failed. [See Section 3.3 and Appendix B.]

An example of the way each of these four constructs might execute a particular instruction sequence on a III generation computer system is indicated in Figure 2-3. Successive instruction executions, i.e. time, are represented from left to right and the two states shown for each of the four constructs are software interpretation and direct execution. Thus, the real machine is always shown in the direct execution state while the vintual machine occasionally traps to software interpretation for certair instructions.

IYDE I YS. IYDE II YMM

The implementation requirement specifies that most VCS instructions execute cirectly on the host. It does not indicate now the VMM gains control for that subset of instructions which nust be interpreted. This may be done either by a program running on the bare host machine or by a program running under some operating system on the host machine. In the case of running under an operating system, the host operating system orimitives may be used to simplify writing the virtual machine monitor. Thus, two additional VCS categories anises

<u>Ivpe I</u>--The VMM runs on a bare machine. <u>Ivpe II</u>--The VMM runs or an extended host [53,75], under the host operating system.

In either case, the virtual machine being created is equivalent to the bare host or a related family member.

Type I (bare host) and Type II (extended host) VCS's are



and the second course when been about the second

ality in the state of the second second field of the second out some one

illustrated in Figure 2-4. In both Type I and Type II VCS, the JMM creates the VCS ervironment. However, in a Type T VCS, the a bare machine must perform the system's scheduling and VMM <u>00</u> (real) resource allocation. Thus, the Type I VMM tay include much code not specifically needed for a VCS. In a Type II VCS, the resource allocation and environment creation furctions for VM's are more clearly srlit. The operating system does the normal system resource allocation and provides a standard extended machine environment. The VMM program runs on the extended machine environment and produces a pseudo-hardware [Another interpretation of Type I and Type II environment. virtual machines is presented in Section 4.2.1

There are different advantages to be derived from using sither a Type I or Type II VCS. If an installation normally runs one particular operating system and most users would prefer to treat the system as an extended machine, then a Type II VMM may ne used to provide a virtual machine for the occasional user who may desire it. He might want to either debug system code or, say, run some foreign operating system. In such a case, the access to the system is likely to be more convenient for the numerous typical users, and, probably, more efficient since there is less overhead in running the preferred operating system. [Another approach to achieving some of these objectives is reported by Parmelee [93].]

If there are numerous alien operating systems to be run and there is no such thing as a preferred ervironment, it may make sense to run a Type I VCS. Of course, the decision to choose



÷

A DESCRIPTION OF THE OWNER OF THE

is automatical contraction and the second

^ORMANN DELEMANTERDAMONICATIVANDERNOO

Type II over Type I depends also on the existence of an operating system that is capable of supporting a Type II system. Another important factor to consider is the amount of work required to bring up either system.

Type I and Type II VCS's are two guadrants of a more general tabular structure which deserves future study. See Figure 2-5. The two rows of the table indicate the host environment on which the monitor program runs. The two columns indicate the target environment produced by the monitor. A Type I VMM runs on a hardware host to produce a hardware target, i.e. virtual machine. A Type II VMM runs on an extended host to produce a hardware target. The boxes marked EMM stand for Extended Machine Moritor. The Type I EMM is a conventional operating system. The Type II EMM runs under one operating system to produce the interface some other operating system. Type II EMM can be a generated by valuable tool for transporting software when a VCS is not used [36,45].



MACHINE ENVIRONMENTS

FIGURE 2-5

antimizer antimized

and bear indicated and the second secon

2.2 COMPARISON WITH RELATED NOTIONS

Because of the recent development of the field of computer systems architecture, and the absence of a generally usec, as well as agreed upon, terminology, it is possible that the notion of a virtual machine may be confused with certain other related ideas. In this section, some common misconceptions regarding virtual machines will be cited to provide some additional insight into the key notions involved in a virtual machine.

2.2.1 Virtual Machine vs. Virtual Memory

Virtual memory refers to an addressing system in which a logical address generated by a program, called a virtual memory address, is mapped into some other, possibly different, physical memory address. Virtual memory may be implemented in numerous days, among them simple relocation, multiple relocation, paging, segmentation, or a combination of these methods [38]. If paging or segmentation is employed, it is often the case that the amount of virtual memory exceeds actual physical memory.

Since one aspect of a real computer system is memory, the corresponding attribute of a virtual computer system is customarily called virtual memory [51]. The virtual memory need not be larger than available physical memory, although it may be, if implemented on a machine with paging. This is the case with CP-67 [9,65,85]. The virtual memory may be even mapped identically into all or part of its correspondingly addressed ohysical memory. This is the case with the proposed CP-65

[54,58], experimental IBM 360/39 [71], or Japanese work [50]. The permitted virtual memory size on method of relocation (or not) are issues which affect the flexibility of the use of virtual memory, or ease of performing system storage allocation. These issues are largely tangential to the notions involved in virtual machines.

Software Short Smith

One perticular connection between virtual machines and virtual memory concerns the interrupt control register locations that most systems have fixed in low physical core. Since the memory of a virtual machine is functionally equivalent to the memory of a real machine, virtual interrupt locations in virtual memory must have the same apparent effect for the virtual machine as the corresponding real locations have for the real machine.

2.2.2 Virtual Machine vs. Virtual Machine Time-Sharing System

Α Virtual Machine Time-Sharing System (VMTSS) is а timesharing system in which <u>each</u> uscr's interface with the system is a virtual machine [81]. It is possible to have a virtual nachine without having timesharing or a VMTSS. The experimental IBM 360/30 [71] or the Jacanese HITAC 8400 [50] are examples of virtual machines running on a non-timesharing system. The virtual 360 under UMMPS [60] is an example of a Type II virtual machine running under a (conventional) timesharing system that is not a VMTSS. Possible confusion between virtual machines and VMTSS's may be related to the success of CP-67 which is a VMTSS.

uonirolaisiskistististististististististististististaanin is, ontro

Kost a she had a she was a far and the set of the set of

2.2.3 Virtual Machine vs. Pseudo Machine

A pseudo-machine [99], also called an extended machine [53] or a user machine [75], is a composite machine produced through a combination of hardware and software, in which the machine's apparent architecture has been changed slightly to make the machine more <u>converient</u> to use. Typically, these architectural changes have taken the form of removing I/O channels and devices, and adding system calls to perform I/O and other operations. These system functions are normally invoked by transferring to some location in virtual memory, as in Multics [86,92], or issuing a Supervisor Call (SVC) or Master Mode Entry (MME), etc. address instruction with suitable code. The supervisor interprets the SVC, executes the desired function, and returns to the user.

In a virtual machine, the machine's apparent architecture is identical to a <u>real</u> machine's architecture. The visibility of all I/O channels and devices is left infact. There are no supervisory functions provided. If a machine wishes to perform I/O, it must utilize its own I/O programs. Indeed, the system toes not provide an SVC handler for the virtual machine.

Possible confusion between virtual machine and pseudo machine can be attributed to an overuse of the term "virtual machine" by certain authors. Too often "virtual machine" has been used to denote a non-hardware "user" machine [111,112].

2.2.4 Virtual Machine vs. Emulated Machine

Frulation is a technique that has been used successfully to

to the second second

map one system architecture into another different architecture [63,82,118]. Emulation at the interior decor interface (nardware interface) causes the layer zero software visibility of the native system to (almost) include visibility of the target system. Similarly, emulation at the exterior decor interface (extended machine) extends the software visibility of, say, layer 1 to include software visibility of layer 1 of the target system [51].

to you have a comparison of the second of th

Well-known examples of emulators include IEM*s 1401 on the 368/30, 7094 on the 360/65, and DOS under OS on the 378/145. In the first two examples, the target and native interior decors are vastly different. This requires the emulator to be incluiented via special additional microcode, and the processor must operate in this non-native mode.

In a virtual machine, on the other hand, the target and host are either similar or identical hardware machines. If a self-virtualizing machine has an emulator implemented on it, then it should be accessible to the virtual machine in the same way that it is to the real one. Thus, if there exists a special instruction, such as Dc Interpretive Loop (DIL) [110] which puts the machine into an emulation mode, then its execution by a virtual machine should put the virtual machine into emulation mode. [IBM has just announced this facility for VM/379 [67].]

2.3 EXAMPLES OF VIRTUAL MACHINES

Y

There are relatively few examples of virtual machines that may be cited. We will show later (Section 3.2) that this is due to the general unsuitability of contemporary designs and hence the significant requirements placed on a computer system that is to support a scitware construction of virtual machines.

2.3.1 IBM Pesearch M44/44X-- Type I FV (Family-virtualizing)

The IBM M44 [91,103] was a highly modified (II generation) IBM 7644, extended to include paging and a number of operational modes, such as Problem/Supervisor, Location Test/No Location Test, and Mapping/No Mapping. Through the use of an operating system, called MOS (Modular Operating System), it oroduced a set of virtual machines called 44X*s. IThese 44X*s are not virtual machines in the strictest sense since they are slightly different trom real 7044*s.] A 44X operating system runs on each 44X and provides the customary services found on most operating systems. The M44/44> system was very imcortant, in the history of computing, for trying out a number of innovative ideas, among them, virtual machines.

2.3.2 IBM Cambridge Scientific Center CP-40-- Type I FV

CP-40 [3,27,59], the forerunner to CP-67, was constructed on an IBM 360/40 that was specially modified through the addition of an associative memory paging box. This system was a VMTSS and supported fourteen virtual 360's. These 350's were standard

360's and did not contain any of the special modifications of the host model 40. The virtual 360's were able to run numerous 360 operating systems, including the standard OS/360 [66,84] and CMS [85]. a conversational system which was developed at Cambridge Scientific Center specifically for use with CP-40. The system suffered from the composite limitations of the 363/40's processor speed, a modest amount of real core, and a slow disk for paging. However, CP-40 did serve to demonstrate the viability of the virtual machine notion. Furthermore, it provided a VMTSS, and with CMS, a convenient conversational system for the 360 at a time when no other general purpose timesharing system for that series of machines was working. Indeed, success with this project led directly to the larger system, CP-67.

TO Part State

2.3.3 IBM Cambridge Scientific Center CP-67-- Type I FV

CP-67 [9,65,85] was begun as an experimental system in conjunction with MIT Lincoln Laboratory in January, 1967. Initially, the CP-40 system was modified slightly to support the different memory relocation system found on the 360/67. Subsequently, the entire CP-67 system was rewritten. CP-67 is a VMTSS which runs on the 360/67 and supports between thirty and forty users. It is, by far, the most widely known virtual machine system. The virtual 360's produced under CP-67 have been very successful in running a number of different 360 operating systems. These include (but are not limited to) C5/36D (in many versions), DOS, DOS-AFL, CMS, LLMFS [85]. The virtual machines have been used in telecommunications applications. The

in and the state of the service of the service of the service of the service of the state of the service of the

University of Grenoble has even run the IBM CS/ASP (Attached Support Processor) system, connecting a real 360/40 to a virtual 360 [9].

The principal limitations of the virtual machines produced by CP-67 are that a small set of programs may not work correctly. These are:

SUNDER STREET, S

(1) Programs which depend on precise execution times;

(2) Programs which depend on the relationship between processing time and input/output time,

(3) Programs which depend on precise real-time, and
(4) Programs which cortain self-modifying channel programs.

The first two restrictions are identical with those to insure compatibility between two different member processors of the 360 family [69].

2.3.4 Extension of CP-67 to support virtual 360/67-- Type I SV

In May, 1969, the author designed a slight extension to CP~67 to support a relatively efficient inclementation of a virtual 360/67 (with virtual relocation) [54]. With slight modifications, this design of a virtual 360/67 was imclemented in the fall of 1969 by F. Furtek. Independently of this work, A. Auroux at the IBM Cambridge Scientific Center also designed a virtual 360/67 extension to CP-67 [95]. The two designs are very similar and the Auroux implementation has become part of the standard (IFM distributed) CP-67 system.

The virtual 360/67 has been limited to a 24-bit addressing

mode single processor machine. With this capability, various 360/67 operating systems have been run, including CP-67 and TSS/360 [95]. That is, CP-67 has been run under itself. In fact, this chain has been tested to a depth of at least four. The principal additional limitation on the operation of the vintual 360/67 over a standard vintual 360 (under CP-67) is that programs that dynamically alter the contents of segment or page tables may not execute as intended. The results may ar may not be identical with a real 360/67. In a real machine, the contents of the associative registers may affect the result and hence results may also be unpredictable.

2.3.5 Virtual 360 under UMMPS-- Type II FV

A A CONTRACTOR OF A CONTRACT O

The virtual 360 under UMMPS [4,60] is a Type II virtual machine system. UMMPS is a conventional multicrogramming system which was written at the University of Michigan to run on a dual processor 360/67. The University of British Columbia has made a few modifications to UMMPS to support a virtual 360. These modifications have taken the form of some additional supervisor call facilities to perform some crucial operations in dispatching the virtual machine. Apparent exceptional conditions occurring in the virtual machine are reflected by the UMMPS supervisor back to a user program which performs, for example, the simulation of privileged instructions. Virtual (standard) 360°s are the only virtual machines supported by this system, although other user interfaces besides the virtual machine interface are provided (such as MTS on a conventional batch system). Extensions to

and a start of the second second second second second second second and the second second second second second

こうぞうがい キャー しんしてい ちがん たいたいきいたいたい ひょうちょうかい ちょうちょう ちょう スカンスカンス ちょう

d the distribution of the desired for the second of the second second second second second second second second

UMMPS to support virtual 36(/67's as well as the standard 360's are being designed.

2.3.6 Japanese Electrotechnical Lab. HITAC-8400-- Type II FV

The HITAC-8400 is the Japanese manufactured version of the RCA Spectra 73/45. It is a conventional third generation computer without a memory mapping system, similar to the IRM System/360. As an aid in debugging ETSS [49], a new time-sharing system for the HITAC-840C, a Type II virtual machine was constructed on the existing, vendor-supplied operating system, TOS/TOOS [50]. The virtual machine system is very similar to the virtual 360 constructed on UMMPS. One interesting ascect of this effort was the attempt at simulating various virtual I/O devices that did not have exact physical counterparts (such as a disclay scope). This was done, of course, since the virtual machine was being used entirely to debug supervisory code for ETSS and not for running production Jobs under it.

2.3.7 Hardware Modified IBH 361/30-* Type I FV

A virtual machine system has been constructed on an experimental 363/30 through a combination of software and nardware, i.e. microorcyramming modifications [71]. These modifications, which have been introduced in order to simplify virtual machine construction, are used to implement a special "monitor" mode, and to relocate the interrupt control area from the regular control program area in physical page 0. The system is not a VMISS and only supports one virtual machine. Its

and the second of the second second

aringiple use is in monitoring and evaluating the performance of existing operating systems such as Cperating System/360 (OS/360), and the Basic, Disk, and Tape Operating Systems (EOS, DCS, TOS/360).

2.3.8 MIT Project MAC PEP-10 ITS-- Type II SV HVM

As will be seen in Chapter 3, it is not cossible to implement virtual machines on the DEC PDP-10 using corventional third generation software techniques. Recently, the author and S. W. Galley have introduced the notion of a hybrid virtual machine for the FDF-19 [56]. With this technique, a11 executive-mode instructions are interpreted while all user-mode instructions execute directly. The MIT Project MAC Dynamic Modeling-Computer Graphics PDP-10 has provided an ideal environment for this implementation. The ITS (Incompatible Timesharing System) provides the requisite features to support a Type II (extended machine host) VCS. Furthermore, the hardware configuration is plessed with sufficient core and, at times, enough CPU cycles to make it possible to run the HVM. [See discussion in Appendix C.]

2.3.9 Grenoble Monitor System for Standard 360/40

and the second second

GMS, Grenoble Monitor System, [58] was designed by the IBM Srenoble Scientific Center to provide multiple 360's on a standard IBM 360, i.e. one without relocation. The major objective of the system was the concurrent support of a limited number of virtual machines, each running CMS, the Cambridge

Monitor System. Since a standard 360 does not provide a memory relocation system, the store <u>and fetch</u> memory protection system must be used in virtual machine construction. ISse Appendix B.J However, because of difficulties in installing fetch-protect on a European model 40, the actual implementation had to compromise the pure 360 hardware definition of the virtual machines. This led to a system in which modifications were made to CMS to allow it to run in this "enhanced" environment.

2.3.10 IEM VM/370

In July, 1972, IBM announced a virtual machine facility for various models of the System/370 [67]. This system seems to be a direct descendant of CP-67. When delivered, VM/370 will be the first formally-supported, commercially-offerred VCS available from any manufacturer.

2.4 RELATED LITERATURE

The early literature on virtual machines dealt almost exclusively with the implementation mechanisms, policies, or applications of ore particular virtual computer system. The treatment of policies, e.g., scheduling algorithms. **D52** replacement algorithms, in these papers involves largely the same considerations in non-virtual machine timesharing 35 or nultiprogramming systems. Therefore, they will not be reviewed nere. Many of these papers, particularly those devoted to remory management appear in the annotated bibliography of Parmelee et al [95]. Since the tnesis is concerned with developing architectural principles and mechanisms, we will not review the virtual machine applications papers either. However, some of the more significant papers are listed in the bibliography.

The virtual machine implementation and machanism papers go vack to an early unpublished paper by Sayre [101,103] which describes the IBM M44/44X work and its implications. Perhaps the first virtual machine report to get wider circulation was the description of GP-46 produced by the IBM Cambridge Scientific Center in 1966 [3].

With the arrival of CP-67, a number of papers were published which describe the mechanisms used to create virtual machines. These papers include Field [47] and Auroux and Hans (in French) [9] which are the most complete. Later papers by IOM authors, a.g. Meyer and Seawright [85], repeat much of the same published material. Two early papers deserve particular note. Fuchi's paper [50] reports on the implementation of the first Type II VMM and also gives insightful suggestions for hardware "improvements" to facilitate virtual macrine construction. Keefe's paper [71] reports on some hardware modifications attempted with an IBM 360/33 in order to simplify production of a non-self-virtualizing /MM.

Among the author's early papers are the first capers that discussed self-virtualizing systems and the development of an deciding which machines empirical basis for are self-virtualizing. "Virtual Machine Systems" [48] recorts on the design of a virtual 360/67 under CP-67 and a virtual 360/65 under "CP-65" (which runs on a standard 360/65). This same report and "Hardware Requirements for Virtual Fachine Systems" [52] contain the first published analyses of the problems involved in implementing a VMM on a specific class, i.e. third generation machines, rather than an individual computer system.

Recently, there has been increased interest in VCS"s. At the (September) 1971 IEEE International Computer Society Conference, the author organized a session of papers [53,94,113] and discussion devoted solely to VCS's. One of those papers, written by the author, "Virtual Machines: Semantics and Examples" [53] proposed some of the terminology introduced in Chapter 2.

At the 1972 ACM International Computing Symposium at Venice, three papers specifically related to VCS's appeared. In "Virtual Input/Cutput in a Virtual Environment" [8], Ancilotti, Cavina and Lijtmaer of Pisa investigate the possibility of allowing an I/O

Page 40

· WERE AND THE REPORT OF A CONTRACT OF A CON

channel to execute a program allocated in virtual memory. "Is Supervisor-state Necessary?" [76] by Lauer and Snow of the University of Newcastle-upon-Tyne is a particularly insightful and well-written paper proposing the organization of a self-virtualizing computer system based upon a relocation and bounds form of address relocation and no supervisor state. As will be discussed in Section 4.8, this design is a special sub-case of the hardware virtualizer of Section 4.5 with f=R-Band $\phi=$ identity map.

The third paper was "Virtualizeable Architectures" [51] by U.O. Gagliardi and the author. This paper was the first paper to suggest the need for firmware support for virtual machines in a complex computing environment. The design that emerged, called the Venice Proposal (VP) in this thesis, features a hardware Virtual Machine Identifier (VMID) Register to specify the active virtual machine. It was the work with U.O. Gagliardi on the VP that led the author directly to the more general model of a VCS oresented in Section 4.2. In Section 4.4, the VP is presented and interpreted as a special case of the VCS model.

A number of other papers, not specifically related to VCS*s, have affected the author*s thinking during the course of this research. Some of these papers discuss hierarchical operating systems, layered protection structures, or deferred resource binding. The early paper by Dennis and Van Horn [1,40] describes an hierarchical operating system and suggests primitives for interprocess control. The paper by Dijkstra [41] describes a view of a system seen as a layered structure. The paper by

Schroeder and Saltzer [106] proposes a hardware mechanism sufficient to support a layered system structure. Evans and LeClerc [46,77] describe a hardware addressing structure in which addresses, e.g. segment numbers, are assigned relative to each individual procedure call. Finally, Schell's doctoral thesis [104] discusses the relationship between logical and physical rescurces and the mechanisms for electrical binding.

was a star proved by

CALLER STATISTICS STATISTICS

TANK STREET

in subset beriod and the surption initial blead is stabilities a subscript subscript of

ないまたいしょうないたいないないないないないないというないないないというないないということ

CHAPTER 3.

PRINCIFLES FOR III GENERATION VIRTUAL COMPUTER SYSTEMS Inadecuacy of Current Designs

3.0 PLAN OF CHAPTER 3

Chapter 3 develops principles for third generation virtual computer systems. The approach adopted is somewhat empirical and relies on a scrutiny of existing III generation hardware and software systems. The chapter is divided into four subject areas: (1) Characteristics of III generation hardware, (2) Hardware requirements for third generation virtual computer systems, (3) Software requirements for Type II third generation VCS*s, and (4) Conclusion.

In the first section, we summarize relevant architectural characteristics of III generation systems.

Section 3.2 generalizes the virtual machine construction used in CP-67 to derive a set of empirical hardware requirements generation architecture is if a third for determining virtualizable. [Appendix A provides a brief tutorial on CP-67.] rules are applied to a number of representative III The generation machines and the results are summarized in a table. B provides the detailed analysis of these case [Appendix studies.] In Section 3.3 we introduce the hybrid virtual machine develop its of less stringent hardware (HVM) and set requirements. Finally in Section 3.4, modest. ad hoc

are a server strates of the server of the server of the strates of the server and the server and the server server server of the server of the

ander er stand for the stand stand stand and a standard of the stand of the stand stand standard and the stand

"improvements" are considered to make more third generation machines virtualizable.

Section 3.5 develops a series of empirical operating system requirements for determining if an operating system on a third generation machine will support a Type II (extended machine host) VMM. The rules are applied to a number of representative III generation operating systems and the results are summarized in a table. [Appendix C provides the detailed analysis of these case stucies.] In Section 3.6, we propose an example of a set of idealized operating system primitives to support Type II virtual machines.

Section 3.7 reviews some of the results of the chapter and provides some perspective of how this work relates to Chapter 4.

3.1 CHARACTERISTICS OF III GENERATION COMPUTER SYSTEMS

One of the most significant architectural innovations of III meneration computer systems is the existence of two processor modes of operation. These two modes, variously called supervisor and problem, executive and user, or master and slave provide two degrees of privilege for programs. Thus, ocerating system programs, when executing in master mode, may typically invoke certain special instructions, while user programs, operating in slave mode, may not. These instructions normally control such critical system features as the input/output facilities and the setting of privilege itself. Attempts by programs in slave mode to issue privileged instructions are normally prevented in one of several ways.

Another architectural characteristic of III generation systems often. the presence of a supervisory call is, instruction. Since a user program is prohibited from issuing privileged instructions, the operating system normally provides a means for users to call upon it for these services. Entry to the operating system is usually via a special instruction, such as Supervisor Call (SVC, BRM, UUO, MME) which traps to a particular location in the monitor. The monitor checks the request. performs (if permitted) the desired function and returns to the user program.

An additional characteristic of the III generation computer system concerns the means of addressing memory. Usually, but not always, some kind of relocation and/or protection system is

provided. The memory relocation may take the form of simple relocation and bounds (R-B), paging, or even segmentation and paging. Most of these systems provide an absolute addressing mode when the memory relocation system is disabled. This usually occurs when master mode is entered.

III generation I/O systems usually communicate with the CPU via asynchronous interrupts which are affected by an interrupt control area in low physical core. Interval timers are set and notify the CPU similarly.

Denning [37] has recently delineated the extent of III generation systems by uncovering the existence of a "Late Third Generation." Since his distinction between III and Late III is often based on facilities of the operating system software, we need not distinguish them. Virtualization is concerned with the software visibility of the interior decor, not of any operating system features. Later in Chapter 4, we take up the likely characteristics of IV generation architectures. Since many of these same "Late Third Generation" facilities, e.g., interprocess communication, are now present in the IV generation interior decor, virtualization must consider them. [See Section 4.1.]

3.2 EMPIRICAL HARDWARE REQUIREMENTS

CP-67 has suggested a general approach for constructing virtual machines on other conventional III generation systems. In this section we will explore the application of the CP-67 virtual machine construction technique and see for which III generation systems it is appropriate. Appendix A provides a tutorial on CP-67 and describes the maps which it employs. The key point in the simulation of a (III generation) virtual processor is that since the host and virtual machine are identical (or similar), the instructions of the virtual machine may be executed directly on the host. However, because of the virtual machine mapping construction it is necessary to prevent certain of the instructions from executing directly on the host. These instructions, if executed directly by a VM can cause serious difficulties in interpretation or control. [See Appendix A.] Any instruction whose direct execution cannot be tolerated is termed a <u>sensitive</u> instruction. Then the key to implementing a virtual machine on III generation systems is to provide complete functional equivalence with a real machine without allowing the direct execution of sensitive instructions.

The scheme adopted relies on a software mapping and permits only the VMM to run in supervisor state. The virtual machine, itself, is run in problem state. Therefore, if the instructions that are considered sensitive are also crivileged, their attempted execution by the virtual machine (in problem state) will cause a trap to the VMM. If there are some instructions

that are considered sensitive but are not privileged instructions, then it may be necessary to resort to massive interpretation in order to preserve the integrity of the virtual machine.

What are the instructions that are corsidered sensitive in a virtual machine system? Certainly any instruction that attempts to change the mode of the virtual machine. As stated above, in order to prevent a virtual machine from gaining control of the host machine, the virtual machine must be run in problem state. Yoreover the virtual machine should not be permitted to put the host machine into supervisor state in such a way as to allow the virtual machine to execute any privileged instructions. In order to maintain a compatibility between the virtual machine and its real counterpart, the VMM must provide a virtual supervisor state. See Figure 3-1. In this state, the virtual machine's privileged instructions are simulated by the VMM. In addition, the virtual machine should not be able to change its state from virtual supervisor to virtual problem state without a privileged instruction. Since both of these states are actually run on the nost machine as problem state, the VMM must be informed directly or it would have no way of knowing that a virtual state change has occurred. Furthermore, it is not sufficient to merely prevent the virtual machine from changing the state of the host machine, it must also be prevented from referencing it. Since the host machine will be in problem state when the virtual machine thinks it is in supervisor state, the answer to the guestion "What state am I in?" may give an inacpropriate result.



I GENERATION VIRTUAL PROCESSOR MAP

Set Charles

£

FIGURE 3-1

Consequently the instructions which reference state information as well as those that actually change it must be privileged instructions. In machines in which this is not true, if the VMM cannot perform some kind of preminterpretation, a virtual state change might occur without its knowledge. As will be seen in Section 3.3, the hybrid virtual machine, HVM, provides another solution to this dilemma.

A CONTRACTOR OF CONTRACTOR

Another problem arising from the fact that virtual machines in virtual supervisor state must, in fact, be run in physical problem state, concerns the interpretation on lack of interpretation of certain bits in the instruction wo.J. Certain machines use an additional bit in the instruction word only when in supervisor state. Other machines use additional bits in the address portion of the instruction. These differences in instruction execution in problem and supervisor state may be difficult to resolve without resorting to instructionby-instruction simulation. [See Acpendix P.]

In addition to changing or referencing the state of the host machine, the virtual machine must be prevented from tampering with or looking at certain sensitive registers and core locations. For example, because the VMM must keep track of the real interval timer for all machines, the virtual machine must be prevented from referencing the real clock. Since in some machines the clock is a register or a fixed location in core, either the instruction that references it must be privileged or some method must be found for protecting it through the use of the protection or relocation system. In order to preserve the

Page 50

ちちんちんかいたいないたいでいたい いいんちちんきん ちんち

virtual machine notion, the VMM must maintain a virtual clock for the virtual machine in some other part of core. Some other sensitive dedicated core locations are the various interrupt registers that are automatically accessed by the hardware in the avent of an interrupt. Since the contents of the interrupt locations are used automatically by the hardware, their values may be considered part of the specification of the state of the nachine. Therefore, the instructions that access these values are sensitive and should be privileged. If they are not, some other means for protection and alerting must be employed.

Other sensitive instructions are those involving the storage protection system and address relocation system. In order to protect the supervisor and (possibly) other virtual machines from an errant virtual machine, the virtual machine may never have access to any location that is not in its virtual memory. This may be accomplished by making all storage protection instructions privileged, or permitting their effect only within the acdress space defined by the address relocation system. In there is no relocation system and absolute addresses are generated, than the storage protection mechanism must be used to set the virtual nemory size, protect dedicated lower core locations from access, and protect the supervisor. In this case, the storage protection instruction must be privileged to permit the supervisor to manage the protection keys for the host machine. If the host machine has a storage relocation system (either relocation registers, paging, segmentation, or a combination of these) ther it must be insulated from the action of the virtual machine by making all

instructions referencing the relocation system crivileged. If it is desired to have a virtual relocation system which makes use of the host machine's relocation hardware then the effect of relocation setting instructions must be situlated. This simulation includes translating virtual memory locations to real core addresses.

Input-output instructions are always sensitive. It is necessary that they be executed only by the supervisor in order to isolate core from secondary storage, protect secondary storage from virtual machines, and enable the efficient scheduling of I/O tasks for the host machine. Furthermore, the YMM may be recuired to translate virtual device addresses into their real equivalents before I/O transmission may take place. If the notification of the I/C instruction does not occur before the I/O operation is started, data may be lost or physical movement may be needlessly initiated. Thus it is important to trap the I/C instruction as soor as possible. A privileged I/O instruction is the best solution.

Thus, we are able to state the following empirical hardware rules which govern whether a virtual machine monitor may be implemented on a conventional third generation computer system.
Empirical Hardware Requirements

(1) The method of instruction execution of non-privileged instructions in both supervisor and problem state must be roughly equivalent for a large subset of the instruction repertoire.

(2) A method of protecting the supervisor (and any other virtual machine, if there is multicrogramming) from the active virtual machine must be evailable. This may be ecomplished, for example, through a protection system or an address translation system.

(3) A method of automatically signalling the supervisor when the virtual machine attempts to execute a sensitive instruction must be available. The trap must not cause unrecoverable errors. It must then he possible for the supervisor to simulate the effect of the instruction. Sensitive instructions includes

a. These instructions which alter or every the state of the machine, e.g. "Is it in supervisor state or problem state?", "Is it in relocate mode or not?"

b. Those instructions which alter or query the state of the machine's reserved registers and core locations.

c. Those instructions which reference the storage protection mechanism, the memory system, or anything else that is specifically used by the VMM in Suilding and managing the virtual mechine.

d. Any I/O instruction.

のためたいいいいいで、たいにないいいいいいいいとうになったないです。

がたいたんというというためにないたという

Application of these hardware rules to a number of familiar computer systems is treated in Appendix B. The results of these case studies are summarized in the table of Figure 3-2. Except for the IEM 363 family of machines, most other contemporary III generation systems are not virtualizable.

are a structure of the state of the

いっていたいないないないないないないできたいないないないない、たち、たちのい

- ECTANOVICAL CARGE SA

Machine	Hardware Rule Violations
IBM 360/67	none
IBM 360/65	none
HITAC 8400	none
IBM 360/85	none
DGC NOVA	none
DDP-516	3, 3a, 3b
GE 635	3a, 3c
GE 655	30
Multidata A	3
XDS 940	1
PDP-10	3a
BBN TENEX	3a

Note: In some case, a machine which violates hardware virtualization rules may still be able to support an HVM. [See Section 3.3.]

II GENERATION VIRTUAL MACHINE CASE STUDIES-SUMMARY FIGURE 3-2

こうちょういうそうないないないないないないないないないないないであるというできょう

3.3 HYBRID VIRTUAL MACHINES AND PEQUIREMENTS

As we have illustrated in Figure 3-2 [also Appendix B] most III generation machines .e not virtualizable bacause of violations of the empirical rules. Thus, it is desirable to define another construct which has the advantages of virtual machines, e.g. functional advivalence to real machines, but is feasible on a larger class of systems then (conventional) virtual machines are. Furthermore this construct must be accomplished without the disadvantages of the complete software interpreter machine. The hybric virtual machine, HVM, defined in Chapter 2 meets both of these objectives.

An HVM is functionally equivalent to a real machine. All instructions issued within the most privileged layer of the HVM are software interpreted while all non-privileged-layer instructions execute directly. This leads to an imclementation on III generation syste 3 in which virtual problem state is mapped into physical problem state but, unlike the conventional III generation virtual machine, the HVM virtual supervisor state is not also mapped into physical problem state. See Figure 3-3.

Without mode-mapping and direct execution of the non-sensitive supervisor state code (as in the convertional III generation VCS), the HVM streamlines the empirical hardware requirements. No tonger need one be concerned that all non-privileged instructions are insensitive to the mode map. Thus, the HVM eliminates empirical Pule 1 from its set of hardware requirements.

Attempts by the virtual machine to enter supervisor state are directed to the VMM. Since the VMM maintains the virtual pure supervisor state as software construct, a weaker form of Rule 3a is all that is needed for HVM. In particular, we must identify only those instructions which enter supervisor state. Thus, since supervisor state is not mapped, a mode query need not be either and the instruction is not sensitive. Finally, since ≅xecution in virtual supervisor state is under the control of an instruction-by-instruction interpreter, the return to problem state instruction is not sensitive. The interpreter itself can read the instruction and observe that a virtual state change has occurred.

As illustrated in Figure 3-2 [also Appendix 3], a number of III generation machines fail the virtual machine empirical hardware requirements either because of Rule 1 or Rule 3a. Thus, they are candidates for an HVM.



illingen und der sichten der sichten inter sichten sichten werden der der der der sichten der sichten der sich

3.4 AC HOC HARDWARE "IMPROVEMENTS"

In addition to the HVM, another approach to constructing virtual machines on third generation architectures might involve promosing some ad hoc "improvements" to permit more systems to satisfy the empirical requirements (of Section 3.2). We term these improvements "ac hoc" since they respond to specific hardware deficiencies which arise in the use of the conventional III generation software virtual machine techniques. Only in Chapter 4 do we consider more major architectural changes that follow from a different view of virtual machines.

The ad hoc improvements we explore and addressed predominantly to machines which violate empirical rule 1 or (one or more subparts of) empirical rule 3 [Section 3.2]. We will assume that the machines discussed in this section do not trap so as to cause unrecoverable errors. Thus, for example, we will not examine the DDP-516 which traps an entire instruction late iAppendix B]. Rather, the difficulties we consider are either machines in which (1) Sensitive instructions is not trap when necessary, or (2) Undesirable side effects are caused in order to guarantee some trap.

An example of difficulty 1 is a machine in which some sensitive instruction is not privileged and so does not trap when execution is attempted in problem state. The PDP-10 JRSTF instruction is such an example [Appendix B]. Another case of difficulty 1 is a machine in which privileged instructions execute as a NOP in problem state. The Multigata Model A or GE

535 demonstrate this difficulty [Appendix B].

An example of difficulty 2 is a machine in which many "inrocent bystander" core locations must be protected in order to be able to protect some other sensitive location because the nachine's relocation or protection mechanism is too coarse or too inflexible. The 360/65 or FITAC 8400 must protect an entire 2K byte core block just in order to make several hundred sensitive locations inaccessible [Appendix B].

3.4.1 Difficulty One-- Non-trapping Sensitive Instructions

Difficulty one may be restated as a problet: Choose the set of privileged instructions so as to include all sensitive instructions which may not be "protected" by some other nechanism, £.g. protection or relocation system. The solution to this problem must be formulated in such a way that if, during the course of VMM design, it is discovered that an unanticipated instruction is sensitive, then the architecture will still be virtualizable.

A valid solution is to permit the supervisor to make any instruction privileged by executing a special <u>Irap Cn Cpcode</u> (TOO) instruction. We assume that initially there are no privileged instructions and that the VMM is in control. The VMM initializes the system by executing a TOO instruction for each opcode, including TCO, that is to be made privileged. At the time of machine design, there is no need to make a commitment on which instructions are to be privileged. When the VMM has been written (and debugged) and it is known which instructions are

1. W. A. A. S.

The second state and the preside of a second state and a second and a second and a

sensitive, then the appropriate initialization may be added to make those instructions privileged. Furthermore, if the computer system is to be used for nor-virtual machine tasks, it may be desirable to return certain instructions to a non-privileged state.

It is necessary to show that the TOO instruction has the desired effect and that, furthermore, the effect of TOO instructions may be simulated for virtual machines. For the purposes of this discussion, we can assume that an occode is the operand of a TOO instruction.

Thus, there are two new instructions introduced into the computer:

T00-- described above

UNTCO-- which turns off trapping for the operand opcode and sets a condition code indicating whether trapping was previously on.

Figure 3-4 gives a simplified example (of one possible implementation) in connection with the IEM 369. The operand field is assumed to be a bits long. The TRAP register is 2**m The trap register is shown as a bit mask. This is just bits. one possible implementation. For example, it could be an associative memory.] Execution of the instruction "TCO o", where o is a valid opcode, causes bit p of the TPAP mask to be set to 1. Subsequent execution (ir problem state) of any instruction whose bit is turned on in the trap register causes a trap to occur. Thus, sensitive instructions may be made orivileged.

The action of the TOO and UNTOO instructions may be



ないこれにからないのでいていていたと

ないないなどの言語を行きためのであるという

and the second states and a second states and a second states and a second states and a second s

÷

TRAP REGISTER AND TOO INSTRUCTION FIGURE 3-4

simulated for virtual machines. The VMM keeps a copy of the virtual machine's trap register, i.e. virtual trap register. Attempted execution, by the virtual machine (running in problem state, as before) of the TOO or UNTOO instruction causes a trap to the VMM. The supervisor then simulates the effect on the virtual trap register. Before the virtual machine is dispatched in (virtual) problem state, the virtual trap register is OR*ed into the (real) trap register. Figure 3-5 illustrates how the virtual trap register is supported.

3.4.2 Difficulty Two-- Undesirable Protection Side-effects

The problem considered here is largely one of inelegance, awkwardness, or unreasonable performance degradation caused by the side effects of mechanisms required in order to limit access to certain sensitive (core) locations. Most computer systems have a rumber of sensitive locations, often termed the interrupt control area [69]. Since the interrupt control area is located in physical core, it may be tampered with by non-privileged instructions. Consequently, it is necessary (as brought out in Section 3.2) to protect these physical locations by means of the protection or relocation hardware of the host. Unfortunately, the protection or relocation system is often too "coarse" to perform this task neatly and, as a result, other non-sensitive locations may be protected as well.

There are roughly two points of view that may be adopted to avoid cifficulty two. In the first solution, we alter the machine design to provide a finer grain protection mechanism.

and an the star of the second

TR_{vm} --- VM trap register

TR_{vmm} --- VMM trap register

TR_s --- traps for sensitive instructions

to run virtual machine ----

TR_{vmm} : = TR_{vm} V TR_s

trap on opcode p ---

	p∈TR _s	p∉TR _s
peTR _{vm}	SIMULATE TRAP	SIMULATE TRAP
p∉TR _{vm}	SIMULATE INSTRUCTION	EXECUTE NORMALLY

VIRTUAL MACHINE SUPPORT FOR TRAP REGISTER

r'IGURE 3-5

the second of the second of

langerikter vir et som såtter en stanen streket staden at en koktation för et an en sen et som staden som som s

ty a strate a strate of the strate of the stratest stratest with the association of the stratest stratest stratest of the stratest stratest

うちょうちょうかい しんない ちょうちょう ちょうちょう

Instead of protecting core or the basis of 2K blocks (as in the 360/65), we can do it or a word basis. We introduce a special new instruction Trap On Location, TCL, which may be used analogously with TOC (above) to initialize the sensitive locations for a virtual machine. After a location has been TOLed, an attempt to reference it in problem state causes a trap. There are a number of different ways in wrich TOL may be implemented. For example, an associative memory can be used to nold locations that have been TOLed. Or, each word of memory may be associated with one bit which indicates if it causes a trap on reference. This second inclementation has been used in the design of the ISPL machine at the RAND Corporation (11,12). The first implementation with a one word (associative) memory has been used in the IBM M44 [S1] and the MIT Art ficial Intelligence Laboratory PDP-10 [61,62] to simulate address stoc switches.

and the second second

のためのないないないないのであるというないないないないないないないないないないであっていたのです。

In the second solution, we remove the sensitive locations from main memory and place them in some unaddressable or differently addressable memory. In order to access this special memory, we reduire a rew privileged instruction. See Figure 3-6. In some machines, e.g. 360/67, 370, this type of auxiliary memory is sometimes called control memory or control registers. There are several good reasons for adopting this solution. First associating sensitive locations with privileged instructions which access them provides a homogeneous mechanism for virtual machine tracping. Another reason for adopting solution two is that some of the locations in the interrupt control area are an infimate part of the description of the state of the processor.

CARLES CONTRACTOR OF THE STATE OF THE STATE



SENSITIVE LOCATIONS IN UNADDRESSABLE MEMORY

FIGURE 3-6

Thus, they should be associated with the processor (on a more individual basis) as the PSW is.

Solution two as an ad hoc improvement was recognized by Fuchi et al ISO3 in their work with a virtual machine for the HITAC 8433.

> We have two proposals for further hardware One is that memory protection should improvments. flexible to inclush read protection. becore *more* The other is that the CAW storige area and the timer should be moved from the main cone memory to The the scratchrad memory. latter takes some trouble in simulator construction and is inelegant from the logical designer's point of view [52].

> If all of these requirements are met, we could build a more natural simulator ... [50].

As mentioned above, the System/370 incorporates some aspects of solution two. Some of the new, sensitive locations have been nade "control registers" and are accessible only via privileged instructions. The new channel and CPU identification numbers are invisible and accessible only via privileged instructions. The new 37% clock is invisible and can be stored into crly with а privileged instruction. Unforturately, it таy bε read by a non-privileged instruction.

ANALY CONTRACTOR STORES

3.5 TIL GENERATION EMPIRICAL SCETWARE REDUIREMENTS

In Section 3.2, we developed the emcirical hardware requirements for Type I virtual machines. In this section, we add the empirical software requirements that must be introduced for Type II (extended machine host) virtual machines. Recall that a Type II virtual machine system [Chapter 2] is one in which the virtual machine monitor runs under a host operating system rather than on a bare hardware host machine.

Thus in Figure 3-7, hardware events such as the trap caused by an attempt to execure a privileged instruction in the supervisor state of the virtual machine [physical problem state], get passed to the host operating system. This pperating system then gives control to the VMM (under it) which may perform privileged instruction simulation for the virtual machine.

A Type II virtual machine may be desirable in a number of circumstances as pointed out in Chapter 2. Because a Type II virtual machine supervisor has a (possibly) rich set of supervisory services (SVC*s) available to it countesy of the host operating system, it often requires less effort to implement a Type II system than a Type I system. At the same time, however, the SVC*s and the SVC mechanism must be of a form that does not make it impossible to implement a Type II virtual machine.

Thus the dilemma. The host operating system must provide those supervisory services (primitives) that make efficient "ne implementation of a Type II virtual computer system, without making it impossible.







FIGURE 3-7

Since the empirical software requirements will complement the empirical hardware requirements, we begin our search with another look at the requirements of Section 3.2. Empirical hardware rule 1 still remains valid. The introduction of the additional structure ' an operating system between the virtual machine, VMM, and the hardware must do nothing to invalidate rule 1.

The Type II system version of rule 2 is that there must be a primitive available to protect the virtual machine monitor from the active virtual machine. Since the virtual machine monitor will be running in an insulated environment that, likely, will prevent it from utilizing the protection or relocation hardware directly, there must be a system primitive that it can invoke to obtain the desired effect.

Rule 3 must be modified to indicate that when the virtual machine trans, the operating system supervisor directs the signal to the VMM for procession. Since the operating system and VMM are no longer a single unit, there must be a method of communicating to the operating system supervisor that the virtual machine is a particular subprocess of the VMM and that all of the appropriate virtual machine traps are to be directed to the VMM. The mechanism for setting up the subprocess and indicating where the signal is to go may be accomplished either by a format subprocess primitive, or it may be an informal procedure that is made up of a number of the other primitives. It is not sufficient for the virtual machine to mere'y signal the VMM. As

Page 69

Section Sectio

an and a substitution of the substitution

pperating system) must not cause unrecoverable errors. It must be possible for the VMM to simulate the effect of the operation. Thus, it is not satisfactory to set up a subcrocess in which anything that appears anomalous is considered to be an error and the subprocess is destroyed. This is actually fairly common among systems.

Since primitives (SVC*s) alter or query the state of the machine, they are sensitive according to hardware requirement 3a. Thus, their attempted execution by the virtual machine is forbidden and must cause a trap to the virtual machine supervisor. Consequently, the operating system requires a orimitive that may be used to turn on or off the SVC*s for a subprocess. In the case of a virtual machine, the VMW would use this primitive to shut-off all SVC*s. "(is, of course, includes the SVC*s to turn on and off the SVC*s.

Note that in a well designed 'user machine' [42,751, system primitives replace the hardware instructions to accomplish the sensitive instructions a,b,c, and d of Section 3.2. Thus, shutting off the SVC's makes these sensitive instructions (along with all other SVC's) trap in the 'user machine.'

Thus, we are obte to state the following empirical software rules which govern whether a Type II (extended machine host) VCS may be implemented under the operating system of a virtualizable third generation computer system.

Empirical Software Requirements

(S1) The method of instruction execution of non-privilegec instructions in both supervisor and problem state must remain equivalent for a large subset of the instruction repertoire.

(S2) A primitive for protecting the virtual machine monitor (and any other virtual machine if there is multiprogramming under this VMM) from the active virtual machine must be available. This may be accomplished, for example, through a protection primitive, address translation primitive, or some more general subprocess primitive.

to tell the operating system (S3)Δ primitive supervisor to signal the virtual machine monitor when the virtual machine attempts to execute a sensitive instruction must be available. The signal to the VMM must not cause unrecoverable errors. It must then be possible for the virtual machine monitor to simulate ~ e sensitive the effect the instruction. of instructions are the same as for the hardware requirements.

Application of these software rules to several computer systems is treated in Appendix C. The results of these case studies are summarized in Figure 3-8. Except for the ITS, Incompatible Timesharing System, which provided the necessary crimitives, the poerating systems examined either failed the empirical rules or required ad hoc or specialized solutions.

Operating System	Software Rule Difficulties
HITAC 8400 TOS/TDOS	Ad hoc solution $(S_2 + S_3)$
IBM 360/67 UMMPS	Specialized solution (S ₂ +S ₃) (SVC SWPTRA)
IBM 360 OS/360	Violates rules (S ₂ +S ₃)
DEC PDP-10 ITS	Satisfies rules
IBM 360 CMS	Specialized solutions are possible.

States and the states of the states

III GENERATION TYPE II CASE STUDIES-SUMMARY FIGURE 3-8

The Manufacture of the second and the second second second

3.6 SOFTWARE F' IMITIVES FOR TYPE IT VCS*S

As we have illustrated in Figure 3-8 [also Appendix C], most III generation operating systems are not designed to support Type II VCS's. The systems which do have largely been reworked to do so. From the experience with the ITS Type II HVM [Appendix C] it becomes possible to indicate the kind of operating system primitives that would simplify the support of Type II third generation virtual machines. To avoid the usual problems with asynchronous I/O, we will discuss the primitives only with respect to the CPU and memory of the virtual machine.

The ITS UUO's, for example, provide facilities slightly different from those needed in supporting virtual machines. Since the goal of the ITS primitives is to succort ar extended machine environment, the primitives provide greater flexibility thar is needed for virtual machines. For example, in ITS a superior may destroy all of the inferiors in the sub-tree below In a virtual machine environment, the VMM need only be able it. to destroy an immediate inferior, since that is the extent of the sub-tree. Support of additional levels of virtual machines is transparent to the VMM and resides with the second clay of ITS and the second VMM which are running in the level one virtual machine. The copy of ITS in the real machine does not know about the level two virtual machine and only sees a level one machine. Hence, the primitive is more general than is specifically needed for VMM construction. See Figure 3-9.

Other primitives, such as the Dennis-Van Horn orimitives

"A" with frankers







Typical Multi-level process tree

Type I Two-level Virtual Machine Tree.

The VMM only sees one level beneath it.

PROCESS TREES FIGURE 3-9

[40] and Lampson [74], are concerned with supporting dynamically changing relationships between processes. This feature may not be recessary with virtual machines.

On the other hand, operating system primitives usually do not directly support such intimate hardware features as the relocation system. Thus, for example UMMPS had to introduce SWPTRA to switch segment numbers.

A set of possible virtual machine primitives is proposed below. The intent of this set is to indicate the simplicity of a VMM that might be programmed using them. This example omits the consideration of I/C or time.

(1) i = virtualmachine loc,], k, ... ;

This primitive creates an inferior virtual machine process. The parameter loc indicates the location to transfer to on a trao from an inferior and i is the virtual machine's identifier. The other parameters might indicate virtual memory size, relocation or protection information, virtual processor instruction counter and register values, maximum cuantum, etc.

(2) <u>dispatch</u> i;

Activate or reactivate suspended virtual machine i. Apply VMmap to derive physical models from virtual values. VMM goes to steep.

(3) fetch status i,), w:

Obtain virtual register or memory location | from

and the second state of th

virtual machine i and write into word w of the VMM.

(4) <u>set status</u> i, j, w:

Write contents of VMM word w into virtual register or memory location) in virtual machine i.

(5) <u>destroy</u> i:

Destroy virtual machine i. This cermits the identifier i to be used again.

Given that the host machine satisfies the Type I hardware requirements, then a Type IT VMM (non-HVM) may be programmed using these stimitives.

> vmm: i:=virtuelmachine loc, 256K, ...; dispatch i: loc: fetch status i,),w; (SIMULATE OFFENDING INSTRUCTION) set status i,),w: dispatch i: end;

The amount of status fetching and setting and simulation of sensitive instructions varies according to the machine's complexity and the difficulty of virtualizing. In a III generation machine which is easily virtualizable, the implementation of the VMM sketched above can be reduced to a straightforward program.

Manual Control of the state of

3.7 CONCLUSION

The main result of Chapter 3 has been the substantiation of the unsuitability of most existing IIJ generation systems for supporting virtual machines. We have shown that the architectural characteristics of III generation systems and the definition of virtual machine lead to a particular software construction of the virtual machine map. For example, the processor component of this software map takes the privileged layer of the virtual processor into the unprivileged layer of the real machine. This immediately leads to a set of empirical requirements, e.g. instruction execution must be insensitive to the software layer mapping, which distinguish III generation virtualizable architectures. A number of examples reveal the startling fact that most third generation architectures are not That CP-67, the best known virtual machine virtualizable. system, could be constructed on the IBM 36)/67 was a happy accident.

In response to difficulties with virtualizing most III generation architectures we propose two alternatives. First, we introduce the hybrid virtual machine which has less stringent hardware requirements but potentially more overhead than the virtual machine. Then, we propose a number of simple ad hoc changes that might be made to III generation machines to simplify virtualization.

The rest of Chapter 3 is concerned with implementing a Type II (extended machine host) VMM on a third generation operating

The virtual machine construction is basically the same system. as before. However, now the VMM must manipulate its components using the primitives of the host operating system. This leads to a series of empirical requirements, e.g. there must be an operating system primitive to shut off all SVC's of the virtual machine, that distinguish Type II virtualizable operating systems. Crice, again, most third generation operating systems are not Type II virtualizable. The magnitude of this lack is pointed out by the ease of implementing a Type II VMM 00 **an** operating system that is virtualizable, e.g. the Type II HVM under MIT's ITS for the PDP-10 [Appendix C]. to further acceptuate this fact, we propose a set of virtual machine primitives and show the aase of developing a Type II VMM using ther.

However, the main point of Chapter 3 has been that no III generation systems (hardware or software) have been designed with the intention of supporting virtual machines. This has forced virtual machines to be introduced via a particular III generation software construction. Requirements imposed by this construction in turn, have disgualified almost all III generation machines from supporting virtual machines. Even the few existing systems, a.g., CP-67, require considerable software support which implies development and implementation costs, and sub-optimally afficient operation.

These consequences lead us to search for machine architectures, specifically intended to support VCS*s. As noted in Chapter 1, these considerations gave rise to the procosals by

Fage 79 g [51]. While the g in connection with tuil machines into result are generally s, by suitable c orincibles for III ated in Sectior 4.9. Lauer and Snow [76] and Gagliardi ard Goldberg [51]. Chapter 4 is developed specifically in connection with acdel of providing for the orderly introduction of virtual the the IV generation, principles which result are generally **arplicable** :0 a11 architectures. Thus, model we can develop principles for III interpretation of the generation systems. This approach is illustrated in Section 4.9.

A Stand & State of State

A REAMERSAN

~

and a state of the state of the state of the state of the second state of the second state of the second state of the stat

efterskysterstydere (beidetssetterseated b

CHAPTER 4.

PRINCIPLES FOR IV GENERATION VIRTUAL COMPUTER SYSTEMS The Virtual Machine Map and its Firmware Implementation

4.0 PLAN OF CHAPTER 4

In Chapter 4, we will discuss the problem of introducing virtual computer system capabilities into the IV generation. A recent paper [51] points out some of the inherent difficulties and complexities caused by likely IV generation architecture and argues for a hardware-firmware assisted virtualizar built into IV generation computer systems. In this chapter, we carry the notion of hardware virtualization somewhat further and provide a general framework for viewing virtualization of a IV generation computer system. Furthermore, the result that everges is applicable to less complex architectures as well.

Section 4.1 introduces the likely characteristics of IV generation computer systems architecture. The orincical architectural development of such systems will be a formalized firmware implementation of processes. A corresponding model for processes is introduced; the process map, 4, maps process names into resource names.

Section 4.2 develops a model for a IV generation VCS. We introduce the virtual machine map (or resource mas) f which maps virtual resource names into real resource names. Then the virtual machine map is combined with the process map of Section

4.1. The composition of the two mappings, f c b, expresses the mapping from process names to real resource names, and thus expresses the mapping of names accessible to a process executing on a virtual machine. Other related notions, such as recursion and Type II virtual machines are introduced into the model. Some elementary properties of the maps and virtual systems are anumerated.

Section 4.3 uses the model of Section 4.2 to point out the unsuitability of a software implementation of the map f (as was studied in Chapter 3) for a IV generation VCS. The argument depends on the complexity of the process map, **6**, and the likelihood of **6** visibility by, say, the inner layer supervisory software. A specific example of the kind of difficulty a software implementation of f might cause is developed for protection rings.

Section 4.4 discusses the Venice Proposal, VP, introduced in "Virtualizeable Architectures" [51]. The VP provices an interim proposal for the design of a hardware-firmware assisted virtualizer (HV) for a IV generation VCS. The advantages and disadvantages of the VP are pointed out using the model of Section 4.2.

Section 4.5 introduces the design of a hardware-firmware virtualizer based directly upon the model of Section 4.2. The initial objective is automatic virtualization of CPU-Kemory processes (with a minimum of software intervention). The new instructions introduced, modifications to the system base and expected cerformance are all indicated.

Section 4.6 provides a number of detailed examples to illustrate instruction executions on virtual computer systems constructed with representative hardware virtualizers.

Section 4.7 briefly considers some of the subtler problems raised earlier in the chapter. Proper virtual treatment of input/output, a technological difficulty, must await progress in the real treatment of input/output. In the interim, some ad koc solutions to I/O, derived from III generation techniques, are provided.

hinner bezerningen (den kunnteisen die Stankner (den in er uit hinner

Section 4.8 applies the model of Chapter 4 to a reconsideration of second and third generation computer systems. One design that emerges, the "f=R-B, d=identity" machine, is similar to the machine recently proposed by Lauer and Snow [76]. The other design yields a modified IEM 352 with a greatly streamlined CP-67.

and a state of the state of the

4.1 CHARACTERISTICS OF IV GENERATION COMPUTER SYSTEMS

The most distinctive architectural characteristic that will found in IV generation machines will be the refinement and be formal implementation in firmware of the process model [31,37,41]. According to this view, activities in a computing system will (e vertormed by a number of cooperating sequential processes, each accomplishing some el ementary task and communicating among themselves via a well-defined mechanism. the processes themselves will execute in Furthermore, an environment extended to include idealized objects, such âS an idealized address space.

This view of the likely characteristics of IV generation architecture is based on the papers by G.M. and L.D. Amdahl [5,7], the Blauuw paper [19], the Morris paper [37], the Infotech book [70], and, particularly, on the existing Mitre Corporation VENUS system as described by Liskov [79]. Furthermore, we observe that key software features of generation <u>n</u> are often those features that are included in the interior decor of generation <u>n+1</u>. Therefore, Denning's recent tutorial [37] on (the software of) Late Third Generation systems strongly suggests that the process model will be incorporated into future systems.

Third generation computer systems rely on an ad hoc, purely software implementation of processes [37]. Thus, the choice of primitives, organization of tables, names of objects, etc., is not fixed and may be chosen by the software support. This implies that, for a given III generation system, there exists no

unique implementation of processes.

anhadhrainn di saralannan kaladhrainn karardaran kararda ar ar shirta ar an shirta ar an an an an an an an an a

STATE PROPERTY CALLS

By contrast, IV generation compute: systems will feature a firmware implementation of processes. The complete collection of status information for all processes in the system will be called the system base [51,87]. The firmware will know the base location of the system base, the RCOT. Since tables and control plocks will be of fixed (pre-assigned) format, knowledge of the ROOT is sufficient for the firmware to locate any element in the system base. The system base will be located in main memory. For largely economic reasons [51,87], the system base will likely also be accessible to the inner layer, most privileged software [see below]. However, during process execution, the system base is referenced and updated directly by the firmware.

The system base will include elaborate process control blocks, dueueing data bases, address space words, etc. [51,87]. See Figure 4-1. The queueing data bases will de used, for example, by the firmware dispatcher to invoke the READY process with highest priority [99]. The queueing data pases will also be used by the firmware in implementing interprocess communication via Dijkstra's P-operations, V-operations, and semaphores [41,79].

The process control blocks, (PCB's) will be pointed to by a process table whose location can be derived from the ROCT. Thus, an integer displacement in the process table, the process-id, identifies a particular process and the location of its corresponding PCB. The PCB will contain such information as temporary storage of the central processor's (CPU's) register

and the second for the second second



FIGURE 4-1

values, the priority of the process, and ar address space word. The CPU will contain a process-id register whose value identifies the process in execution. Thus, there exists no notion of an absolute mode: execution is always on behalf of some process. Loading the process-id register activates the subject process and instructs the firmware to obtain register values, etc. from the process control block (FCB). In particular, the address space word is referenced and used to form the memory map. Thus, there exists no notion of an ausolute addressing mode; all addresses are mapped.

The idealized address space will likely be a segmented layered structure [16,35,86,72,1⁰6]. Such a structure can he viewed as a generalization of the two layer Master/Slave modes of III generation systems. As the layer or ring 11.61 number of an active process decreases, the ability, by that process, to access segments in its segment table increases correspondingly. In ring), the most privileged layer, supervisory processes can also be expected to utilize a small set of necessarily privileged instructions. Privileged ring procedures of processes will also process exceptions. This is the likely be invoked by generalization of the trap to master mode in III generation systems.

The I/O systems of IV generation computer systems will likely be similar to those of the late III generation, e.g. IBM System/373. In particular, I/O-memory processes will very likely not be formalized to the extent that CPU-memory processes have been. I/O processor (channel) operations will execute in an

absolute mode without process-id's or segmented addresses [8]. All address absolutizing, status maintenance, etc. will be beenformed by software in an ad hoc manner [70]. As a result of the lack of progress by IV generation designers in I/O process formalization and implementation, the IV generation virtual I/O considerations are largely the same as those of the III generation. Consequently, most of the discussion of Chapter 4 will omit I/O; only in Section 4.7 will we illustrate III generation techniques for introducing virtual I/O in IV generation systems.

Abstraction

A model of (CPU-memory) process structure may be obtained by a scrutiny of the system base. Execution by the CPU is always on behalf of some process. Hence, the names used by individual instructions are always process names, whether they are local or global (system-wide) names. For example, data or instruction addresses are always local names, defined by the segment table of the active process control block (PCE). Process-id's are global process names and their meaning is defined for 311 processes via a single table in the system base.

Effectively, then, the system base relates two classes of objects--those used by processes (whether local or global) and those related to the actual resources. The process names are idealized constructs while the resource names are determined by the hardware. There is no requirement that process names have

the same form or structure as resource names. Thus, in particular, it becomes possible to speak of a process having a segmented, multi-dimensional address space. In contrast, the real (and virtual) memory resource address space must, of course, be linear to mirror physical construction. The distinction between derived and primitive resource names, such as a segmented and linear memory, is discussed in "Virtualizeable Architectures" [51].

We call the names used by crocesses, process names or P-names and those used by the hardware, resource names. In particular, the real resource names used by the obysical hardware are called F-names. [However, as we shall see below, V-names are also resource names.] It will be necessary to discuss a mapping function, **4**, which is implemented via the system base and maps process names into resource names. See Figure 4-2. [In Figure 4-2 and all figures that follow in Sections 4.1-4.4, process spaces are represented by circles and resource spaces by squares.]

Let the process space names be P={pA,p1, ..., p3}. As discussed above, P includes the process memory space names and semaphore names (for the active process), all process-id's, etc. Let R={r},r1, ..., rb} be the (real) resource space names. R also includes the processor and (at least conceptually) the I/O resource names, as well as the memory names. Then, for the active process, we provide a way of associating process names with (real/virtual) resource names during process execution. See
art a and a set of the destability of the set of the second set of the second second second second second second

$$\phi: P \rightarrow R \quad \cup \{e\}$$

$$\phi(x) = \{y \text{ if } y \text{ is the resource name for process name } x \in \{e \text{ if } x \text{ does not have a corresponding resource}\}$$



Examples :

Henereter mannaret daw in particular and Annalismer Annalismer and Andre manners and Alt in the second discussion and an internet discussion and a

 ϕ (1 | 142) = 6142 if seg 1 at location 6000

 ϕ (12 3) = e if seg 12 does not exist

 ϕ (sem 2) =-5000 if sem 2's header at location 5000

 ϕ (process id 40) = e if process 40 does not exist

THE PROCESS MAP ϕ

FIGURE 4-2

tinteredity. A. Orekeletiki hirticriteret e. Recenterenteriterin

Figure 4-2. To this end, via the system base, we define, for each moment of time, a function

6: $P \rightarrow R$ U {e} such that if x e P, y e R, then

The value $\phi(x)=e$ causes an exception to occur to some exception handling procedure, presumably in an inner layer of this process on this machine. To avoid confusion with virtual machine faults [Section 4.2], process faults will always be called exceptions.

Section 4.6 provides detailed examples of the process map, **b.** For clarity, a few simple examples follow. Throughout Chapter 4, we adopt the segmented address notation, sld, where s is the segment number and d is the offset within the segment [86,92].

Examples: 0(11142)=6142 if segment 1 is located at 6030

♦(20011)=e if seg 210 does not exist.

♦(315utu6)=e if 5 true outside bounds of segment 3.

(sem 2)=5305 if sem 2*s header at location 5330



wards swittent stread and have betweened and

4.2 MODEL OF A IV GENERATION VCS

In order to properly model the running of a process on a virtual machine, it is necessary to first clarify what is meant by a virtual machine and what are virtual resources. To do this, we formally introduce and identify an important new concept, the <u>virtual machine map</u> (VMmap) or resource map. The VMmap expresses the relationship between the resources of a virtual machine and the resources of the real host machine. Consequently, the VMmap is a notion which exists completely independently of the process shouture. Separation and isolation of the resource map (VMmap) has often been confused or overlocked by previous authors [51,75,108], and this perhaps accounts for the lack of success in designing easy-to-program virtualizable architectures.

<u>VMmap</u> f

San and the Color State Color and

A resource name used by a virtual computer system will be called a virtual name, virtual resource name, or V-name, and the set of virtual names defines a virtual space or environment. A resource name used by the obysical handware is called a real name, real resource, or R-name, and the set of real names in called the real space or environment.

In the resource soaces, real and virtual, we must include the processor, memory, I/O system and any connections between them. For example, resource names must exist for every

articered and a the product of the restriction of the structure of the str

addressable unit (bit, byte, word, etc.) in the physical memory. Resource names must exist for every orcore of the processor. Names are also needed for all I/O devices.

Therefore, the virtual computer system construction requires a mapping function that will map the complete set of virtual resource names into their corresponding physical resource names. This mapping function is illustrated in Figure 4-3. No a priori assumption is made about the nature of the mapping function f. [However, the implementation part of the VM definition [Section 2.1] implies that once the mapping is established most instructions of the virtual machine execute directly on the nost.1] The mapping function f is called the virtual machine map. resource map, or VMmap.

For future reference, we denote the virtual space names by $V=\{v0,v1,\ldots,va\}$ and the real space names by $R=\{n0,r1,\ldots,rb\}$. V is the resource space of the virtual machine and R is the resource space of the host. We make no assumation, here, about the relative sizes of V and R. Both V and R include the memory space names. But they also include the other resources, as well. For example, we can imagine ordering the opcodes by assigning them the names following the memory space names in V and R. Lauer and Snow (76) have observed that it may be sufficient to consider only the memory space names since other resources are often coded to appear as memory. For example, they observe that the accumulators of the DEC PDP-10 and the I/O devices of the CEC PDP-11 are each addressable as memory locations.

Since we assume no a priori correspondence between virtual

. . -

$$f: V \longrightarrow R \cup \{t\}$$

 $f(y) = \{z \text{ if } z \text{ is the real name for virtual name } y \\ t \text{ if } y \text{ does not have a corresponding real name} \}$



Examples :

mand the state of the second of Annales

In the stand with the part of the state of t

f(6142) = 12142

f (25000) = t if 25000 out of virtual memory

f (processor 3) = processor 2 if virtual processor 3 is real processor 2

f (processor 2) = t if virtual processor 2 not assigned

THE VMmap f

FIGURE 4-3

and real names, we must incorporate a way of associating virtual names with real names during execution of the virtual Sachine. To this end, we define, for each moment of time, a furction

f: V--->R U {t}

such that if y e V and z e P then

(he value f(y)=1 causes a trap or fault to some fault handling procedure in the machine R. For clarity we always term a VM fault a fault, never an exception. The spaces V and R define two vintual machine levels. We say that the vintual machine V is at a higher level than the machine R. If the level of R is n, then the level of V is n+1. If P is the physical tachine then it is level 0. Thus, a VM-fault is a VM-level fault and control casses to a fault handling machine at the next lower level. See Figure +-4.

Running a Vintual Machine: f o 🌢

In IV generation systems, running the virtual machine V={v0,v1,...vb} implies running some process P={p(,p1...pa} on the virtual machine. This defines the mapping

◊: P--->V U €e}

as before, with virtual resource names, V, substituted for real ones in the resource range of the map. See Figure 4-5a.

The virtual resource names, in turn, are mapped into their

lo so transfishiki maturitati ata da Annendrum astrumantaka edikeranan da maturan an ina ana ana ana ana ana a



1

El Ester

×1, 12

VM LEVELS FIGURE 4-4

Page 96

ļ



のためというなななない

のないがないのです

the second state of the second state of the second s







foφ FIGURE 4-5

いっき ちょういい いいというかいち なの

any manufation and

122.2634

real equivalents by the map, $f:V \rightarrow R_{A}$ See Figure 4-5b. Thus, a process name x corresponds to a real resource $f(\phi(x))$. See Figure 4-5c. In general, process names are mapped into real resource names under the (composed) map

White the first were stated and success

f o 4: P ---> R U {t} U {e..

Section 4.6 provides examples which illustrate mapping process names under the composed map f o 4.

<u>f vs. 4</u>

There is no a priori requirement that f or \$ be of a particular form or that there be a fixed relationship between them. [However, in some sense, 6 corresponds to how the users would like to program the machine while f corresponds to how the system would like to utilize its resources.] We discuss this point further in Section 4.8. We have indicated that in IV generation systems, the memory component of the map 6 will likely be segmented. Since f maps resource spaces, its memory component cannot be segmented. The only requirement of f and 6 is that the range of 6 be included in the domain of f. (Figure 4-5c)

We further distinguish between f and ¢ by contrasting the notions of level and layer in IV generation virtual computer systems. As previously stated f establishes a level relationship between a virtual machine and its corresponding real machine. Thus, f exists as an interlevel map. On the other hand, & is an intra-level map, mapping process names into a particular level of

and the second states and the second s

Page 98

resource names. One characteristic of the 5 map in IV generation systems will be layering. Thus a IV generation VCS will be multi-layer per level. ESee also Section 2.1.1

In the event of a IV generation VCS process name exception, control should be given to a more privileged layer within the same level. A virtual name fault, howevery should cause a fault to a process in a lower level number machine. Sea Figure 4-6.

Another contrast between f and & concerns their visibility by software procedures. Since & is an intralevel map, the inner layer privileged software of that level will likely manipulate &. However, f is an interlevel map. There should be <u>no way</u> in which software at level n+1 (regardless of layer) can access the f which maps level n+1 into level n. Thus, the f-map is <u>invisible</u> to all software at level n+1

Ivpe II Victual Machines Revisited

As discussed in "Virtual Machines: Semantics and Examples" [53], and Chapters 2 and 3, a Type II (extended machine host) virtual machine organization is one in which the VMM runs, not on a bare machine as the supervisor, but rather on an extended host under the host supervisor. Depending upon the uses of virtual machines, whether or not there exists one preferred, tavored operating system, and whether that operating system has certain capabilities to permit its construction, a Type II virtual machine system can be a useful and possibly easy to implement artifact.



ないなど

General States



PROCESS EXCEPTION AND VM FAULT FIGURE 4-6

There is a very similar, corresponding concept to Type II systems in IV generation virtual machines. In a Type II IV generation system, there is a slightly different VMmap f° which maps virtual resource names into corresponding process names for some process running on the real machine. Thus, for virtual names V, as before, and for process names Pr, for a process running on the real machine, we define, for each moment of time, a new function

f": V---> Pr U {t} such that if y e V and z e Pr then

> f*(y) = {z if z is the real P-name for V-name y { {t if y does not have a corresponding P-name

The value f*(y)=t causes a VM fault to occur to some fault nancling procedure in the process Pr (or some other process cooperating with Pr).

Since the process Pr is defined on the real machine via (its system base) the function ϕr_{1} a P-name generated by the VMmap is napped into the corresponding R-name. Thus, we have, for V-name y, the R-name $\phi r(f^{*}(y))$. This mapping is illustrated in Figure 4-7.

Now for any process PV running on the virtual machine there is defined (via the virtual system base) the function \$v, which maps the set of P-names, Pv, into their corresponding V-names, V. Thus, the action of running the process PV causes a P-name, x, to be mapped into the R-name \$r(f*(\$v(x))). This mapping is flustrated in Figure 4-8.





TYPE IVIRTUAL MACHINE FIGURE 4-8

Observe that in a Type II system, there exists a VM map f^{*} for each process which creates a virtual machine (or machines) at a higher level number. By invoking the layering of the real machine we can control the ability of processes to create virtual machines and if desired can legislate that there be only one.

Section 4.6 [E ample 8] provides a detailed example of a Type II IV generation VCS. However, a simple example, here, may show the intent of this construction. Let f'(x) = 31x. Then the Type II VMmap's memory component maps the virtual machine's virtual memory into one segment, say 3, of the process. This permits the normal operating system to run as the host on the real machine and perform resource allocation for the virtual machine (on a segment basis).

Secursion

As discussed in "Virtualizeable Architectures" [51] and Chapter 2, recursion for virtual systems is more than a matter of conceptual elegance or a consideration of logical closure. Indeed, it is a capability of considerable practical interest. In its simplest form, the notion of recursion for virtual machines is that, although it makes sense to run standard processes or the virtual machine, in order to test out the VMM software on a VM it is also necessary to be able to run at least a level 2 virtual machine. The real machine is level 0.

In the discussion which follows, we use a Gewey-decimal tree naming convention in which a virtual machine at level n has n

ちんごうたい とうちょう うくうしょう

Adverse of the second designed and

syllables in its name [51]. This tree-name is used 85 a subscript for both the virtual resource space, e.g. V1.1, and Where several different f1.1. VMmao, e.g. corresponding processes are running on virtual systems, the same tree-raming scheme is used, less precisely, to represent the process map on this the corresponding virtual machine, e.g. 41.1. In discussion, all f's are of the same form, and all t's are of the The subscripts are used to indicate difterent same form. antities.

In a Type I IV Generation VCS, if f1: V1 ---> R f1.1: V1.1---> V1 e: P ---> V1.1

And the second state of the second state of the second states

then the action of running the process P causes the P-name x to be mapped into $f1(f1.1(\phi(x)))$ or f1 of f1.1 of (x). See Figure 4-9.

In this function fi o ti.1 o 0, we identify three possible faults or exceptions:

(1) The process exception to an inner layer procedure of level 2, i.e. 6(x)=e.

(2) The level 2 resource (virtual machine) fault to an inner layer procedure of level 1, i.e. fl.l o \$(x)=t.
(3) The level 1 resource (virtual machine) fault to an inner layer procedure of level 0 (the real machine), i.e. fl o fl.l o \$(x)=t.

For the general case of level n recursion, we have P-name x being mapped into

that externees the debye of the observed when we have the second of the source of the second of t

f1 o f1.1 o ... o f1.1 o \$ (x)

Thus, for a Type I VM, regardless of the level of recursion, there is only one application of the map 1 followed by n applications of an f map. This is an important result that comes out of the formalism of distinguishing the f and 0 maps. An obvious result to be discussed later is that in a system with a complicated 0 but with a simple f, n-level recursion may be easy and inexpensive to implement.

By analogy, we define Type II VM recursion. See Figure

: Pr --->R
f*1: V1 ---> Pr
 o1: P1 ---> V1
 f*1.1: V1.1 ---> P1
 o1.1: P1.1 ---> V1.1

ara ny ara-daharan'i Ara-dara any karatara waang dini sa kadan inanandarata taku di karatara taku taka dinang t

Then the action of running the process P1.1 causes the P-name x to be mapped into Δl o f*1 c $\Delta 1$ o f*1.1 o $\Delta 1.1$ (x).







大山たちの政治には



TYPE IL RECURSION FIGURE 4-10

4.3 UNSUITABILITY OF SOFTWARE f

Before dismissing the possibility, it will be necessary to scrutinize IV generation systems to see if a software implementation of the VKmap f, is viable. We will see if the techniques of the III generation VCS can be apolied to the IV generation.

As has been developed in Chapter 3, a III generation VCS is normally constructed by:

> (1) Using the available memory mapping hardware to help manage the memory of the virtual machine, i.e. virtual memory.

> (2) Mapping the master and slave modes of the virtual machine into the slave mode of the physical host machine and simulating the effect of privileged instructions by the VMM.

> (3) Trapping all I/O instructions and mapping virtual
> device names into real devices via software in the VMM.
> [W_ will ignore I/O.3]

fhree of the architectural characteristics that ene might
expect to be useful in a software implementation of f on a IV
aeneration system are:

(1) Segmented address space

(2) Rings

(3) Emulation.

One might expect the address mapping to be useful in establishing the virtual memory while the ring protection system

an a burning and the second of the second of the second to be a compression of the first of reserves the second second

show where we are a second second

a sources of the sour

night provide the layered access control needed to isolate the VMM from its subject VM. As a last resort, one might expect to fall back on the extensive emulation facilities of IV generation systems [5,1:9].

Construction of the state

Unfortunately, as suggested in Section 4.2, none of these features helps the implementation of a VCS. Indeed, they make virtualization potentially more difficult since they will have to be virtualized as well. (This difficulty is similar to the problem noted by Cheatham [25] regarding the self-definition of particularly rich programming languages.] The problem results because segmentation, rings, and emulation are visible, i.e. they are part of the interior decor, and are represented by constructs in the system base which may be seen and menipulated by ("privileged" ring () software procedures. We can illustrate this difficulty with a look at rings.

Scrutiny of Rings

A software implementation of the VMmap f on a IV generation VCS requires (1) maintenance of a virtual system base by the virtual machine monitor, and (2) the control of access to the real system base by instructions of the virtual machine [51]. By analogy with the III generation "software VMmac", we prevent access to the real system base by not running ring 0 of the virtual machine as ring 3 of the real machine. Into what ring can ring 0 be mapped?

Two of the properties of rings which must be creserved in

any mapping are

(1) monotonicity

A PARTICIPATION AND A PARTICIPATION

(2) finiteness.

That is, the rings define an ordering of privilege: ring i is more privileged than ring } if i<] [57,106]. Furthermore, there is a finite, relatively small number of rings in the system's universe, e.g. Multics has eight [35,106].

Therefore, two ring mapping schemes are presented in Figure

f(x) = Ex if 0<x=<m E Et if x=0

maps rings identically excect in the most privileged ring. All ring 0 instructions are simulated on an instruction by instruction basis. An attempt by an outer ring to call a ring 0 procedure must cause a process exception which is understood as a fault to the VMM. This construction is the IV generation equivalent of the hybrid virtual machine, HVM. [See Sections 2.1 and 3.3.] The empirical requirements are similar, e.g. attempted calls to ring 0 can be made to fault. The performance of the IV generation HVM is significantly degraded because of the likely frequent software intercretation of all ring 1 instructions.

The second mapping scheme,

maps ring [into ring 1. To preserve monotonicity of rings, ring 1 must be mapped into ring 2, etc. However, because of the finiteness property, it is necessary to map a (particular) pair



Ring mapping scheme 1: $f(x) = \begin{cases} x & \text{if } 0 < x \le m \\ t & \text{if } x = 0 \end{cases}$



RING MAPPING SCHEMES FIGURE 4-11

of adjacent distinct rings of the virtual machine to a single ring of the real machine. In a III generation computer system, if master mode is understood as ring 0 and slave mode as ring 1, the processor mode component of the software VFmap for III generation systems [as in Chapter 3] is just the second mapping scheme, f(x), with m=1 and i=2. Analagously with III generation ampirical requirements, it is necessary to prevent the processor from determining in what physical ring execution is taking place. This is rather difficult because of the potential visibility of the ring number in the instruction counter ("ring register field") or in stacks, for calls acress a ring boundary [106].

Another problem which might develop concerns the absolute interpretation of physical ring number for certain side effects. For example, one proposal [196] has the hardware identify ring numbers with the segment number of the stack segment for that ring, or ring numbers, i.e. 9-1, 2-5, 6-7, with different access properties to gates.

One simple hardware proposal that might be made to avoid some pitfalls of the second mapping scheme is a ring relocation register. See Figure 4-12. The ring relocation register could be used to linearly displace ring numbers and, thus, prevent ring) of the virtual machine from having the same privilege as ring 0 of the real machine. However, problems still exist concerning the finiteness property. What ring does ring 7 get mapped into? In recursion, what ring does ring m-1 get mapped into, etc.?

and A law and the constraints and y and a second state of the second of a last

Thus, the mapping of ring numbers via software or with nardware assistance does not necessarily help ir the



and the second state of th

implementation of IV generation virtual machines. As we shall see, the software mapping technique is unratural and unnecessary given a proper understanding of the model of Section 4.2. Fourth generation systems formalize the process map in firmware. Thus, in introducing virtual machines into the IV generation, it is natural to consider how firmware may be used to formalize the VMmap, as well.

Sto Benevictor

4.4 THE VENICE PAPER

In "Virtualizeable Architectures" [51], called the Venice Paper or Venice Proposal, abbreviated as VP, the authors propose an interim solution to the problem of virtualization of IV generation computer systems. The VF observes that because of the nigh frequercy of references to the system base, its maintenance as pure software construct is likely to be boomed to failure. [See Section 4.3 above.] Thus, the solution procosed by the VP is that hardware-firmware be provided to assist virtualization. Each virtual machine will maintain its own system base, visible to itself. However, when the virtual machine is actually dispatched it will be running with a special "transtated system base" maintained for each virtual machine by the VMM. Thus, any attempt by software running on the virtual machine to read its system base will be directed to the virtual system base and will go unintercepted. However, any attempt by the virtual machine to write its system base faults to the VMM which can make the corresponding change to the translated system base before redispatching the virtual machine.

Thus, the Venice Proposal provides a major improvement over previous virtual system designs in permitting read access to the system base to proceed without interruption. Only on attempted write access to the system base will the familiar fault-simulate-redispatch sequence be required.

and his and a subscription of the subscription of the subscription of the subscription of the subscription of the

Furthermore, the VP provides a virtualization design that permits a limited two level recursion of virtual machines to

ા અને આવેલી અને આવેલી છે. આ જેટલા છે. આ જેટલા જેટલા છે. આ જેટલા છે. આ જેટલા છે. આ જેટલા અને આ જેટલા આવેલી આ જે

lti viun antitus kakalaisis sisaa sada araa sada araa sa

pperate with hardware dispatching of the appropriate parent virtual machine on a write-access fault to an inferior virtual machine system base. The performance of the second level virtual machine could be comparable to the one level case.

We can interpret the VP in terms of the model developed in Section 4.2. Let P,V,R, 4, and f be as before. Then the VP maintains, for a virtual machine, a translated system base,

 $\phi t = f \circ \phi$

and this is, in fact, the process map that is active when a process of the VM is dispatched. See Figure 4-13. The VMmap f is strictly a software map and never appears in firmware accessed tables; of is accessed by the firmware. The VMM statically composes f with o to produce of. Thus, f o o is "calculated" in advance of the process being dispatched. Consequently a modification to o cannot be reflected in a corresponding dynamic modification to ot. To maintain integrity, a madification to o must cause an immediate VM-fault to the VMM, which calculates a new value for of and redispatches the VM.

Note that this analysis (as with the discussion of Section 4.3) implies that there must be an easy way for the hardware to tell that an attempt is being made to modify the system base. A special instruction might be reserved. [See Section 3.4.] Otherwise, all addresses in a physically "overlayed" segment structure might have to be checked, significantly degrading performance.

A detailed illustration of actual instruction executions in the VP is presented in Section 4.6.

de. A strategy and the second second second and the second second second second with a second second second second



Reserved Whomster in the second state of the second s

the stream and the ball of the second

VENICE PROPOSAL FIGURE 4-13



VENICE PROPOSAL WITH RECURSION FIGURE 4--14

In the recursive case, the VP*s VMM statically calculates, via software, the translated composed map

Summary

The VP proposes an implementation of a IV Generation VCS. Unlike the purely software techniques discussed in Section 4.3, the VP does not require the "compression" of privileged layers of the real machine to less privileged layers of the host. Consequently the VP should produce efficient IV generation virtual machines. Furthermore, the VP should be feasible in many cases in which pure software techniques are impossible to use. However, the VP suffers from the following disapventages:

> (1) Performance degradation due to faults on system base write instructions

> (2) Storage wasted due to translated system base. This could be particularly significant in the recursion example.

(3) Complexity and software needed, in the VMM, for (1) and (2).

In the rest of this chapter, we shall consider a proposed design which eliminates these weaknesses.

and a second second second

Dillares Manara and a straight and a straight and a straight and a straight a straight a straight a basis of a basis of

4.5 THE HARDWAPE VIRTUALIZER

The Hardware Virtualizer (HV) is a hardware-firmware implementation of the VMmap f (for a IV generation VCS). In this section, we shall develop the arguments in favor of an HV, show how an HV can be constructed, show that the construction is feasible in terms of IV generation hardware (realization) components, and indicate that satisfactory performance will occur. As in Section 4.1-4.4 above, because the IV generation treatment of I/O is so similar to III, we do not incorporate T/O into the HV. Similarly, because of the intrinsic problems with time in a VCS we do not discuss time. These difficulties are considered in Section 4.7.

Acaument for HV

It is our claim that software construction of the VMmap f is unnatural and unnecessary given a proper understanding 01 virtualization as developed in the thesis [especially Section 4.21. In numerous other areas of computer systems research as pasic and underlying principles have been developed or discovered, ad had and often cumbersome software constructs have bean replaced by elegant hardware (-firmware) mechanisms. Cne can turn for confirmation to management of multi-level memory i 28,38,391 or protection rings [57,1°6]. The firmware implementation of the process model in IV generation systems [51,79,87] is a tour-de-force, including the two previous

he construction of the second of the second second of the second of the second of the second second of the second of the

examples within it as special subexamples.

We claim that the hardware virtualizer HV brings about a simplification of the virtualization mechanism. The HV simplifies and eliminates most of the VMM software, reduces the main storage requirements of the VMM, and produces satisfactory operating performance. Furthermore, it permits virtualization to take place where, using previous software techniques, it ordinarily would not be possible.

HV Construction

and the state of the second second

Our design of a hardware virtualizer follows directly from the model of Section 4.2. The maps f and are distinct in a IV generation VCS. Therefore, we implement them as completely different constructs, caoable of being referenced and manipulated independently. Execution by the virtual machine causes the macs, f and a, to be composed dynamically to form f o a at execution time.

The construction of an HV must address the following points:

- (1) The database to store f
- (2) A mechanism to invoke f
- (3) The action on a VM-fault

State of the state

In the discussion which follows, we shall develop the basis for an HV design somewhat independently of the particular form of the VMmap f which is being implemented. Although we shall refer to certain particular f structures, such as the R-R or paging form of memory map, the actual detailed examples are postboned until

Section 4.6. [The discussion which follows assumes the IV generation process map 4 presented in Section 4.1. As we note in Section 4.8, suitable simplification and interpretation of the 4-map allows us to extend this construction to second and third generation architectures as well.]

Database to Pepresent 1

The VMM must create and maintain a database which represents the VMmap f for the next level virtual machine(s). This database must be stored such that it is invisible to the virtual machine, including its most privileged software. Let us assume that for economic reasons [51] the database must be stored in main memory. Then the VMmap itself must be such that f is not in the (virtual) memory of the virtual machine.

The VM database, logically should be an addition to the system base. A pointer at a fixed location from the RCOT coints to a Virtual Computer System Table (VCSTAB). Each entry in the VCSTAB points to a Virtual Computer System Control Block (VCSCB) ar colloquially, VMCB. Thus, the i-th entry points to the control block for virtual computer system i.

The VCSCB provides the implementation of the VMmap f for the virtual machine. It contains the memory map, processor(s) map, and, at least conceptually, the I/O map. In addition, there may be other status and/or accounting data for the virtual machine.

As noted in the VCS model (Section 4.2), the maps 4 and f are completely different. In particular, the memory components

of the maps serve different purposes and so are likely to be of different form. The 4 map likely will be segmented. Depending upon how the VM's will be used, the f map might be R-3, paped, or nulti-level paged. If a limited number of virtual systems are used irregularly for operating system debugging or subsystem transferability, ar R-3 map might be reasonable. If fictitious memory resources are required, some form of page map may be used. [See performance/cost discussion below.] When we discuss a page map, we, usually, do not illustrate fictitious resources, e.g. demand paging.

The processor component of the VCSCB incluses (<u>per processor</u> of the VCS) the saved values of the processor registers (the last time the VM was running). Note that this is different from the <u>per process</u> saved register values kept in the respective PCB's of each process on the VM.

These registers typically might include:

P - current process ID IC - instruction counter T - top of stack STR - status ER - base registers

GR - general registers

SR - scientific registers (floating point)

[In this discussion, we assume that ROOT is zero for simplicity.]

Additional processor related information kept in the VPCB includes canability information, for the particular virtual processor (not the real processor and not processes running on

and indextooktooktokeenterooktorootootootootootootootootootoo

the virtual processor) indicating manticular features and instructions, present on absent. These include, for example, scientific instruction set, VM instruction set (LVMID instruction), etc. Additional information might include a perenal opcode fault mask on a fictiticus rescurce mask. The oncode mask can be used to fonce certain opcodes in the VM to fault to the VMM. [This notion is cerived from Section 3.4.] A fictiticus resource mask might indicate whether (bounds type) faults are to be directed to the VM on the VMM.

<u>Mechanism to invoke f</u>

In order to invoke the VMmap f, the HV requires one additional visible register and one instruction for manipulating it. The register is the VMID and contains the number of the VCS in the VCSTAE that is currently executing. The new instruction is LOAD VMID, LVMID. See flowchart in Figure 4-15.

For the handware virtualizer design to be successful, the VMIE register (and the LVMIE instruction) must have three crucial promenties [51].

> (1) The VMID register absolute contents may reither be read nor written by software.

> (2) The only modifications possible to the value in the VMID register are accerding a low order IC syllable (as a result of an LVMID instruction) or removing some number of low order ID syllables (as a result of a VM-fault).
> (3) The VMID of the real machine is the rull



and the state of the

al and a second s

AAMER'S BAADANS

- ALE CONSTRUCTION

in a start of the

2.55

identifier.

n na tre na subtra averal to na subsection de subsection de la tre () (

These properties all follow directly from the desire to model the relative resource contexts provided by invocation of f.

Needless to say, LVMID "loads" (appends) the VMID with the ID specified as the operand of the instruction. If the VM capability does not exist for this machine, a VM capability exception occurs, which is a process exception to a privileged procedure in the same machine. If the VMID is loaded successfully, the firmware checks the appropriate VCSTAB entry for validity. If it is marked as absent or invalid, a VCSTAB exception occurs to a procedure in the same machine.

Otherwise the processor enters virtual mode and (assuming no faulting because of VCSCB values) loads the processor registers (P, IC, BR, etc.) from the VCSCB.

Now the memory component of f is active, say relocation and bounds, R-B. This causes a loading of the R-B into scratchoad registers, say, Rcomp-Bcomp. Thus, every virtual memory name will be transformed to a real name by the R-B map, i.e. value currently in Rcomp-Bcomp. [Of course, the scratchpad registers Rcomp-Bcomp, may be neither read nor written by any software. When the real machine (VMID=null identifier) is running, Rcomp=0 and Bcomp=physical memory size.] In a segmentation system, the result of looking up a segment P-name and obtaining the V-name must be subjected to the R-B map. However, since the segment table is not the root of the system base, the firmware must first locate it from the true ROOT and the process's P-name, i.e.
where a state with the state of the state of

gungsandergelauen investigenet investigen strander van der strander van der en der oor oor oor oor en onder de de oor

process-id. This endeavor requires the firmware to make several references to main memory in the course of "walking" that system base list structure. Each of these references to the virtual memory must be mapped to the real memory equivalent by applying R-B, i.e. adding Rcomp-Bcomp.

Eventually, the location of the segment tacks is established and it is saved in a scratch pad register. Thus, any subsequent memory reference may directly utilize the segment table to obtain a virtual memory location which is then mapped, via R-B, into its real equivalent.

For reasons of performance, the original machine requires an associative memory to store the most recent "segment number-memory location" pairs. Then, the hardware virtualizer's associator will store "segment rumber-real memory location" pairs. On setting a new VMID, the associator can be cleared. Another possible choice, which can be made on the basis of cost/performance, is to have a still larger associator and keep the VMID as a field to be used as part of the search key.

Recursive invocation of virtual machines adds little additional complexity to this implementation. If the VM capability bit in the VMCB is set to one, then the virtual machine has the same (software visible) handware virtualizer VM features as in a real machine. Therefore, a V4M running on the VM may attempt to create an inferior virtual machine by creating a VCSTAB entry and a VCSCB for the level 2 VM in its system base. It may also dispatch the (level 2) VM by executing an LVMID instruction.

As noted in the Venice Paper [51] and above, recursion requires that the VMIC be made a "multi-syllable register". When the real machine is executing, the VMID value is rull. When the VMM disoatches a VM, it loads (appends) a value into the VMID, say 1. If that virtual machine is running a VMM which creates and dispatches a VM, say ID 1, then 1 gets appended to the VMID. Thus, the new composite value in the VMID is "1.1" which gives the ID of the currently executing VM. As stated above, this value may not be either read or written by any software of the virtual machine. See the tree structure shown in Figure 4-16.

To handle a small, finite depth of recursion requires the VMID register to be sufficiently "wide" for specification. In practice, one should not expect the VM level to go more than 3 or 4 deep.

As in the non-recursive case, the execution of the LVMID instruction by the level 1 VMM causes its VCSTAE and VCSCB entry to be checked for absence or validity. If the LVMID completes successfully, the machine is now in virtual mode at the next level.

Now the memory component of the new f is active. [In this discussion, a map of the form R-B.] This causes an addition of the new R-B to the scratchpad register Rcomp-Bcomp where the current R-B value "total" is being maintained. However, in addition to the running sum, a collection of scratchpad registers are set aside to hold the R-B values of each level. [As Lauer and Snow [76] observe, it may also be possible for the firmware to store the reverse "trail" ir main memory.] This permits

ward a strate to the start of the start a start as the start



ないよるというないのというというとうないと

VM TREE STRUCTURE FIGURE 4-16

100

restoring the previous composed map by subtracting on a VM-fault. In terms of the VCS model, the running sum of the R-B's, Rcomp-Bcomp, provides a calculation of the composition of the two f maps (in the memory dimension), say f1 o f1.1. Then the composed map is combined with the Ø-map.

Thus, every virtual memory name will be transformed to a real name by the composite R-B map, Rcomp-Bcomp. This is done, just as in the single level case, with the composite R-E map taking the place of the simple R-B map. This includes walking the level 2 system base, etc.

Just as in the single level case where an associative memory was used to store 'segment number-real memory location' pairs, so in the recursive case, an associator stores 'level 2 segment number-real memory location' pairs. The performance implications of the associator (map composer) design will be discussed below. Section 4.6 provides detailed examples, including illustrations of the sacciator during HV operation.

Our discussion of the VM invocation has utilized an R-B memory component in the VCSCB. The mechanism described works for other memory maps, as well. [However, we are assuming that there is only one f map structure, common to the entire system.] For example, a paged memory map might be stored in the VCSCB. After the LVMID instruction, map composition in the memory dimension is dynamically performed on a page basis. This implies that the associator for a paged HV is of a slightly different form from the R-B HV. Illustrations are provided in Section 4.6.

VM-fault

On a VM-fault (VM-level fault), control must pass to the VMM. If the VMM is running on the real machine, control passes to the real machine. If the VM is running at level n, then control passes to the VMM at level n-1. The faulting mechanism (firmware) must inspect the VMID register and discard the low order syllable. The resulting value is the ID of the VMM's machine. See flowchart in Figure 4-17.

The characteristics of the VM-fault can be derived from the empirical work with III generation systems [Chapter 3]. In particular, the VM-fault must be completely invisible to the VM. The VM can have no way of testing or discovering whether a fault has occurred.

On a VM-fault, the processor's registers must be saved in the VCSCB and the associator must be burged. A fault code can be stored in the VCSCB and the VMM is started. After processing the fault, the VMM may return to the VM by issuing an LVMID instruction.

Feasibility of Construction

The HV described above is feasible to construct on a IV generation system. The hierarchical levels of control and map composition facilities add little additional complexity to the process model. Three characteristics of IV generation realizations [19] will be the use of large control stores, elaborate micro-logic, and associative memories. The hardware



and the second

Ś

AND A DESCRIPTION OF A

はないないのないのようななない

A STATE OF A

Page 130

and a good and a second and a second second and a second second second

virtualizer will utilize these same building blocks.

Eerformance Assumptions and Analysis

The expectation of acceptable performance by processes running on the virtual system depends, to some extent, on the expectation of acceptable performance in the real system. As discussed above, the process map \bullet in IV generation systems will likely involve rather elaborate constructs which must be accessed by the firmware for every instruction execution. For example, in the case of segmentation, it will be necessary for the firmware to obtain the resource location names (from F-names) via \bullet for every instruction. This will typically requires

> (1) The most recently referenced segments to have P-name, R-name pairs stored in some small finite rumber of associative registers, and

> (2) The recently referenced segments to be maintained in main memory by the software.

Needless to say, any architecture which incorporates paging (as well as segmentation) in the • map will certainly be implemented this way. [As we shall see below, caging properly belongs in the f-map instead.]

That a IV generation architecture (with segmentation, etc.) should perform acceptably depends significantly on the principle of locality. From the initial notion of "program locality" as noted by Dennis and Denning, Madnick [60] has generalized and identified two specific aspects of locality that are used.

(1) Temporal locality

and the state of the

If the logical addresses <a1, a2, ...> are referenced during the time interval t-T to t, there is a high probability that these same logical addresses will be referenced during the time interval t to t+T.

This behavior can be rationalized by crogram constructs such as loops, frequently used variables, and frequently used subroutines [80].

(2) Spatial locality

If the logical address a is referenced at time t, there is a high probability that a logical address in the range a-A to a+A will be referenced at time t+1. This behavior can be rationalized by program constructs such ast sequential instruction sequencing, and linear data structures (e.g. arrays) [80].

In IV generation systems, because of the complex 4 and the cost required to "start up" the map, it is likely that the (software) scheduler and (firmware) dispatcher will enforce an additional locality:

(3) Process locality

A STATE AND A STATE AND A STATE

If the process identifier of the process executing at time t is P, then there is a high probability that it will be P at time tf1.

Virtual machines and the hardware virtualizer and a new notion.

(4) Virtual machine locality

If the VMID of the currently executing VM at time t is x1.x2.xn-1.xn, then there is a righ probability

that the VMID will be x1.x2.xn-1.xn at time t+1. Furthermore, the other values that it can take are x1.x2.xn-1 or x1.x2.xn-1.xr.xn+1. The first value occurs on a VM-fault, the second on recursion. On multi-level faults, other values are possible.

VM locality implies that, with proper implementation, one level or multi-level recursive virtual machines need not have significantly different performance from real machines. Another way of phrasing this observation ist

Iemporal and spatial locality are name invariant.

Regardless of what a page-size block is called, or how many times it gets renamed (via a VMmap) there is still an intrinsic probability of reference to it by the executing process. Thus, a map composer aided by an associative store should provide comparable performance in the virtual machine to that obtained in the real machine.

We examine the performance of a virtual machine vs. a real machine for the R-B virtualizer and for the paged virtualizer. Performance of the HV can be judged by a direct comparison of instruction execution rates of the VM vs. the real machine. The use of a cache, look-ahead, etc. makes the comparison difficult. However, we can constrast the time required to develop a physical memory address which then gets sent to the memories.

In the R-B virtualizer, as skatched above, let the following parameters represent access times,

- a associative memory
- M main memory

s - scratchpad memory

Let pk(R) be the probability that a segment entry will be found in an associative memory of size k, in the real machine. Let pk(V) be the corresponding probability for our virtual machine. Then pk(V)=pk(R). Let us fix k so this probability is just p. Then the average time to develop a physical memory address is approximately

Ar = a*p + M*(1+p) for the real machine

Av = a*p + (H+s)*(1-p) for the virtual machine Since the scratchpad maintains a cumulative value of R-B. Rcomp-Bcomp, during recursion, Av is constant regardless of the level VM. Reasonable parameter choiccs are a=50 nsec, M=500 nsec, s=50 nsec, and p=.99. Then Ar=54.5, Av=55, and Ar/Av>.99. This result implies that, for an R-B virtualizer, regardless of the level of virtualization, Av is approximately equal to Ar.

In the paged virtualizer, the associator is of different form from that of the real machine. Whereas the real machine's associator relates segments to physical memory, the associator in the paged virtualizer must relate page-size blocks within a segment to the physical location of the page-size block. Therefore, in order to contrast the performance of real and virtual machines, we let pk be the probability that a segment page-size block entry will be found in an associative memory of size k. For a particular value of pk, k will be larger in the segment-page associator than in the simple segment associator.

The parameters a and M are as above. We assume that for aach level VM a scratchpad register holds the base of the page table. Furthermore, we assume that all scratchcap references are overlapped with main memory fetches and so are ignored for this analysis. Then

 $An = a^*pk + (r+1)^*M^*(1-pk)$

where n is the level of VM. If we set n=0, we obtain the result for the real machine,

 $A0 = a^{+}pk + M^{+}(1-pk)$

Figure 4-18 plots the ratio A0/An as a function of n for a=50 nsec, M=500 nsec and representative values of pk. In practice n is unlikely to be greater than 3 or 4. Figure 4-19 plots A0/An as a function of nk for n=1,2,3,4.

From the graphs, it can be seen that performance of a paged 4V will be respectable, depending upon the environment and application, for a range of values of pk. For n=1,

۵K	ADIA	
900	•66	
950	•74	
998	.92	
995	.95	
999	.99	

Through choice of 4, pk can likely be made to be .99 [105]. Then, the virtual machine runs at 92 percent the speed of the real machine (at least as far as address development goes). This speed might be satisfactory for a normal production environment.

With pk=.99, a level 2 VM runs with A0/12=.34. This is likely to be satisfactory for debugging the VMM.



and the second second

Page 136

FIGURE 4-18





4.6 DETAILED ILLUSTRATIONS

In the previous sections, we have indicated the structure and operation of typical IV generation hardware virtualizers. The intent of the present section is to make this operation perfectly clear by illustrating with a nurber of detailed concrete examples in which we show the execution of actual instructions by the V1. Most of the examples provide the same sequence of instructions being executed by (similar or) identical processes on the virtual machines. This is done so as to provide a basis of comparison between the different VMmaps. The examples generally use stylized instructions, i.e. LOAC, STORE, P. V. SIGNAL--PROCESSOR, and do not worry about accumulators, base registers, etc. Furthermore, the examples CO not cover fictitious (memory) resources, I/O, time and other potential special difficulties. The claim is that CPU-memory processes are sufficiently interesting and complex in IV generation systems. If we can run them well in a virtual machine we have accomplished nuch.

Figure 4-20 shows our stylized typical IV generation system oase located in main memory. All pointers illustrated in the figure are absolute storage locations. Associated with each pointer is the size of the table that is being pointed to. The firmware has available to it the base address of the system base, ROOT. The Running Process Word (RPW), Process Table Pointer, etc. are located at fixed, known displacements from the ROOT. Thus, the firmware may locate information about any process by



AS A KANAGARA

SYSTEM BASE

Carl Service States

FIGURE 4-20

starting at the ROOT and tracing through the system base. The ith entry of the Process Table points to the Process Control Block (PCB) of process i. The PCB stores accounting information, CPU register values (for this process), etc. There are also oointers to a semaphore table and a segment table. The semaphore table is discussed, together with the Running Process Word, Active Process Link Table, and G-Ready Process below in the example illustrating P and V operations on semaphores. [See Example 6 below.] Each entry of the segment table is a segment descriptor which is a pointer to that segment (if present in main memory), together with status, and (ring) access information. While we assume a layered ring structure, none of the examples presented below will use it explicitly and so we will represent a segment descriptor by its location and length. The handling of access violations will be very similar to the bounds violations which will be illustrated below.

Rather than reproducing this complete system base in the examples, below, we will usually just show the segment table of the currently running process as in Figure 4-21. This is just the memory component of the process map **6**. This, of course, implies that the name of the currently running process, Pi, is stored in the Running Process Word, RPW.

In the examples, we will also indicate the current value of the executing processor's instruction counter, IC, as a segmented address. This is the IC value that, say, has been loaded from the PCB when process Pi was dispatched. Again, we will not indicate the IC's ring information in the examples.



Main Memory

NOTATION FOR SEGMENT TABLE OF RUNNING PROCESS FIGURE 4-21

こうしんからい かいうまたい アオンパンちか ちぞうかう

· And the second s

In Figure 4-22, we show the extension to the system base that we will use for virtual machine support in the Type I VCS examples. All of the structure shown in Figure 4-20 is present nere. However, in addition, we now have a pointer at a fixed displacement from the ROOT to the Virtual Computer System Table (VCSTAE). [Note: In general, all pointers have the size of the table they point to stored as a field. Thus, attempted accesses may be checked.] This table has one entry per virtual computer system, which is a pointer to its Virtual Computer System Control Block (VCSCB). The VCSCE has, say, four parts: general status information, the virtual memory map, the virtual processor(s) map, and the virtual I/O map. The examples, below, will not discuss the status information or I/O map. The I/O discussion follows in Section 4.7. Thus, we will normally show just two components of the VCSCB in the drawings. Most examples will illustrate a one processor computer system in which case the processor map reduces, largely, to the storage of register [These are, of course, the register values of the values. processor when it VM-faulted, not of any particular process being run.] Under those conditions, the VCSCB is sometimes called, colloquially, the Virtual Machine Control Block (VMCB).

The memory map component of the VCSCB is illustrated for several different memory maps in the examples. In the examples of a conventional relocation and bounds memory map R-B, the values are shown stored in the VCSCB. The values are represented as the relocation value and the length of the contiguous allocation. See Figure 4-23. In the examples of a conventional

Page 143

おおいろう ひょうきょうぎ しょうちょう ストリー ひょうぞくい ひみのう ニットアイプロデオテアイ いたいちょうちょうちょう

ちんちちょうごろうう

AN EXAL AN

1

and the factor of the factor

and the second of the description of the second states and the second second second second second second second



AL (04157) 5 (04

ACCESSION A





Martin Martin Martin Martin Martin Martin Martin and

A CONTRACTOR OF CONTRACTOR

ノストロンシン

- ANA ENGLARS 24

(fixed size block) paged memory map, the VCSCB contains a pointer to the page table. See Figure 4-24. The page table may contain status information such as presence, usage, atc. These considerations are essentially the same as that discussed for JII generation virtual computer systems [see Chapter 3]. In our examples we do not illustrate fictitious memory resources and always show all pages in main memory. [This, of course, is not a requirement.] Thus, status information is smitted from the examples.

The Virtual Machine Identifier Register (VMID) is also illustrated in the examples. Perhaps, more accurately it should be called a VCSID buy the name VMID is kept for historical reasons. The value of the VMID identifies a particular VCSCB (or set of nested VCSCB's in the recursive case) and defines the map (or maps) that we have been calling f.

Example 1 - Type I VCS, Single Processor, P-B memory map

の語言にはないない。自己になった。語言になってないではない。

In this example, we illustrate some situle instruction executions on a virtual computer system implemented as a Type I JCS with single processor and relocation and bounds memory map. Figures 4-25 and 4-26 show part of the system base used by the VCM and part of the system base of VM1, the first level virtual wachine (number 1). The dashed lines in Figure 4-25 indicate the segments of process P1 in the real machine, i.e. the VMM. The dashed lines in Figure 4-26 indicate the segments of process P1



und de De Annalez (; 2005) hanne de State () and a state anna hanne antara (de Ballerine de La casa a state a

EXAMPLE 1 : THE LVMID INSTRUCTION

FIGURE 4-25



EXAMPLE 1: EXECUTION AND EXCEPTION

FIGURE 4-26

そうがいたいかいきょうで、いたいかい、そうかんできるなるというできたいで、そうできょうちょうという、たいでいたいかいでいるとなるになったが、したいし、そうかい

うんいいうがい ひままなる ひがまたい ふく

in the VM. The dark lines in Figure 4-26 indicate the VM's memory relocation and bounds.

Figure 4-25 illustrates the initiation of a VM by the VMM. Since process P1 of the real machine is running, the RPW in the system base is P1 [although this is not shown explicitly in the figure] and the processor's IC is 11230. This segmented address means segment 1, word 200.

The execution of instructions is traced in the commentary to the left of Figures 4-25 and 4-26. We will elaborate on the commentary for this example. The other examples can be tollowed without as much elaboration.

The instruction counter value must be mapped via the process map & from a process name, 1120F, into a resource name, 1250. This is done as a result of the segment table lookup. Since VMIF=NULL, execution is on the real machine, and the resource name, 120, is a real resource. Consequently, we must fetch and attempt to execute the instruction located at physical address 120F. This is the LVMID. A similar address calculation is carried out for the instruction's operand. The operand obtained is '1' and that value is loaded into the VMIE register. Consequently, VM1 is activated and the information stored in the VCSCB, i.e. memory map, processor maps, etc., gets loaded.

Figure 4-26 picks up the action when the V*TE is 1. Since process P1 of VM1 is now running, the RPW of the VM*s system base is P1 [although this is not explicitly shown in the figure] and the processor*s IC is 21530.

The instruction counter value must be mapped via the process

map & from a process name, 21500, into a resource name, 2500. This is done as a result of the segrent table lookup. The resource name developed is a virtual name since the VMIC=1. Therefore, it must be mapped via the VMrap f into its corresponding real equivalent. Since the VMmas*s remory component is of form R-F, the real equivalent is obtained by checking the bounds and adding the relocation. Both are expressed in, say, 1000 word units. Thus, f1(2300)=5500. Consequently, we must fetch and attempt to execute the instruction located at physical address 5500. This is the LOAD instruction. A similar address calculation is carried out for the instruction's operand [see steps (e) - (g)] and the next instruction's address [steps (h)-(k)].

When we try to develop the address of the STORE instruction's operand, i.e. 112000, we cause an exception. On application of the process map 6, we discover that 2001 exceeds the bounds of segment 1 in process P1 (in VM1). Thus, an exception occurs to, say, a privileged layer of the current machine. This privileged procedure may be in process P1 or it might be in some other process that gets dispatched as a result of the exception. [See the semaphore discussion, below.] In any case, this event occurs without the knowledge or concern of the VMH. The VMID is still *1* as all VM local exceptions that occur are handled automatically.

In a typical IV generation system, similar exceptions and local exception handling might occur for access violations or dynamic segment linking. These are events that will likely occur

ちょうしょうかん かいっち しいちちん かいちょうかん ちょうちょう ちょうちょう ちょうちょう かっちょう しょうちょう

በተሰሙት ድርጅም አሳት ይቆይት በጥታቸው በይቀላይ ትርጉር እንደ የሰው በበት "

with higher frequency than in earlier III generation systems and it is important to have them handled with the minimum of intervention.

Figure 4-27 is a continuation of Example 1. The intent of this extract is to illustrate possible referencing of the system base in the virtual machine. Since IV generation systems will be unlikely to have an absolute addressing mode, segmented addresses must be used to address the system base even though these entries may use absolute addresses as pointers. [We have illustrated the access to the system base using conventional LOAD and STCPE instructions. This, of course, assumes that the layered ring mechanism is being used to control access to the system base. A special instruction might be used instead.]

The VMID contains '2' so VM2 is currently in execution. Assume that the active process, F1, is running in a privileged layer and thus may access the system base. Since P1's segment table lies in segment 0, the segment descriptor for segment 2 may be addressed as 01102. Both steps (g) and (n) which LOAD and STCRE a system base value, respectively, occur directly without interruption. This is because the VMmap f is distinct from the system base being modified. In a III generation software implemented VMmap f, both steps (g) and (n) would have to fault and be simulated. As we shall see, below, the VP [51] bermits (g) to execute directly but faults on (n).

[Below, we will discuss potential problems arising from the



FIGURE 4-27

and the first of said the set of the set of the set of the set

use of an associative memory which remembers the most recent map compositions. Under certair circumstances, unpredictable results may occur in both the real machine and in the virtual machine.]

Figure 4-28 is a further continuation of Example 1. VM2 is currently running and from f2, i.e., VMCB2, we see that only 6000 words of memory are allocated to it. The active process, P2, has a segment table which indicates that 9000 words of memory are allocated to the process. No validity check is performed when the system base entries are stored: they are verified only when used.

Thus, in the example, in step (e) application of the process map $\phi(213000)=8000$ does not cause a process exception. However, application of the VMmap, f2(8000) causes a VM-fault to the VMM. The VM-fault faults to the machine whose VMID is formed by dropping the low-order syllable from the current VMID. In this case, the fault is to the null identifier, or the real physical machine.

In order to speed up the operation of the crocess map 4, the real machine will likely use an associator. Similarly, operation of the virtual machine requires an associator to speed up the composed map f o ϕ . [See Section 4.5.] Figure 4-29 sketches these corresponding associators and values that would be stored after execution of the instructions of Figure 4-26. Running on



of starts in a low of the start of the starts and the second starts of the start of the start of the starts of たちないないないないないないというないないで、

ويوليد المالي المناسبة محالم سيعاد والمحالية والمحالية

ARTICLE CONTRACTOR OF CALLER

Associator — Real Machine		Process	P _i
After Figure 4-26 Steps	Seg #		Validıty,etc.
(ь)	2	2000	
(e)	1	4000	

Associator —	Virtual	Machine 1	

63.09 A 27 3 - 24

Section Section

Process P

After Figure	4-26 Steps (ر)	Seg #	Physical Address	Validity, etc
		2	500	
	(f)	1	7000	



والعادية المؤلف الألكان والكري والمعادين ما المراسم والمعادية والمعادية والمعالية والمعالية

the real machine there would be no steps (c) and (f),}

In both cases we assume that a change in the Punning Process dong, RPW, causes the associator to be cleared of entries. In the VM case, we further assume that a change in VMIC does likewise. In any event, a change by an active process to its map b or certain other parts of the system base, e.g., RPW, may cause unpredictable results (in the real or virtual system).

Example 2 - Recursion Acded to Example 1

In Example 2, we illustrate the effect of recursion on our previous example. The two level-one virtual machines are the same as before. However, in this example, a VMM runs on VM1. Figure 4-3. illustrates the VMmap fragment in the system base of VM1. After VM1 executes the 'LVMID 1' instruction, the VMIE row contains '1.1'. The prackets on the right size of the figure indicate the amount of memory resource owned by the real machine, VM1, and VM1.1.

When we pick up the action, process P1 of VM1.1 is executing a sequence of instructions similar to that of Example 1. [Note that P1's segment table entries are slightly different in this example.] After applying the process map $\phi(2153)$ we must apply f1.1 and then f1 until we finally man into physical resource names. The corresponding associator is also illustrated in Figure 4-30. Once again, segment numbers are mapped into physical locations. This time, the entries represent f1 o f1.1 o



TALLOW ON FORTHING AND

And a state of the state of the

「ころ」とうないとうなん、ちょうちょうできょうできょうできたか、ないに、いいないない

EXAMPLE 2: EXECUTION AND ASSOCIATOR

FIGURE 4-30

Solution of the second of the larger in the R-B recursive case than in the real machine or in the single level R-B case.

While we do not illustrate the other fragments from Example 1, they behave similarly. The process exception has a local effect and does not alert the VMM. The system base may be referenced in LOAD or STORE instructions without causing a VM-fault. Finally, a VM-fault by V.1.1 is directed to VM1.

Example 3 - Similar to Example 1 illustrating VF

In the next example, we contrast how the Verice Proposal [single level - no recursion] might work for the instruction sequence fragment of Example 1. Again, we assume two fixed partitions of the main memory for the virtual machines. Since the VP does not explicitly provide the map f [as the HV does] f must in effect be provided in software. What the VF does provide, however, is the composed map $\phi t = f$ o ϕ for all processes of all virtual machines. Therefore, the VCSCB's of the VP do not provide the memory, processor maps etc. Instead each VCSCB points off to the ROOT of a complete copy of the translated system base. Thus, in particular, there exists a path from a VCSCB to its processes' - CB's and corresponding segment tables.

In Figure 4-31, we abbreviate this structure by showing a dotted line pointer from the VCSCB to segment table. The contents of the segment table for process P1 of virtual machine 1



小学生がないないないない いったかい かいたいかい いいちょうかい いちょうかい

ANY MANAGER DE MANAGER DE LES

here a crossed the Rubbing this in the internet of the second

gives the absolute physical location of the segments, e.c. the map of = f o o [for level 1 VM].

Thus, in step (b) of the instruction execution, f1 o & is applied to 21500 yielding the physical location 5550 directly.

This ex: ole does not indicate reading or writing system base instructions, exceptions or faults since we have already discussed these cases in the earlier text.

Example 4 - Type I VCS, Paged Memory Map

The next example, Figure 4-32, illustrates how a paged memory map might be added to the VCSCE. We assume that the page olock size is 1000 words (which is also the minimum segment unit]. We do not illustrate fictitious resources or page faults. Although the map f1 is somewhat different from Example 1, its application produces identical results. See steps (a)+(1) for confirmation.

In the paged memory map, the associator plays a significant nerformance role. Figure 4-33 illustrates the possible form of this associator. Whereas the R-B associator did not differ structurally from the 'real' associator latthough the addresses stored were different], the paged associator has somewhat different structure and will likely be larger than the others [for performance considerations].

Note, once again, that this a sociator is different from that used in the 134 360/67 [68] or the Honeywell 645 [34,86]



FIGURE 4-32
Associator

After Figure 4-32 Steps

The second state of the se

And a state of the state of the

51602	Seg # −Page #	Physical Location	Status
(c)	2 - 0	5000	
(f)	1 0	7000	

EXAMPLE 4: PAGED ASSOCIATOR FIGURE 4-33

- ----

lik meterdi. Likikatakan di terukatan dara katan di dara kata kara kara karan dara karan dari karan dar

etc. since the f and ϕ map structure is different. The use of baging in Example 4 is for memory recource management only and does not contain any of the process structure normally found in baging maps. In particular, s. Figure 4-34 for an illustration of the difference. Thus, while intation, a software visible construct, is found in the ϕ map, paging, a resource allocation mechanism, is best added to a system as part of an f-map. The use of this approach implies that complete optrating systems will be runnable without modification in a paged environment. This result should be contrasted with OS/360 and TSS/361 [10]. See Section 4.8, below.

Example 5 - Example 4 with Pecursion

In Example 5, Figure 4-35, we add recursion to the paged memory map of Example IV. As can be seen, this example is very similar to the recursive P-B example of Figure 4-33. The principal apparent difference is the form of the associator. However, because of locality considerations, is discussed above [in Section 4.5] acceptable performance should result. Thus, we have shown that [unlike VP] recursion does not introduce idditional complexity, even in the case of a complicated memory map such as paging.

all and a second

a historicania develición na successiventa



645, 360/67: One page map per segment



HV vs. 645, 360/67

Souther and State an



いたいので、ことになったが、ことには、

FIGURE 4-35

Example 6 - Semaphones in a Vintual Machine

The intent of this example is to further demonstrate that the virtual system base may be used directly in most instruction executions without the intervention of the VMM. We will itlustrate now firmwore dispatching and operations on demaphores might work in a virtual machine. [We will assume the reader is familiar with semaphores and Dijkstra* P and V operations [41]. This is crucial to the operation of a IV generation VM since IV generation *Jobs* can be assumed to be made up of a number of cooperating sequential processes [87].

The previous examples of this section have illustrated that baged or other memory maps in the VMCB introduce no additional complication that cannot be solved as in the R-B case. Thus, we will illustrate semaphoras with an R-B map.

First, we illustrate semachcies and process dispatching on the real machine. In Figure 4-36, frame (a) shows the queue of ready processes pointed to by the system base. Each (circle) antry in the queue represents a process link for some active process in the system. These links contain the process name and, so, effectively a pointer to the corresponding FD3. The queue is prganized, presumably, in priority order, although we do not illustrate on concern ourselves with the priority field.

In frame (b), the first process is nemoved from the D-Peady processes by the firmware dispatcher. Its (process) identifien is set in the Running Process word, PPW. Let us assume that process PI frequites a P(sem) and the semachone count of sem is C. Then P1 is blocked and but on the queue of semaphone sem. The

1.1.1

and succession and the second

and the second second

ren har an subscripter and the reaction of the

and the second second



FIGURE 4-35

next process, P2 in the figure, is made active [Frame (c)]. Process P2 executes a V[sem] which removes P1 from the semaphore queue and returns it to the O-Ready processes [Frame (d)].

All of these manipulations of the system base are normally carried on entirely by the firmware [79]. We will show that this can be the case in the virtual system as well.

Figure 4-37 provides a more detailed picture of the system pase illustrating the semaphore tables and process links. Each PCB points to a semaphore table which defines the semaphores for that process. If the semaphore exists, then the semaphore table entry is a pointer to the semarhore header for that semaphore. In Figure 4-37, the semaphore "sem" is called setaphore number 3 by process P1 and semaphore number 4 by process P2.

The semachore header contains the count and, if negative, a pointer to the top of the semaphore queue. The cointer is given as a relative displacement to the origin of the Active Process Link Table. Subsequent queue entries are linke together similarly. Note that this is one cossible v-ry simple for semaphores that may be used in this illustration of the machines. We do not claim it to be the rost of the organization.

Figure 4-37 represents the system base bef, for the executes ite P[sem] instruction. The executes of P[sem] instruction the b at f maps as indicate in the commentary. Finally, Figure 4-38 shows he system witten the P[sem] instruction. Execution of the suprecuent visem] instruction by process F2 may be traced in a similar manner.



+ - a man hard and the state of the state of the



Example 7 - Type I MCS, Multiple Processors

This example illustrates the extension introduced by multiple real and virtual processors in a computer system. For the purposes of the example we assume that all processors are identical CFU's. ECIcarly this work can be extended directly to non-timing-dependent I/C processors as well [15,17,96].]

Figure 4-39 shows the modification to the VCSCB. The processor map now points off to a virtual processor table, each entry of which gives the (instantaneous) map between virtual and real processors. Also the LVMID instruction must be expanded to include a second operand providing the processor number as well as the VCS number.

Thus, when a physical processor, say, processor 3 executes *LVMID 1,1*, the VMID for the processor is set and real processor 3 is set in the processor map as corresponding to virtual processor 1. Later, when virtual processor 1 executes *SIGNAL 2* the instruction executes directly by looking up the corresponding real processor value. When *SIGNAL 3* is attempted, since the map value is invalid, a pending bit is set. [A VM-fault may also processor 3 ...]

Note that this example illustrates just one way that the f map may be introduced for multiple processors. The virtual-real processor index is a kind of page map and the dispatching of the VCS without all processors assigned is an example of a fictitious

A Contraction of the second second

and the second and the second of the second s



EXAMPLE 7: MULTIPLE PROCESSOR'S PROCESSOR MAP

FIGURE 4-39

a service and the

resource. We can imagine other kinds of processor f maps. For axample, with a relocation-bounds processor VMmap, an induced real processor number is obtained by adding the relocation to the virtual processor number after checking that the bound is not exceeded. One can imagine a system organized around this map requiring all virtual processors to be assigned by start-up, i.e. no fictitious (dynamic) resources.

Example 8 - Type II VCS, Single Processor

In this example, we give one possible interpretation of a IV generation Type II VCS. To show the operation, we return to the instruction sequence from Example 1. Our system base now has the VCSCB hanging off the PCB (of some particular privileged process). We have the same memory, processor, etc. components of the map. However, now virtual resource names are mapped into process names. Thus, the memory map illustrated takes virtual memory addresses into segments. In our example, we map addresses into displacements within segment 1 of process P1. Steps (t)=(d) successively apply the process map of the real machine, 4r. See Figure 4-49. The a generation of the map form as that used in Example 1.

machine, in See Figure 4-49. The a complete the be of the same form as that used in Example 1. As seen before, in Chapter is the rest of the a number of advantages to running a Type II VMM. The secretion system, a Type II VMM permits the predor part securiting system to run



EXAMPLE 8: TYPE I VCS FIGURE 4-40

Be with a state of the and the second and and the second and the s

without degradation due to the VNmap. Furthermore, it simplifies the construction of the VMM since the host operating system takes over the allocation of real resources. In this example, virtual nemory is being mapped into segments of a process running on the real machine. Therefore, the allocation of TETORY for these segments and the response to segment-absent exceptions may be nandled standard by the pritsrage system allocation TETORY procedure.

the submitted distance with the state of the submitted of the submitted of the submitted of the submitted of the

4.7 SOME PRACTICAL PROBLEMS OF IV GENERATION HV

There are a number of practical problems that prevent the hardware virtualizer, as described, from providing the complete implementation of a IV Generation VCS. Some of these difficulties, e.g., time, are intrinsic to virtual computer systems. Other difficulties, such as the ad hoc unformatized time-dependent I/O process, are technological and may pass as we gain greater understanding of computer systems architecture. Until such time as these problems are overcome, we can still rely on III generation, software VMmap techniques. He will illustrate how an interim solution can be developed for I/C.

The difficulty with I/O lies chiefly with its control, nct with memory or device mapping which may be accomplished via the VCSCB. The interim solution we opt for directs I/O completion interrupts to the real machine. Ke make the Connect (Start) I/O instruction (and all other I/C instructions) absolutely or vileged such that it may only be issued by the physical machine and attempted execution by a VM causes a VM-fault to the VMM. As with III generation systems, the VMM translates the request and issues the I/O request itself. 0n the I/0 termination, an inturrupt occurs to the VMM running on the absolute machine. Regardless of the VM being run at the time of the interrupt, its status may be saved, and control is returned to the virtual machine (by the physical machine) after interrupt processing. The VCSCB is expanded to include an I/O interrupt bending bit which is set by the VMM. Execution of the LVMID

instruction to start up a VM while the pending bit is set for that VM, causes an I/O interrupt to occur to that VM.

While the asynchronous I/O interrupt goes to the absolute ohysical machine, the connect instruction (CIO) VM-fault is still a conventional HV fault up one level to the VMM. This is an improvement over the III generation handling of I/O in which the connect (start) instruction (via the software f) traps to the ohysical machine. The real trap handler must reflect the trap pack down to the appropriate level VM, running the VMM. Thus, even for software I/O map, in the IV generation HV, recursion is much easier and more natural than in III generation systems. See Figure 4-41.

Another suggestion for I/O control can be perived from Lauer and Snow [76]. Easically, they suggest allowing the virtual machine to issue I/O instructions subject to the device map in the VCSCE. I/O completion interrupts then come in to the real machine which passes the information down the VMM chain via so-called "spurious interrupts."



vendintantilittetta", Alleh

È.

INTERIM I/O PROPOSAL

FIGURE 4-41

4.8 II AND JIT GENERATIONS PEVISITED

LEAST CALL AND DESCRIPTION OF THE PARTY OF THE

The work of Chapter 4 has been directed toward introducing the HV into IV generation systems. We have assumed that IV generation systems define an architecture with a process mac \$ implemented in firmware. We have then expanded the design to form a new system which incorporates a level invisible firmware VMmap, as well.

We have assumed that ϕ is a rather complex map. However, the principles we have developed are completely independent of the complexity of ϕ . Thus, it becomes possible to examine other computer architectures in which the maps f or ϕ may be of different form from those considered. The important properties which must be preserved, however, are that ϕ corresponds to the software visible structure within a level of a virtual machine whereas f corresponds to the map <u>betweer</u> two levels of virtual machines. We will consider two representative examples.

4.8.1 f=R-B,0=identity

And indicated and the indicated the character the test

If we let &=identity map, then we have a computer design without intra-level structure. There is only one mode, i.e. "Supervisor state is not necessary" [76]. There is no memory napping within a level. The level structure looks very much like 9 first or second generation design of the sort found in a CEC PDP-8 class minicomputer.

The introduction of an P-B f-map allows us to develop virtual machines for this system. [As in Section 4.5.] Because

of the lack of structure in each machine level, it may be desirable to allow programs to communicate between levels by causing a VM-level fault with a conventional bounds violation This now defines a programming environment in which the code. VM-level fault is used to call on supervisory services, much as the "diagnose" instruction in CP-67 [2,23,65]. In Figure 4-42, the VMM running at level n-1 produces a VM at level n. At level run c "two-level operating system" OS [in the figure], **n** , we which produces this enhanced environment at level n+1. The system will work as long as level J to level n-1 each run the VMM. This is true because of VM-recursion and level invisibility of the f-maps.

The "f=R-B, = identity" machine is essentially that proposed by Lauer and Snow [76]. They observe:

> more general than most existing hardware in the It is sense that it naturally encourages a hierarchical structure without imposing artificial boundaries or an omnipotent supervisor. But it lacks some of the important features of modern systems, particularly segmented virtual memories and a suitable parameter passing mechanism ... [76]

The introduction of intralevel structure, e.g. process model and segmentation, can best be accomplished using the techniques developed earlier in Chapter 4. However, the "&=idertity" machine may still be a useful system to construct, particularly in conjunction with a line of minicomputers. In this cortext, poth f=R-E and f=paging could lead to interesting results.





4.8.2 f=baging, d=supervisor/problem-state [no memory mapping]

The work of Chapter 4 indicates how baging could have been ...to an IEM System/360 Virtual Computer System. introduced Instead of baging being a part of the visible intra-level structure. it should have been introduced as part of a formal f-map. Under these circumstances, there would be no need to rely on the techniques discussed in Chapter 3, 3 "software i.e., V"'map". Privileged instructions would be able to execute Iwithout the dir activ trap simulate discatch sequence], exceptions istra.level Would be directed to the privileged scriware at the current level. recursion hluce be vastly simplified, and CP-67, tne VMM, i.e. bluck be greatly streamlined. Once again, a key point is that paging, a mechanism for resource allocation, should properly be part of the formap, not the d-map.

CHAPTER 5.

5.0 SUMMARY

The thesis has developed appropriate terminology, an empirical basis, and a model which represents the execution of processes on a VCS. The model features two maps: (1) a process map called & which maps process names, e.g. segments, semaphores, process-id*s, into resource names, e.g. memory locations, processor numbers, and (2) a vertual machine map (VMmap) f which maps virtual resource names into real resource names. The process map is strictly an intra-level map expressing relationship within a virtual machine; the VMmap is an inter-level map expressing a relationship between (the resources of) two adjacent levels of (virtual) machines. Thus, the action of running a process on a VCS consists of running it under the composed map f o 0. Application of the model has led to a clear interpretation of virtual machine recursion, Type II virtual machines, and other important properties of virtual machines.

The model also allows us to understand the implementation of III generation VCS's as the software construction of the VMmap f, utilizing parts of the hardware process map **b**. However, the most important result of studying the model is that it leads directly to the design of a hardware virtualizer for proposed implementations of IV generation VCS's. The design has the

characteristic that all process exceptions are handled directly software intervention. A11 within the executing VCS without rescurce faults (VM-faults) by a VCS are directed to its VMM on knowledge of processes on the VCS. the real machine without Fault handling, invocation and execution of VCS's works directly regardless of recursion. Furthermore, preliminary performance studies indicate that this design will provide performance of the virtual machine comparable to the real machine for the likely choices of the maps f and d.

was in the market of the second second of the second of th

al 2 services the state of the solution of the solution of the solution of the solution and the manual

5.1 SUGGESTIONS FOR FURTHER RESEARCH

The development of the principles reported in this thesis has also pointed out a number of areas for future research. The model and suggested implementations of Chapter 4 have been directed largely toward virtualization of a single processor [without I/0] computer system. Example 7 of Section 4.6 sketches an approach to multiple processor configurations and Section 4.7 suggests an interim solution to I/O. However, there is still nuch work to be done in incorporating either of these features into the model and the hardware virtualizer. As Section 4.7 points out, progress in the treatment of virtual I/O is somewhat tied to progress in the treatment of real I/O. As long as timing dependencies and asynchronous interrupts exist in I/O systems, it will be difficult to treat the virtualization of I/O in homogeneous manner with the rest of the system.

Another area for further development of architectural principles concerns machines with reloadable control store [97,98]. The of the thesis has been directed to work virtualization of interior decor, and, as part of the definition of a VCS, we have assumed the existence of some "native mode." However, the whole concept of interior aecor in future systems is likely to be extended by reloadable control store. The incorporation of such machines into the theory will have an important impact upon preparation and extension of code for the microprocessor.

The study of VCS°s seems to provide an excellent meeting

pround for a number of traditionally separate disciplines of computer science. One area, in particular, concerns the development of models which provide an interior decor level description of the execution of processes on a computer system. The empirical work of Chapter 3 has provided an essential foundation upon which to create a useful abstraction of program execution in a III generation VCS.

The thesis has been concerned with techanisms for implementing VCS's, rather than specific policies for applying to them. Thus, we have not discussed issues of resource allocation. performance, or throughput as applied to the real or virtual systems. This includes the introduction of fictitious resources, i.e. demand paging, or particular multiplexing algorithms to be used in a VMTSS. It is ouite important to develop models that will make possible reasonable resource allocation decisions. One aspect worthy of study concerns treatment of the process interval timers and the firmwara process dispatcher. Should the virtual machine dispatcher be implemented as a firmware adjunct or will software be sufficient? How does this choice depend on the intended use of the system, i.e. many or few virtual machines. Another auestion concerns possible conflicts and counterproductive algorithms between different levels of virtualization. For example, if the 0-map is segmentation and the f-map is paging, then memory compaction by the virtual machine may be a sub-optimal policy. Another example hight involve "double spooling" -- once by the virtual machine and once by the VMM. Performance models and studies will be a necessary

part of future VCS theory and implementation.

Contraction of the second s

Contraction of the second

and a stand of the second second in the second second is the second

575434

2.50 2025

APPENDIX A TUTORIAL ON CP-67

A.0 Introduction

いたいたというため、たいに見たいです。ためにたいが、たいいうこのにたたというかのに

In this appendix, we shall illustrate the typical software implementation of virtual machines on third generation systems by reviewing the mechanisms used by CP+67 [9,32,54,85].

In order to faithfully produce a virtual 360, CP-67 must provide a virtual processor, virtual memory, and virtual I/O charnels and I/O devices. The basic approach that CP-67 adopts is to handle high frequency events in an automatic manner with little, or no, overhead, while handling infrequent events in a less efficient manner. This rotion loads CP-67 to a rather efficient handling of virtual processor and virtual memory. Virtual I/C events which occur at a much lower frequency may be processed with greater overhead.

CP-67 is able to simulate a 360/67 in both of its modes of operation. In one mode, the 360/67 behaves as a 360/65, a standard 360 without memory mapping. In the other mode, called by IRM "extended mode", the 360/67 uses a number of additional registers and instructions. In extended mode, the 360/67 has the ability to enter relocate mode. The 360/67's operation in extended mode, but not relocate mode, is very similar to the operation of a 360/65.

A.1 CP-67's Virtual Memory

「ないたかないが、「ないないない」というないないないないないないないです。

The simulation of virtual memory for virtual machines is perhaps the easiest of CP-67's chores. The 360/67 has a memory relocation mechanism, called DAT (Dynamic Address Translation), which features a segmented-paged address space [112]. The 24 bit virtual address produced by the 360/67 in relocate moce is interpreted as a four bit segment number; an eight bit page number, and a twelve bit displacement within the page. Segment and page tables are located in core.

CP-67 always runs virtual machines in relocate mode. Thus, the paging mechanism theoretically permits virtual machines to address a memory of up to 2**24 or 16 million bytes. This virtual memory may far exceed the amount of physical memory actually available. In practice, however, for reasons of afficiency virtual computer systems are normally configured to have a significantly smaller virtual memory, usually less than 1 million bytes.

The paging mechanism may be used directly as long as the virtual machine, itself, does not attempt to execute in (virtual) relocate mode. If the virtual machine, e.g. 350/67, goes into relocate mode, it is necessary to map the relocated addresses of the virtual machine into their corresponding real addresses. This facility can be provided with reasonable efficiency via a software algorithm as described in Goldberg [54] and illustrated in Figure A-1.

In Figure A-1, the page map (C) provides the correspondence between addresses in the relocated virtual memory (3) and the



日本により、「日本」に、日本には、日本になった。

Location of the first of the second second



いたためなどのないであるとう

MAPPING RELOCATED VIRTUAL MEMORY TO REAL MEMORY FIGURE A-1

(nor-relocated) virtual memory (C). Thus, relocated virtual memory page 2 is mapped into virtual memory page 7. Similarly, the page map (E) provides the map from virtual memory (D) to real memory (F). Thus, virtual page 7 is mapped to real page 3. Since all addresses must ultimately be mapped into real addresses and the 36C/67 hardware provides only a single level of page mapping [this example ignores the "segmentation" of the 36O/67], it is necessary to use software to derive the composed page map (A) which maps directly from (B) to (F). Therefore, when the virtual machine that is illustrated executes in (virtual) relocate mode, relocated virtual page 2 is mapped into real page 3.

NAME TO AND A CONTRACTOR

A.2 CP-67*s Virtual I/C

The simulation of virtual I/O channels and devices is berformed in a number of different ways by CP-67. Virtual machines may be configured with I/C devices that are realized as dedicated, spooled, or shared devices. A dedicated device is a device such as a tape frive or an entire disk pack which may be used by only one virtual machine as a time. Consequently, tape drives are attached to some virtual machine for a given duration. The virtual device address need not be the same as its physical address. A correspondence between the two is established at the time that the device is attached.

A shared I/O device is a physical device which may be simultaneously part of the configuration of a number of different virtual machines. A shared I/O device is divided among each of

y, ity with a statistic for the state of t

the different owners in such a way that each virtual machine believes that it alone owns the device, but that the virtual device is smaller than the actual physical device. An example of a shared device is a disk pack that has been divided among several virtual machines. These so-called <u>mini-disks</u> behave exactly like the ordinary IBM 2311 disks except that they have fewer cylinders. Another example of a shared I/C device is a transmission control unit, such as the IBM 2792, in which each virtual machine may have only a small number of the transmission lines into the device. Such sharing artifacts, like the are introduced for or "efficiency" mini-dist. cost considerations. They improve the sharing of large resources that might otherwise be ineffectively utilized.

and the second secon

Spooling is the third method which CP-67 uses to simulate I/C devices. Each virtual machine has attached to it, a virtual card reader, virtual card punch, and virtual line printer. Rather than provide a separate real device for each virtual machine, CP-67 utilizes buffered disk files to simulate these devices. On output, data directed to the virtual printer or bunch are written on a special disk file. When the real physical device becomes available, the temporary file is crocessed all in one piece, Similarly, card input is first read in by CP-67 and written on a disk file. Card images are then crovided or each virtual I/O read request.

Dedicated I/O device simulation is the most general approach used by CP+67 and may be used for devices which are unsupported, or supported in other ways. Thus, alien devices, such as the IBM

2250 graphics console [83] may be connected providing that the virtual machine contains the appropriate I/O mandling support. CP=67 need only provide very simple interrupt henching. Devices, such as the reader, ounch, or printer, which are normally scooled devices, may be attached to a single virtual machine if desired. Devices, such as disk packs or telecommunications control units, which are normally supported by CP=67 through sharing, may be dewiceted to a single virtual machine device basis. Thus, CP=67*s handling of I/O devices is quite flexible and allows for configurations which are somewhat different from the actual resources available.

「さんなるなどをになるとなる」ことになっていたのではない

Since a virtual device address may differ from its corresponding physical device address (that is used to realize it) it is necessary to perform some kind of device mapping before I/0 transfer may take place. The device mapping 3ก. is illustrated in Figure A-2. A virtual device is located on some particular virtual channel and virtual control unit. Once the device correspondence is made with the real I/O device, its particular real control unit and channel may be identified. 1/0 transmission requests, CCW (Channel Command Word) chains, must be translated to reflect the corresponding real I/C sevice involved the physical location in core memory). The CCW (as well as translation is done and the I/C request is placed on the queue for the real channel. When a completion interrupt comes in on the real channel, this information is posted back to the corresponding virtual channel.

Because of the recessity of translating CCW chains before



I/O initiation, CP+67 introduces the additional restriction that no channel program is changed by the CPU or the channel during the interval between execution of the START I/O instruction and the channel end interrup1 except that performed by the OS indexed sequential access method (ISAM). The restriction arises because CP-67, having translated the channel program, will have no way of knowing that it has been changed and should be translated again [8].

The actual mechanism that allows CP-67 to gain control before a virtual machine can start up I/O will be explained below in the section on the virtual processor.

A.3 CP-67's Virtual Processor

 C^{p} -67 simulates a virtual processor, e.g. 363, by using the virtual machine mechanism that was suggested in Chapter 1 and Chapter 2. Since the virtual processor and real host are similar (or identical), the virtual processor's opcodes may be executed machine. instructiondirectly on the host Thus, su by-instruction interpretation is not required. However, it is necessary to prevent certain instructions from executing directly on the host machine. These instructions if executed directly by virtual machine serious difficulties can cause in interpretation or control. Any instruction that will rot be permitted to execute directly on the host machine is termed a sensitive instruction.

The approach that CP=67 adopts to prohibit the direct execution of sensitive instructions is to run the virtual machine

always in (chysical) proplem state on the host machine. Recall that a real 36j has two modes of operation: (1) supervisor state, in which all instructions are permitted to execute, and (2) problem state, in which only a subset of the instructions are allowed direct execution. The attempt to execute a privileged instruction (one which may only be executed in supervisor state) while in problem state causes a program interruct, i.e. trap. Bv running а virtual machine only in problem state, CP-67 effectively prevents the direct execution of all privileged instructions. If the sensitive instructions are privileged then CP-67 has succeeded in creventing their direct execution.

CP-67 maintains a number of tables in its supervisory area. Among them are copies of all working registers that the virtual machine believes that it has. The registers include a virtual program status word PSW. This register indicates the state that the virtual machine believes that it is in. If the virtual machine is in (virtual) problem state and a privileged instruction is attempted, CP-67 reflects this fact back to the virtual machine. The virtual machine is cut into (virtual) supervisor state so that it may process the crogram interrupt The real host machine still remains in problem state itself. while the virtual machine is running. If the virtual machine is in (virtual) supervisor state and a privileged instruction is attempted, CP-67 recognizes that this would be a valid instruction on the real machine and simulates the effect of the instruction.

Some of the privileged instructions of the 360 which CP-67

nandles in this manner are the state charging instructions Load Program Status Word (LPSW) and Set System Mask (SSM), the storage protection instructions Insert Storage Key (ISK) and set Storage Key (SSK), the I/O instructions Start I/O (SIO), Halt I/O (HIC), Test I/O (TIO), and Test Channel (TCH), some special instructions Diagnose (DIAG), Read Direct (PDD), and Write Direct (kRD), and the special 360/67 instructions (for relocation) Load Multiple Control (LMC), Store Multiple Control (STMC), and Load Real Address (LRA) . Since the Start I/O (SIO) instruction is privileged, the attempt, by a virtual machine, to perform I/O is immediately recognized. It is at this point that CP-67 may perform the mapping betweer virtual and real I/O devices (as **Jescribed** above) and then initiate the corresconding I/O operation itself. The storage protection instructions a∿e orivileged and this allows CP-67 to keep a copy of a virtual machine's protection keys and use them when the virtual machine is dispatched. Likewise, the instructions which start up the relocation system are privileged and so virtual relocation may be mapped like virtual I/C [54].

Thus, CP-67 simulates a virtual 360 processor through a combination of hardware and software. It insulates the (real) host machine from actions of the virtual machine by always running the virtual machine in problem state and in relocate mode. Because the host is in problem state, it is immune to tampering by sensitive instructions. Because the host is in relocate mode, certain reserved registers in low physical core are isolated from the virtual machine. The actions of the clock
and these interrupt locations are simulated for the virtual machines by CP-67.

Same of Carry Strate States . 5 . 4

する状態をたいで

the Kon With Station

Page :98

APPENDIX B

CASE STUDIES OF SOME III GENERATION MACHINES

3.0 Introduction

In this appendix, we provide a number of examples of third generation machines which are and which are not virtualizable. For the machines which are not, we indicate which of the empirical rules of Section 3.2 are violated. These results are summarized in Figure 3-2. This set of examples is not meant to be an exhaustive survey. Rather, it is intended to give the reader representative examples of problems that one encounters with contemporary machines. Indeed, it appears that whether or not a particular nost supports virtual machines is merely a matter of chance. This is due to the fact that one of the header the supporting virtual machines.

3.1 IBM 36 /07

CP-67 has been in existence for several years now, so there can be little doubt concerning the 363/67's ability to support virtual machines. The initial CP-67 ran on a 353/67 out produced only virtual standard 363's, e.g. 360/35. Under these conditions, CP-67 was not self-virtualizing and so not strictly under consideration here. The extension of CP-67 to support a virtual 360/67 makes it self-virtualizing and appropriate for study here.

alfagetas un ferese and a ferre and a ferre and a ferral a

CP-o7 is a Type I VCS which is a timesharing system, i.e. VMTSS. It produces virtual machines using the basic mechanism tescribed in Argendix A. The 360 has two narcware modes of operation and so it is appropriate to apply the rules developed in section 3.2. Rule 1 is satisfied since the instruction execution is identical for non-privileged instructions in both supervisor and problem state. This is a characteristic of al! 36.'s, not just the 300/37. Rule 2 is satisfied since the 360/67 JAT relocation system may be used to isolate the virtual machine from the VMN. Rule 3 is satisfied since sensitive instructions cause recoverable traps. In particular, in the 360/07, 311 of the instructions which can cause problems are privileged. Thus, Rule 3a is satisfied by LPSW, SSM, SVC, DIAG, LMC, STMC, LRA [as explained in Appendix Al. Rule 3b is satisfied because the relocation system makes the hardware's reserved core locations inaccessible to the virtual machine. Again, the instructions which reference the reserved registers, such as the PSW or Control Registers are privileged. Rule 3c is satisfied because sensitive instructions and privileged. the appropriate Furthermore, the use of the memory relocation system in "managing" the virtual machine does not introduce additional Insolvable complications. Thus, relocation of virtual namory may occur without error. Any traps that occur in reconciling virtual addresses with real addresses or even relocated virtual addresses with real addresses do not cause unrecoverable arrors.

Two points are of considerable interest in this discussion. First, the 360/67 prechecks instructions which may cross page

boundaries. Thus, instructions which may cause difficulty, such as the Add Becimal instruction, are suppressed before execution rather than terminated during it [54]. The second point is that, in reconciling relocated virtual addresses with real addresses, the DAT mechanism may be set to trap until this memory happing is correctly established. Rule 3d is satisfied because the various instructions which reference I/C are privileged.

Thus, not surprisingly, the 360/27 satisfies all of the ampirical hardware requirements stated in Section 3.2.

3.2 IBM 360/65

energia and a submark of the second second second second second second second second and a submark second secon

In this section we discuss the problem of implementing a OP-(3 system. CP-53 is a VCS which runs on a 350/63 to produce virtual 360/65°s.

As was mentioned above, a standard 350, e.g. 360/45, satisfies VCS handware requirements one and two. With regard to Rule 2, the 360/65 has no relocation system so that it must rely on its storage protection system to protect the VMM. It also must use the protection system to prevent instructions from referencing the special system-wide registers (Rule 30) stored in low core since, in general, they may be accessed without privileged instructions. Rules 3a, c, d are satisfied because the relevant instructions are privileged. Thus it is sufficient to see if the storage protection system is adequate to satisfy conditions 2 and 3b.

Since we are using the protection system to protect queries as well as ulterations, we are led to the conclusion that we need

the 36.*s five bit key store-and-fetch protect system. With this fecture each 2K byth half-page is assigned a storage key via the orivileged SSK instruction. Four of the pits form a store protect key and the fifth bit may be set to indicate fetch protect as well. If the four bit key in the PS4 corresponds to the four bit key assigned to a 2K block, then the task is permitted to fetch or store into the block. If the keys co not match then the task has only fetch access to the block unless the fetch protect bit is also set. If the key in the PSW is zono then the task has fetch and store access to any block of storage regardless of the Key stored with it.

The second and a second second second second and the second second second second second second second second se

Thus, for CP-63, a zero storage protection Key in the PSH is inalogous to being in supervisor state. It gives the holder more bower than he may be able to use wisely. It is too much power to give to a virtual machine and so the virtual machine's key in the PSH must be some other value. Only the supervisor may actually run with storage key zero. The virtual machine runs with storage key "virtual zero" in which the supervisor performs a key translation for the virtual machine.

While this scheme adequately limits virtual manory and protects the supervisor, the dedicated locations in low core (the first 2K block) cause it trouble. Since there is no relocation scheme on a standard 36, and the machine can generate and store absolute addresses, the virtual machine runs unrelocated. Thus, the virtual machine's addresses .-2K correspond to the real addresses -2K. In order to prevent the virtual machine from accessing the interval timer, new PSW's etc., the first 2K block

nust be fetch and store protected from all virtual machines. In order to preserve the notion of a virtual machine, there must be 3 set of corresponding virtual special registers stored somewhere alse. These may be kept in upper core. Attempts to access registers will be trapped by the storage protection mechanism and the supervisor may simulate the effect. This mathod of using the protection keys to prevent the virtual machine from executing sensitive instructions works. Unfortunately since the protection system on the 36g involves 2K blocks of storage, the core above byte location 128 (decimal) in block zero is protected unnecessarily. As a result any instruction in core location tess than 2K or any instruction accessing data in cone location leas than 2K, will be trapped autometically. This will occur for all instructions, not just for the sensitive ones. Consequently in order to preserve the virtual machine notion, the supervisor must simulate each such instruction. This can cause a significant loss in efficiency.

Actually, there is at least one somewhat unlikely condition under which such an instruction may not be correctly simulated. If a storage-storage type instruction, like AUD DECIMAL is interrupted <u>during</u> execution, due to a protection exception, the instruction is terminated and partial results may be lost.

3.3 HILAC SAUL

The HITAG 84JD is the Japanese equivalent of the KGA Spectra 70/45. As such, it is very similar to a standard 36, e.g. 36J/65, and so, the preceding discussion is completely

applicable. It satisfies the empirical nardware requirements in the same way that the 363/65 did, above.

3.4 IBM 36./35

The IRM 36./85 is a standard member of the 35 family. Its interior decor is similar to the other standard models, e.g. 360/65. Its principal structural change from the other models is the introduction of a small buffer memory called the cache. Since the cache is invisible at the machine architecture level, i.e. not part of the interior decor, it need not be virtualized, and need not even enter into virtual machine considerations. Since any standard 369 may be virtualized, the 350/85 may pe.

3.5 Data General NCVA

The Data General NOVA and SUPERNOVA are sixteen off dimicomputers. Recent work by the author and L. Dickman has red to the design of a VCS for the NOVA. The vintual NOVA is facilitated by the memory allocation and protection obtion that is available. This option introduces paying and two processor modes-- supervisor and user. Since all instructions which manipulate state information, e.g. the page table registers, are "I/G instructions" and all I/O instructions are privileged, Rules datable, and d are satisfied. Rules 1 and 2 are similarly patisfied. Therefore the NOVA is vintualizable.

3.6 Honeywell DOP 516

The Honeywell DDP-516 is a 16 bit mini-computer with two

nandware modes of operation, a store (but not fatch) protection system, and no relocation system. The machine cannot be virtualized for a number of different reasons.

Rule 3 is generally violated. Some of the sensitive instructions are trapped, either as a result of the privilege mechanism on the protection mechanism. However, for those instructions, the trap is deferred one full instruction execution. Thus, the state of the machine will be changed before the supervisor gains control. In fact, it is even possible for the instruction following the sensitive instruction to store into that instruction before the trap has occurred. Now, the VMM has no hope of simulating the instruction.

NILLOH DISLOSING

Rule 35 is violated in a monner similar to the PDP-10. [See below.] The instruction used to return from supervisor state to problem state is ERM (Enter Pestricted Mode). Unfortunately, this instruction is not privileged, Thus, its execution in virtual supervisor state will not cause a trap. A virtual state change may occur without the knowledge of the VMM.

Rule 3b is violated because of the lack of a relocation or fetch protect system. This difficulty by itself need not be fatal since the introduction of a suitable restriction can still produce a useful virtual machine (as in Fuchi [+9]). However. Rule 3b is violated in another more severe manner with the IMA (Interchange Memory and Accumulator) instruction. If IMA is attempted into an area that is protected, the accumulator will be loaded before the protection violation is signallod. Consequently, it is impossible to recover the original contents

Page 205

of the accumulator and simulation is not possible.

3.7 GE 635, b55 (Honeywell 6...)

The Gc 635 cannot be virtualized because of violations of Rules 3a and 3c. The LDT (Load Timer) and L3AR (load 9358 Address Register) instructions are sensitive instructions that hull execute rather than trap when issued in problem state (slave node). The GE 655, a successor to the 635 attempts to correct this error by providing a trap on LDT or LBAR in problem state. However, there are still difficulties with the machine decause of visibility to the physical master/slave node vis the hon-privileged store indicators instruction.

3.8 Multidata Model A (SEL)

The Multipata Model A is a 16 bit mini-computer with a hardware implemented paging system. A virtual Rachine system is impossible since Rule 3 is completely violated. All privileged instructions execute as NOP when in problem state. These instructions include the status switching instructions and I/O instructions.

3.9 XDS 946

she hield with the ministeries and and and the the ministeries of the destination of the states between the states of the states o

The XDS 940 cannot be virtualized because of a fundamental violation of Rule 1. Bit 0 of each instruction word indicates whether or not the memory mapping is to be on for that instruction. The bit has effect only in supervisor state but is ignored in problem state where it is assumed that all addresses

are mapped. Thus, running the virtual supervisor state as physical problem state does not provide a mechanism to distinguish between mapped and unmapped addresses.

3.1. DEC PDP=1u

The DEC PDP-13 may not be virtualized because it victates Rule 3a. The JRSTF (Jump and Restore Flags) or "JRST 1," (Jump to User Program) instructions are used to return from supervisor state to problem state. When executed in (Shysical) problem state, the instructions do not trap. Thus, it is impossible to letect a virtual state change.

Another violation of Rule 3a concerns the accessibility of the "flags" which indicate what physical mode the machine is in-Thus, a program in virtual supervisor state might discover from the flags that it is in (physical) problem state and become "confused".

As with several other machines which violate Rules 1 or 3a, it is possible to construct a hybrid virtual machine on the PDP-1. [See Section 3.3.] The HVM is possible since an attempt by a virtual machine to enter supervisor state traps to the VMM. The means of entering supervisor state are via a (privileged) UUD. Thus, an HVM, Type I or Type II, is possible on the PDP-10. The organization of a Type II HVM is discussed in Appendix C.

3.11 BEN TENEX

TENEX [2.] is a PDP-1, that has been modified to include nardware paying. This modification has not changed the situation

with respect to the JRSTF instruction. Thus, TENEK is not virtualizable but an HVM may be constructed.

27 2 24 5 Start Back

÷ż;

A TRACTA

ないというでいまたというないでいたとれるかというない

うりがいたいたちは

WINKS

APPENDIX C

CASE STUDIES OF THIRD GENERATION TYPE II SYSTEMS

C.O Introduction

Man Musical Association of the A

STRATE AND A CHARTER AND A

In this appendix, we consider examples of several different operating systems which do and do not support Type II (extended machine host VMM's. For the machines which are Type II virtualizable, we indicate the modifications that had to be made to support the empirical software requirements. For the systems which are not, we indicate the empirical rules which are violated. These results are summarized in Figure 3-8. The PDF-13 did not satisfy the hardware requirements but the ITS operating system satisfies the software rules. Consecuently, ₩e are able to exhibit a Type II hybrid virtual machine for this system. As with the hardware examples of Appendix B, the operating systems investigated in this section were not designed with the thought of supporting virtual machines.

C.1 HITAC 8400

The HITAC 8400 [49] has a Type II virtual machine which runs under the TCS/TBOS operating system. The HITAC 8400 is (more or less) Type I virtualizable [see Appendix P]. TOS/TDCS satisfies software rule S1 but does not satisfy S2 or S3. Consequently, modifications were made to provide protection on an ad hoc basis. Other ad hoc solutions were introduced to eliminate some of the violations of Rule S3. For example:

A supervisor call instruction (SVC) is not a privileged instruction but it causes interrupts changing the current program state to P3. As TOS/TDCS had no set-exit function for this instruction, we used an illegal code instruction instead of a SVC instruction to permit simulation through a (sic) illegal code trap [49].

The authors indicate some of the Type II requirements.

The supporting operating system must provide a "set exit" or "ON" function for these types of traps caused by hardware. Aside from privileged instructions, the so-called SVC (supervisor call) instruction must also be "set exited" and simulated. The SVC normally communicates with the current supervisor but now it must be forced to direct to the supervisor simulated [49].

Recognizing the inelegance of their system, the authors propose some improvements:

> As for the TOD/TDOS, the interval timer should have finer precision and set-exit functions for address errors and SVC should be provided as standard features. In addition, "set storage key," "set time," and "set CAH" functions should be prepared [49].

C.2 360/67 UMMPS

A Type II virtual machine has been constructed on the UMMPS system for the IBM 363/67 [4,60]. Since UMMPS did not satisfy Rules S2 and S3, it was modified to include several new primitives. The most important of these is called SVC SWPTRA. A very brief description of the virtual machine monitor (called "the monitor", below) and the action of SVC SWPTRA is given in an informal user's guide.

> When you run the monitor, you tell it (by way of the VARY command) which virtual devices are attached to which real devices. You then issue the IPL command and the monitor fakes the IPL sequence. After this, you

algerting ----

AN CONTRACTOR OF A CONTRACTOR OF A

communicate with the virtual system by way of commands which make available the 360 console functions (display and alter memory, PSW restart, external interrupt, console request, etc.)

The monitor acquires space for and loads the virtual system into segment 3. The reason for using a new segment is this: when the virtual machine is running, it must have addresses 0-255K (or 0-512K or whatever). This is accomplished by SVC SWPTRA, which swaps the segment table entries for segment 0 and segment 3, detaches segments 0 and 1, and 4+, and transfers into the virtual system. This takes care of the addressing problem.

Now the virtual machine is running in problem state. Therefore, whenever it attempts a privileged operation (SIO, LPSW, SSK, etc.), a program interrupt occurs. At this point, the supervisor (UMMPS) swaps segments D and 3 back again and reconnects the other segments and transfers into the monitor. At this point the monitor can fake the privileged operation or not as it sees fit.

The sequence is therefore like this: The monitor transfers control to the virtual machine by way of an SVC SWPTRA. Any task interrupt in the virtual machine (program interrupt, time overflow, asynchronous cevice exit, etc.) will cause the supervisor to re-establish standard addressing and transfer into the monitor. [60]

Thus, SVC SWPTRA comes fairly close to being a primitive that starts up a subprocess. In its case, the particular choice of which interrupts trap back to the virtual machine supervisor nave been decided in advance. There is no general flexibility in deciding more precisely what is permitted in the subprocess. However, since the subprocess is a particular kind of virtual machine, ϵ .g. System/360 (360/65), there is no real need to provide the additional flexibility.

Note that the UMMPS virtual machine is not self-virtualizing. That is, the VMM runs on a 360/67 to produce a virtual 360/65. The work is currently being extended to

include a virtual 360/67.

and the second second

C.3 0S/360

OS/360 [66,84] is the principal operating system being run with the IBM System/360. As such it is extremely difficult (and possibly foolish) to implement ad hoc changes in it. One of the principal reasons that CS/360 is run is so that an installation will be somewhat compatible with the rest of the IBM world. Making ad hoc changes to OS/360 sabotages that objective.

Thus, in contrast to UMMPS, say, where the operating system was modified to support virtual machines, one would not expect to find changes to 05/360. This, in part accounts for the lack of Type II virtual machines under 05/360 sinch a standard 360 is Type I virtualizable. [See Appendix 8.]

In this section, we will examine some aspects of CS/360 and show how the public version of OS (pre-1970) violates Rules S2 and S3. We use as a guide, "Notes on Construction of Subsystems Hithin Operating System/360" by E. Satterthwaite [100]. Satterthwaite's objective is slightly different from implementing virtual machines, but the difficulties he uncovers are equally applicable. Since JS/360 is continually evolving, remarks made about it often tend to be time-dependent.

OS/360 violates rule S2 because all memory available to a Job is assigned a single key, and that key appears in the PSW of every task associated with that Job. This implies that the virtual machine is able to destroy the VMM. Sattertrwaite suggests that OS/360 provide a pair of SVC instructions, which,

after appropriate validity checking, switch the protection key of a part of a partition (region) between zero and the key associated with the corresponding lob.

OS/360 violates rule S3 in many different ways. Of the interrupts and exceptions, only program interrupts may be signalled to the virtual machine monitor in such a way as to be recoverable. For program interrupts in the virtual machine, e.g. attempts to execute privileged instructions, the VMM is able to issue an SVC SPIE to give the address of the appropriate "error handling" routine.

Other interrupts such as I/O, timing or SVC interruptions are not handled as well. For example, although the virtual machine supervisor can specify a time interval and an exit routine to be given control upon the expiration of that interval, there does not appear to be a method by which the exit routine may examine or alter the PSW or general register contents existing at the point of interruption. Thus, it is impossible to simulate the effect of this instruction. The worst failing of 0\$/360 concerns the lack of an hierarchical viewpoint concerning There does not appear to be a way in CS for a virtual SVC's. machine supervisor to mask out the SVC's of its virtual machire. Such а mask should enable a specified exit routine to conditionally inhibit the execution of selected supervisor services. Without this trap, there does not appear to be a way for the virtual machine's SVC's to be signalled, a violation of Rule S3a. Thus, in Figure C-1, execution of an SVC by the virtual machine (in problem state) causes a harcware trap to the



のないので、「ないないないない」のないで、ないないないないで、

OS/ 360 DOES NOT MASK SVC'S

FIGURE C-1

The partie of the concertion of the state

system and control passes to OS/360. CS/360 then interprets the SVC as a request for its, service, performs the service and returns to the calling virtual machine (in problem state). Thus the VMM never gets control and it cannot put the virtual machine into virtual supervisor state for the intended SVC handling. Note the similarity between the need to mask out SVC's of the extended machine and the need to cause tracs on sensitive instructions in the hardware machine. A masking SVC is, thus, very similar to the TCO instruction of Section 3.4.

C.4 MIT POP-10 ITS

While the PDP-10 is not virtualizable, it is, however, oossible to construct an HVM on it. [See Section 3.3 and Appendix 8.] Furthermore, wher running under MIT's ITS, Incompatible Timesharing System [42,43], it becames feasible to construct a Type II Hybrid Virtual Machine (HVM) [56].

ITS is an hierarchical operating system in which one process can control a sub-tree of other processes. In particular, ITS provides primitives (UUC's) which enable a process to create a subprocess, read the core image of the subprocess, and get aierted when the subprocess attempts to reference cutside its core image or attempts to execute a privileged instruction or UUC. The entire signalling mechanism (user trap mode) is controlled by one bit, called the UTRP bit, in the process's control block ("system variables"). A UUO to turn the UTRP bit on or off is available to the parent process.

Thus, it becomes a simple matter to design a Type II HVM.

National New York and National Action of the State of the

The VMM is the superior, the virtual machine the inferior in a process tree. The VMM sets the UTRP bit to direct all inferior traps to the VMM. The VMM permits the HVM to execute directly in user mode. When the HVM attempts to enter virtual executive mode, the VMM is alerted. The VMM interprets each instruction until the virtual machine returns to user mode. See Figure C-2.

C.5 CNS. The Cambridge Monitor System

CMS [85] is a rather bare single-user conversational operating system which runs or a standard 360. It is most often found in conjunction with CP-67. In this usage, CMS runs on a virtual machine under CP-67. CMS provides a number of operating system primitives that may be used by a prospective VMM. However, since CMS is 3 single user conversational system (most often run on a virtual machine), it does not provide any features that may be used to protect or signal the VMM 3s required by rules S2 and S3. At the same time, CMS does not make the nardware features (instructions) to accomplish these tasks inaccessible. Thus, with rather simple alterations, the necessary mechanisms may be added to CMS. While no virtual machine under CMS has been discussed in the literature, varicus installations, such as MIT Lincoln Laboratory, have made most of the modifications that make this possible.



TYPE I HVM UNDER PDP-10 ITS FIGURE C-2

APPENDIX U

GLCSSARY

The intent of the glossary is to provide, in one place, very ariet summary definitions for some of the terms or abbreviations in the thesis. For this reason there is no attempt made to .isec give very complete or particularly formal definitions. This Hossary is merety intended as a convenience for the reader, merinding him of centain Terms abbreviations. For L.C. TOPE information, the reader is directed to the section numbers indicated with most entries.

<u>architecture</u>: attributes of a system as seen by the programmer, i.e. the conceptual structure and functional behavior, as distinct from the organization of the data flow and controls, the logical design, and the physical implementation

<u>Associator</u>: nardware device for cerforming rapid parallel search

control program: virtual machine monitor

EMM: exterded machine monitor manping program between two different operating systems [2.1]

<u>emulation</u>: technique for simulating an interior decor on a machine with some different one

<u>exception</u>: trap caused by process violation [different from fault]: control should remain within executing virtual mechine [4.1]

<u>family</u>: compatible series of computers with possibly minor interior decor differences, e.g. 366/40 and 345/67,

<u>family-virtualizing</u>: the virtual machine is not identical to the nost but is a member of the same computer family [2.1]

<u>firmware</u>: microprogram which defines interior secon

EV: fraily-vintualizing 12.11

f: vintual machine map [4.2]

1: crocess man [4.1]

<u>demenation</u>: architectural characteristics of the principal hachings of that generation [In the thesis, generation always peters to architecture, never to physical implementation, etc.]

<u>4V</u>: horsware virtuslizer [4.5]

HVM: hyprid virtual machine [3.3]

<u>nargware virtualizer</u>: collective title for hardware-firmware resign which directly supports the (IV generation) VCS model [4.⁶]

<u>nost machine</u>: bare machine on which virtual machine monitor runs [2.1]

<u>lost operating system</u>: cherating system under which Type II vintual machine monitor runs [2.1]

<u>nyprig virtual machine</u>: virtual machine in which all instructions issued within the most privileged laver are software interpreted and all other instructions execute pirectly [3.3]

II <u>ceneration</u>: second generation architectures typical of IBM 7990 [4.3]

III generation: third generation architectures typical of IBM 360 [3.1]

<u>IV generation</u>: fourth generation architectures with elaborate, rich process structure implemented in firmware [4.1]

interior decort software-visible definition of a system, i.e. architecture

intra-level map: map within a virtual machine level which astablishes the (layer) structure within that lavel; the map may be visible to privileged software of that level [4.1,4.2]

inter-level map: (VMman) map between two levels of virtual machines, e.g. level n to level n+1, which is invisible to the nigher level, e.g. level n+1 [4.7]

invisible: unable to be detected in software

A CONTRACTOR OF THE PARTY OF TH

<u>layer</u>: restricted access structure (within a level) such as haster/slave modes or rings [2.1.4.2]

<u>level</u>: number of VMmaos which must be successively applied to map a virtual resource name into a real resource name: also corresponds to the number of syllables in the VMID: thus, the real machine is level \leq

<u>native</u> mode: operation of system, e.g. CPU, with preferred order code, not emulation

order code: instruction set

<u>200</u>: process control block, process status kept in system base [4.1]

2-name: name used by process to refer to some entity

process map: abstraction of the map from process hamas, e.g. segment numbers, to resource names, e.g. remory locations [4.1]

<u>privileged</u> instruction: instruction which executes (correctly) only in orivileged mode, i.e. supervisor state, ring zero [3.1]

<u>Rename:</u> real resource rame [4.2]

3-3: relocation-bound form of address relocation, as in DEC PDF-10

relcadable control store: modifiable microprocessor storage

nescurce map: virtual machine map

ning: layered protection system, ceneralization of master/slave mode, supervisor/proplem state [4.1]

ning 1: most privileged ring [4.1]

<u>2001</u>: origin pointer of system base of typical IV generation machine [4.1]

<u>RPM</u>: running process word, in system base, contains identifier of active process [4.6]

<u>SV</u>: self-virtualizing [2.1]

<u>self-virtualizing</u>: the processor of the virtual commuter system is identical to the host [2.1]

<u>semephore</u>: means for interprocess communication in typical IV generation computer system [4+1,4+6]

<u>sensitive instruction</u>: instruction which, because of IJI teneration vintual machine software construction will give incorrect result if permitted to execute cirectly on the bost tachine [3,2]

<u>supervisor call</u>: (SVC) instruction which is used to communicate with the operating system [7:1]

system base: database directly accessed by fintware (and updated by privileand coftware) of typical IV generation system to support implementation of the process model [4.1]

<u>TOC</u>: trap or accode instruction, as hoc suggestion for designing 3 (III generation) machine in which privileges instructions are chosen at execution time, and thus can guarantee. III generation software virtual machine construction [3.4]

Ivpe I VMM: the VMM runs on a bare machine host [2.1]

<u>Type II VMM</u>: the VMM runs on an extended machine host, under the nost operating system

UUD: supervisor call instruction on DEC PDP-10

VCS: vintual computer system

<u>VCSCB</u>: VCS control block in system base of hardware virtualizer provides VMmap for virtual machine [4.5]

<u>VCSIAB</u>: VCS table in system base of handware virtualizer contains pointers to VCSC3's for virtual machines [4.3]

<u>M</u>: vintual machine inever vintual memory)

VMCH: SAME (S VOSCB [4.5]

<u>VMIC</u>: virtual machine icentifier register in hardware virtu lizer indicates active virtual machine [4.5]

LVMID: load VMID instruction in hardware virtualizer, appends . next syllable to the VMID register and dispatches the virtual machine [4.5]

<u>UMM</u>; virtual muchine monitor, software which mediates between the virtual machine and host [2.1]

<u>JMISS</u>: virtual machine timesharing machine [2.2]

<u>VP</u>: Verice paper or proposal for firmware assisted virtualizer made in [51] [4.4]

V-name: virtual resource name [4.2]

Brought for any the second second

<u>VCS model</u>: composition of abstract process map and resource map to express running a process on a virtual computer system [4.2]

vintualization: act of creating/running virtual machine(s)

<u>virtualizable</u>: able to create (self-virtualizing) virtual machiny(s)

1M-fault: resource fault by virtual machine to VMM [4.2]

VM-level fault: VM-fault

THE ALL MADE AND A LODGER

<u>VMmsp</u>: abstraction of map between two levels of virtual resource allocation (completely distinct from process mas) [4.2]

<u>VM-recursion property</u>: in self-virtualizing virtual machine, ability to run a VMM on the VM [2.1.4.2]

EIBLICGRAPHY

- Ackerman, N.B. and Plummer, N.W. An implementation of a multiprocessing computer system. <u>ACM Symp. on Operating</u> <u>System Principles</u>, Satlinburg, Tennessee, (Oct., 1957).
- 2 Adair, R. and Barc, Y. CP-67 measurement method. <u>IEM</u> <u>Corp., Cambricae Scientific Center Rept. No. 6320-2072</u> (May,1971).
- 3 Addir, R., Bayles, R.U., Comesu, L:W. and Creasy, R.J. A virtual machine system for the 360/40. <u>Campricae Scientific</u> <u>Center Rept No. 6320-2007</u> (May, 1966).
- 4 Alexander, M.T. <u>lime sharing supervisor programs</u>. Univ. of Michigan Comp. Canter, (May, 1969, revised May, 1970).
- 5 Amdahl, G.M. and Andahl, L.D. Fourth-generation hardware. Detamation (Jan., 1967).
- 6 Amdahl, G.M., Blaauw, G.A. and Brooks, F.P. Architecture of the IBM system/360. <u>IEM Journal of Research and</u> <u>Development</u> (Arril, 1964).
- 7 Amdahl, L.D. Architectural questions of the seventies. Datamation (Jan., 1370).
- 8 Ancilotti, P., Cavina, R. and Lijtmaer, N. Virtual input-output in a virtual environment. <u>ACM International</u> <u>Comp. Symp. Proc.</u>, Venice, Italy, (April 12-14, 1972), pp. 3.2-312.
- 9 Auroux, A. and Hans, C. Le concept de machines virtuelles. <u>Revue Franczise d'Informatique et de Becherche</u> <u>Drerationelle</u>, 2e annee, 15 (1968), pp. 45-51.
- 10 Bairstow, J.N. Yany from one: the virtual machine arrives. <u>Computer Jecisions</u> (Jan., 1973), pp. 29-31.
- 11 Balzer, R.M. The ISPL language specifications. Rand Corp., R-563-ARPA (Aug., 1970).
- 12 Balzer, P.M. ISPL machine principles of operation. Rand Corp., R-562-ARPA (Aug., 1978).
- 13 Bard, Y., Margolin, E., Peterson, T. and Schatzoff, M. CP-67 measurement and analysis, IX regression studies. IBM Corp., Cambridge Scientific Center <u>Rep1. G320-2161</u> (June, 1970).

- 14 Pord, Y. Performance criteria and measurement for a time-sharing system. <u>IBM Sys</u>. J. 10, 3 (1971), pp.193-216.
- 15 Paskin, H.S., Borgerson, B.P. an Foberts, R. PRIME--A mosular architecture for terminal-priented systems. <u>Proc.</u> of <u>SUCC</u> (1972), pp.431-437.
- If Bensoussan, A., Clingen, C.T. and Daley, R.C. The multics virtual memory. <u>2rd ACM Symp. Op Sys. Prin.</u>, Princeton U. (Gct. 20-22, 1959), pp.30-42.
- 17 Bell, C.G. and Newell, A. <u>Computer structures: readings and</u> <u>examples</u>. McGraw Hill, N.Y. (1971).
- 18 Pelliro, J. and Potin, Ph. Mecanismes stun hyperviseur. Report of IBM Scientific Center, Grenoble, France.
- 19 Plaauw, G.A. Handware requirements for the fourth generation. Fourth generation computers: user requirements and transition. ed. F. Gruenberger, Prentice Halis (1975).
- 20 Pohrow, D.G., Burchfiel, J.D., Murphy, D.L. and Yomlinson, R.S. TENEX, a paged time-sharing system for the PEP-10. <u>Comm. of ACM</u> 15, 3 (March, 1972), pp. 135-143.
- 21 Brooks, F.P., jr. Architectural philosophy. <u>Planning a</u> <u>Computer System</u>, ec. W. Buchholz, McGraw Hill, N.Y. (1962), pp. 5-16.
- 22 Buzen, J. Optimizing the begree of multiprogramming in demand paying systems. <u>IEFE Comput. Soc. Conf.</u> Poston, Mass. (Sept. 22-24, 1971). pp. 141-142.
- 23 Calloway, P.H. Performance considerations for the use of the virtual machine capability. IRM Conn., T.J. Watson Research Lab., Yorktown Heights, N.Y., <u>Rept. RC-3360</u> (May 12, 1971).
- 24 Calloway, P.H., Considine, J.P. and Thompson, C.H. Uses of virtual storage systems in a scientific environment. <u>IBM</u> <u>Systems Journal</u> 11, 3 (1972) pc. 205-218.
- 25 Cheatham, T.E. The recent evolution of orogramming languages. <u>Proc. IFIP Cong.</u>, North Holland Pub. Co., Amsterdam (1971).
- 26 Comeau, L.W. A study of the effect of user program optimization in a paging system. <u>ACM Symp. on Op. Sys.</u> <u>Prin.</u>, Gatlinburg, Tennessee (Oct., 1967).
- 27 Comeau, L.W., Lindquist, A.B. and Seeber, R.R. A time-sharing system using an associative memory. <u>Proc. of</u> <u>IFEE</u> 54, 12 (Dec., 1960), pp. 1774-1779.

- 28 Corbato, F.J. System requirements for tultiple access time-shared computer. <u>MAC-IR-3</u>, MIT, Combridge, Mass.
- 29 Corbato, F.J., Saltzer, J.H. and Clingen, C.T. Multics--the first seven years. <u>Proc. of SJCC</u> (1972), pp. 571-583.
- 30 Corbato, F.J. and Vyssotsky, V.A. Introduction and overview of the MULTICS system. <u>AFIPS Proc.</u>, <u>FJCC</u> 27, I (1965), pp. 135-196.
- 31 CCSINE Task Force VIII. An undergraduate course on operating systems principles (June, 1971).
- 32 <u>CF-67/CMS</u>, Program 369 D-95-2-005, IBM Corc., Program Information Dept., Hawthorne, N.Y. (June, 1969).
- 33 Creasy, F.J. General description of the research time sharing system with special emphasis on the control program. <u>Unpublished memo No. 1 of IBM Scientific Center</u>, Cembridge (Jan., 1965).
- 34 Daley, R.C. Proposed 645 follow-on processon specification. Multics 645 Follow-on Handware Design Meto, MHDM-11, MIT, (June 23, 1977).

- 35 Daley, R.C. and Dennis, J.B. Virtual memory, processes, and sharing in Multics. <u>Comm. of ACM</u> 11, 5 (May 1968), pp. 306-312.
- 36 Dertmouth University. Specifications for GECOS ITI executive commands as implemented on the DTSS (Phase II) GECOS subsystem. Kiewitt Computer Center, Cartmouth University, Hanover, N.H.
- 37 Cerning, P.J. Third generation computer systems. <u>Computing Surveys</u> 3, 4 (Dec., 1971).
- 38 Denning, P.J. Virtual memory. <u>Computing Surveys</u> 2, 3 (Sect., 1970), pp. 153-189.
- 39 Dennis, J.B. Segmentation and the design of multiprogrammed computer systems. <u>JACM</u> 12, 4 (Cct., 1965), pp. 589-502.
- 4C Dennis, J.B. and Van Horn, E.C. Programming semantics for multiprogrammed computation. <u>CACM</u> 9, 3 (March, 1966), pp. 143-155.
- 41 Dijkstra, E.W. The structure of THE tultionogramming system. <u>CACM</u> 11, 5 (May, 1968), pr. 341-145.

- 42 Eastlake, D.F. ITS status report. <u>Memo No. 228</u>, AI Lab, MIT (April, 1972).
- 43 Fastlake, D.E. ITS 1.5 Reference Manual. MIT Artificial Intelligence Laboratory Memo No. 161A (also MAC-M-377). July, 1969.
- 44 Envlating DOS under CS for IBM System/370. IBM Systems Efference Library Form GC26-3777, IBM Corp., White Plains, N.Y.
- 45 England, A.C. and Byon, B. A DEC 10/53 simulator designed to run on ITS. Unoublished Project MAC meto (Aug., 1972).
- 46 Evans, D.C. and LeClerc, J.Y. Address mapping and the control of access in an interactive computer. <u>AFIPS Conf.</u> <u>Proc.</u> 30 (1967 SUCC), pp. 23-30.
- 47 Field, M.S. Multi access systems-the virtual machine approach. <u>IBM Campridge Scientific Center Pept</u>. <u>320-2033</u> (Sept., 1968).
- 48 Forgie, J.W. A time-and memory-sharing executive program for quick response. 1955 FJCC.
- 49 Fuchi, K. and Kaneda, Y. Utilization of a simple maging mechanism. <u>Bul. Flectrotech Lab</u>. 34, 4 (1977), cp. 55-63.
- 52 Fuchi, K., Tanaka, H., Namago, Y., and Yusa, T. A program simulator by partial interpretation. <u>2ng Symp. gn 2p. Syst.</u> <u>Prin.</u>, Princeton, New Jersey (Cct., 1969), pp. 97-164.

- 51 Gagliandi, U.O. and Goldberg, R.P. Virtualizeable architectures. <u>Prcc. of 1972 ACM International Corp. Symc.</u>, Venice, Italy (April, 1972), pp. 527-538.
- 52 Goldberg, P.P. Hardware requirements for virtual machine systems. <u>HICSS-4</u>, <u>Hawaii International Conference on System</u> <u>Sciences</u>. Honolulu, Hawaii (Jan., 1971).
- 53 Foldberg, R.P. Virtual machines: semantics and examples. <u>IFEE Comput Soc. Corf</u>. Boston, Mass. (Sept., 1971) co. 141-142.
- 54 Goldberg, R.F. Virtual machine systems. <u>MIT Lincoin</u> <u>Laboratory Report No. MS-2687</u> (also 28L-1936). Lexington, Mass. (Sept., 1969).
- 55 Goldberg, R.P. Unpublished lecture, Applied Mathematics 251a, Hervard University (Dec. 7, 1970).

- 56 Goldberg, R.P. and Galley, S.W. A Type II Hybrid Virtual Machine for the DFC POP-10 under ITS. MIT Project MAC internal note (Oct., 1972).
- 57 Granam, R.M. Protection in an information processing utility. <u>CACM</u> 11, 5 (May 1965), pp. 365-353.
- 58 Hans, C. et al. GMS guide de l'utilisateur. <u>Rept. of IBM</u> <u>Scientific Center of Grenoble</u>, Grenoble, France (July, 1972).
- 59 Hoernes, G.E. and Hellerman An experimental 360/40 for time-sharing, <u>Datamation</u> 14, 4 (April, 1958), pc. 39-42.
- 60 Hoja, J. and Madderom, P. The virtual machine facility--now to fake a 360. Univ. of British Columbia--Univ. of Michigan Internal Note.
- 61 Hollowsy, J. POP-1) paging device. <u>Hardware Merc 2</u>, "IT AI Lab. (Feb., 1976).
- 62 Holloway, J. Paging device signal names. <u>Harddare Memo</u> <u>24</u>, MIT AI Lab. (Feb., 1970).
- 63 Husson, S.S <u>Microprogramming: principles and practices</u>. Prentice-Hall, Erglewood Cliffs, New Jersey (1973).
- 64 IBM, Adding computer virtually. <u>IBM Computing Papt</u>., Vol. III, 2 (March, 1967).
- 65 IBM, Control program-67/Cambridge monitor system. <u>IHM Type</u> <u>III release No. 360D-05.2.005</u>. IBM Corp., Program Information Department, Hawthorne, N.Y.
- 56 IBM, OS/351 Concepts and Facilities. GC23-5535.
- 67 IEM, VM/370: Planning Guide. 6020-18(1-0)
- 68 IBM, IBM system/369 model 67 functional characteristics. SFL++.A27-6719.
- 69 IBM, IBM system/360 principles of operation. SFL+++A22+6821+
- 70 Infotech. <u>The IV Generation</u>. Infotecn, Maidenhead, England (in publication).
- 71 Keefe, D.D. Hierarchical control programs for systems evaluation. <u>IBM Sys. J.</u> No. 2 (1968), pp. 123-133.
- 72 Kelch, J.A., Seawright, L.H. An introduction to CP-67/CMS. <u>IBM Campridge Scientific Center Rept</u>. <u>323-2032</u> (Sept., 1968).

25 You MIT HER REPORT OF STREET OF STREET

73 Lemoson, B.W. An overview of the CAL time-sharing system. Comp. Center, Univ. of Calif., Berkeley (Sept. 5, 1969).

· VARIAN ALS

A CONTRACTOR OF A

- 74 Lampson, B.W. Cynamic protection structures. <u>AFIPS</u> <u>Conf</u>. <u>Proc.</u> 35 (1969 FJCC), pp. 27-38.
- 75 Lamoson, B.W., Lichtenberger, W.W. and Pirtle, M.W. A user machine in a time-sharing system. <u>Proc. of IEEE</u> 54, 12 (Nec., 1966), pp. 1766-1774.
- 76 Lauer, H.C. and Snow, C.F. Is supervisor-state necessary? <u>International Comp. Symp. Proc</u>., Venice, Italy (April 12-14, 1972), pp. 293-301.
- 77 LeClerc, J.Y. Memory structures for interactive computers. Ph.D. Thesis, Univ. of Calif., Berkeley, Calif., Project Gerie, Document 40.10.110 (May, 1966).
- 78 Lindquist, A.R., Seeber, R.R. and Comeau, L.W. A time-sharing system using an associative memory. <u>Proc. IEEE</u> 54, 12 (Dec., 1966), pp. 1774-1779.
- 79 Liskov, E.H. The design of the venus operating system. <u>CACM</u> 15, 3 (March, 1972).
- 80 Madnick, S.E. Storage hierarchy systems. Ph.D. Thesis, MIT, Project MAC (May, 1972).
- 81 Madnick, S.E. Time-sharing systems: virtual machine concept vs. conventional accroach. <u>Modern Data</u> 2, 3 (March, 1959), pc. 34-35.
- 82 Mallach, E.G. Emulation: practice and principles. <u>ACM</u> <u>Computing Surveys</u> (forthcoming).
- B3 McGrath, M. Virtual machine computing in an engineering environment. <u>IBM Systems Journal</u> 11, 2 (1972) PP. 131-149.
- 94 Mealy, G.H., Witt, Rule and Clark, K.A. The functional structure of OS/360. IPM Systems Journal 3. 1 (1966) pp. 2-51.
- 85 Meyer, P.A., and Ssawright, L.H. A virtual machine time-sharing system. <u>IBM Systems Journal</u> 3, 3 (1970), cp. 199-218.
- 86 MIT Project MAG Ine <u>Multiplexed Information and Computina</u> <u>Service:</u> <u>Programmen's Manual</u>. MIT Project MAC, Rev. 10; 1972.
- 87 Morris, D. The nature and benefits of modular operating systems. <u>The IV Generation</u>. Infotech, Maiderhead, Ergland (in publication).

- AB Morris, D. and Cottofson, G.C. An implementation of a segmented virtual store. Dept. of Comp. Sci., University of Monchester, Manchester, England.
- 89 Morris, D. and Dottofson, G.O. A virtual processor for real time operation. Deot. of Corp. Sci. University of Manchester, Manchester, England.
- 9) Motobayeshi, S., Masuda, T. and Takarashi. N. The HITAC 5(20 timesnaming system. <u>Proc. ACM Summer Conf</u>. (1969).
- 91 OfNeill, R.K. Experience using 3 tire-shared multi-programming system with dynamic adoress translation. <u>AFIPS</u> 30 (1937).
- 92 Ordanick, E.I. The multics system: an examination of its structure. HIT Press, Cambridge, Mass. (1972).
- 93 Parmetee, R.P. Preferred virtual machines for CP-57. IBM Cambridge Scientific Center Pert. G329-2063. (to appear).
- 94 Parmelee, R. Virtual machines: some unexcepted amplications. <u>Erac. ISSE Comput. Soc. Conf.</u> Boston, Mass. (Sept., 1971).
- 95 Parmetee, R.P.Peterson, T.T., Tillman, C.C. and Hatfield, D.J. Virtual storage and virtual machine concepts. <u>IBM</u> <u>Svs. J.</u>2 (1972), pp. 99-129.
- 96 Reigel, E.W., Fober, U. and Fisher, D.A. The interpretera microprogrammable building block system. Proc. of SJCC (1972), pp. 705-723.
- 97 Rosin, R.F. Contenborary concepts of microcrogramming and emulation. <u>Gorn</u>. <u>Surv</u>.1 (Dec., 1969), nn, 197-212.
- 98 Posin, R., Frieder, G. and Eckhouse, R. An environment for research in microprogramming and emulation. <u>4th Horkshop on</u> <u>Micropregramming</u>, Sonta Gruz (Sept., 1971), preprints.
- 99 Saltzer, J.H. Traffic control in a multiplexed computer system. <u>MAC-IE-32</u> (July, 1966).
- 100 Satterthwaite. E. Notes on construction of subsystems within operating system/36%. <u>Stanford Univ. Comp.</u> <u>Group Rept.</u> GGT* No. 43 (March, 1966).
- 101 Sayne, D. Adding computers virtually. IEM Corp., <u>Comput</u>. <u>Rept. III</u>, 2 (March, 1967) pp. 12-15.
- 162 Sayre, D. Is automatic "folding" of programs efficient prough to displace manual? <u>CACM</u> 12, 12 (Dec., 195°), pp.656-660.

- 103 Savre, D. On virtual systems. IBM Corp., T.J. Watson Research Lab., Yorktown Heights, N.Y. (April 15, 1966).
- 104 Schell, P.R. Dynamic reconfiguration in a todular computer system. MIT Project MAC, Cambridge, Mass. (June, 1971).
- 105 Schroeder, M.D. Performance of the GE-845 associative memory while multics is in operation. <u>AGM-SIG2PS Workshop</u> <u>on Sys. Perf. Eval</u>., Harvard Univ. (April, 1371), pp. 227-245.
- 106 Schroeder, M.D. and Saltzer, J.H. A hardware architecture for implementing protection rings. <u>CACM</u> 15, 3 (March, 1972), pp. 157-170.
- 1:7 Schwemm, F.E. Experience gained in the development and use of TSS/360. <u>Proc. of SJCC</u> (1972), pp. 559-570.
- 108 Spier, M.J. and Organick, E.I. An operating system model featuring demand scheduling of virtual resources. Presented at 3rc Hawaii International Conf. on System Sciences, Univ. of Honolulu, Hawaii (Jan., 1976).

- 109 Tucker, S.G. Faulation of Large Systems. <u>CACM</u> 8,12 (Dec., 1965).
- 110 Tucker, A.P. and Flyrn, M.J. Cynamic microcrogramming: processor proanization and programming. <u>CAC</u> 14 (April, 1971), pp. 240-250.
- 111 Van Horn, E.C. Corputer design for asynchronously reproducible multicrocessing. <u>MAC-TP-34</u> (November, 1966).
- 112 Watson, P.W. <u>Lineshering system design concepts</u>. McGraw Hill, N.Y. (1970).
- 113 Winett, J.M. Virtual machines for developing systems software. <u>Proc. IEEE Comput. Soc. Conf.</u> Boston, Mass. (Sept., 1971).
- 114 Wirth, N. On multiprogramming, machine coding, and computer organization, <u>CACM</u> 12, 9 (Sept., 1969).